The

Economic and Social Research Institute

Irish Educational Expenditures – Past Present and Future

A. DALE TUSSING

Paper No. 92

May, 1978

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE COUNCIL 1977–1978

- *T. K. WHITAKER, M.SC. (ECON.), D.ECON.SC., LL.D., President of the Institute. *I. F. MEENAN, M.A., B.L., Chairman of the Council.
- R. D. C. BLACK, PH.D., Professor, Department of Economics, The Queen's University, Belfast.
- *F. B. CHUBB, M.A., D.PHIL., Professor, Department of Political Science, Trinity College, Dublin.
- N. J. GIBSON, B.SC. (ECON.), PH.D., Professor, Department of Economics, The New University of Ulster, Coleraine.
- PATRICK A. HALL, B.E., M.S., DIP. STAT., Director of Research, Institute of Public Administration,
- *W. A. HONOHAN, M.A., F.I.A.

THE MOST REV. JAMES KAVANAGH, M.A., S.T.L., Bishop of Zerta.

- *KIERAN A. KENNEDY, M.ECON.SC., B.PHIL., PH.D., Director of the Institute. IVOR KENNY, M.A., Director General, Irish Management Institute.
- MICHAEL J. KILLEEN, B.A. (MOD.), B.COMM., D.P.A., Managing Director, Industrial Development Authority.

T. P. LINEHAN, B.E., B.SC., Director, Central Statistics Office.

*P. LYNCH, M.A., M.R.I.A., Professor of Political Economy, University College, Dublin. CHARLES MCCARTHY, B.L., Chairman, Human Scieńces Committee.

EUGENE MCCARTHY, M.SC. (ECON.), D.ECON.SC., Federated Union of Employers. *M. D. MCCARTHY, M.A., PH.D., D.SC., President, University College, Cork.

G. A. MEAGHER, B.COMM., D.P.A., Secretary, Department of the Environment.

- *C. H. MURRAY, LL.D., Governor, Central Bank.
- J. C. NAGLE, M.COMM.

D. NEVIN, Assistant General Secretary, Irish Congress of Trade Unions.

THE MOST REV. J. NEWMAN, M.A., D.PH., Bishop of Limerick.

TADHG O CEARBHAILL, Secretary, Department of Labour.

- REV. E. F. O'DOHERTY, M.A., B.D., PH.D., Professor, Department of Logic and Psychology, University College, Dublin.
- D. P. O'MAHONY, M.A., PH.D., B.L., Professor, Department of Economics, University College, Cork.

LABHRAS O NUALLAIN, D.ECON.SC., Professor of Economics, University College, Galway.

- *W: J. L. RYAN, M.A., PH.D., Professor of Political Economy, Trinity College, Dublin. REV. L. RYAN, M.A., D.D., L.PH., Professor, St. Patrick's College, Maynooth. T. WALSH, D.SC., Director, An Foras Taluntais.
- *REV. C. K. WARD, B.A., S.T.L., PH.D., Professor, Department of Social Science, University College, Dublin.
- NOEL WHELAN, B.COMM., M.ECON.SC., PH.D., Secretary, Department of Economic Planning and Development.

*Members of Executive Committee.

Irish Educational Expenditures — Past, Present and Future

Copies of this paper may be obtained from The Economic and Social Research Institute (Limited Company No. 18269) Registered Office: 4 Burlington Road, Dublin 4.

Price £4.50

(Special rate for students £2.25)

A. Dale Tussing is Professor of Economics at Syracuse University, Syracuse, New York. He was on secondment to The Economic and Social Research Institute for 1975/76. The Paper has been accepted for publication by the Institute, which is not responsible for either the content or the views expressed therein.

Irish Educational Expenditures — Past, Present and Future

A. DALE TUSSING

©THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE DUBLIN, 1978

ISBN 0 7070 0012 2

Acknowledgements

In conducting the research and writing this paper, I received advice, assistance, and information from a large number of people throughout Ireland. It seems unlikely that, at one sitting, I can remember them all; so it is appropriate to begin my acknowledgements with an apology to those inadvertantly omitted.

My permanent position is Professor of Economics at Syracuse University, in Syracuse, New York. I spent the academic year 1972/73 as the guest of Professor Patrick Lynch, in the Science Policy Research Centre, University College, Dublin, engaged in work unrelated to the present paper. It is fair to say that Professor Lynch was responsible for my first coming to Ireland, and for this I am eternally grateful. In addition, as an acknowledged authority in the subject matter of this paper, he provided invaluable aid and advice in the research for this paper.

In 1975/76, I was able to return to Ireland, to spend fifteen months as visiting Research Professor of Economics at The Economic and Social Research Institute, at which time most of the research for the present paper was conducted. The topic of the paper was suggested by Professor Kieran A. Kennedy, Director of the Institute, who also provided support and encouragement. Other members of the Institute's professional staff also provided me with aid and advice. Professor Robert O'Connor, Deputy Director of the Institute, helped with school matters, whilst J. G. Hughes was especially helpful in the areas of migration and population. Both read the entire manuscript, and suggested a great many corrections and improvements. John McGregor and Brendan Whelan gave advice, and their own labour on technical matters dealing with statistical technique and computer programming. Finola Kennedy, who had in connection with a previous ESRI paper done some exploratory research in the subject matter of Chapter 5, provided me with her notes and encouraged me to proceed. Agnes Breathnach also provided invaluable assistance in research on that chapter.

Many others, both inside and outside the Institute, provided vital information, advice, and assistance. Some are acknowledged in footnotes. However, some require special mention here. These include Dr Kenneth Milne, then Secretary, the Church of Ireland General Synod Board of Education; Fr. Leo Quinlan, then Director, the Catholic Primary School Managers' Association; Fr. John Hughes, S.J., then Director, the Secretariat of Catholic Secondary Schools; Roy Johnston, of Trinity College, Dublin; Joy Rudd, then of the Department of Labour, now of Trinity College, Dublin; William Hyland, of the Department of Education; John Horgan TD; and, finally, Margaret Larminie, a remarkable person (as is suggested by the fact that she was, when this research was conducted, administrative assistant to Fr. Hughes, in the Secretariat of Catholic Secondary Schools, and is now successor to Dr Milne, as Secretary to the Church of Ireland General Synod Board of Education), whose aid and advice were always available and often required.

None of the above are responsible for the final manuscript, for this I alone am accountable.

| | CONTENTS |
|--|----------|

| | General Summary | Page 11 |
|------------|--|------------|
| Chapter 1 | Introduction Structure of the System. The Future of Enrolment. The Future of Educational Expenditures. Determinants of Expenditure Differentials. Private Educational Expendi- tures. What is not Covered. Plan of the Work. | 16 |
| Chapter 2 | The Structure of the System National Schools. Private Primary Schools. Secondary Schools. Vocational Schools. Comprehensive and Community Schools. Third Level. | 21 |
| Chapter 3 | Sources and Resources of the System The Early Period. The Managerial System. How Could Ireland Afford the System? The End of Cheap Education. Spartan and Frugal System. The Interest of the Church. Contributions of the Religious. Technical Content. The Recent Period. The Near Future. "Baumol's Disease." | 43 |
| Chapter 4 | Enrolments and Expenditures: The Coming Explosion Enrolments. Population. Participation Rates. Enrolment Numbers. Pupils, Types of School and Teachers. Expenditures. What can be Done? | 72 |
| Chapter 5 | Private Current Expenditures on Education Other Estimates. Household Budget Survey. Investment in Education Estimates. CPSMA Survey. Other CPSMA Data. Convent Primary Schools. Survey Results. Private Primary Schools. Protestant National Schools, 1974/75. Private Expenditures. National Schools. National Schools, New Grants Scheme, 1975. Regression Analysis. Secondary Schools, 1974-75. Regression Analysis. Estimated Total.Expenditures on Secondary Schools. Conclusions. | 115 |
| Chapter 6. | Summary and Concluding Remarks First Level. Secondary Level. Third Level. Policy Recommendations. Concluding Remarks. | 162 |

Page

| APPENDIX 4.A. | Page |
|--|------|
| Adjustments to Published Population Estimates and Predictions | 179 |
| Underestimate of Numbers of Schoolchildren. Decline in Number of Births. Net Migration Assumptions. | |
| APPENDIX 4.B. | |
| Alternative Enrolment Forecasts for 1986 from New Population Projections. | 181 |
| | |
| APPENDIX 4.C. | |
| Teacher Requirements | 182 |
| National Schools. Second-Level Schools. | |
| APPENDIX 4.D. | |
| Assumptions and Methods used in Illustrative | 184 |
| Expenditure Forecast. | |
| | . • |
| | |
| ی ـ | |
| | |
| | |
| | |
| | |

.

LIST OF TABLES

| Table | · . | Page |
|-------------|---|------|
| 2.1 | Number of Persons Receiving Full-time Education on 1st February 1974, by Type of Institution, and Number as Per Cent of Total. | 22 |
| 2.2 | Types of Expenditure Financed from Public and Private Sources, National Schools, Private Primary Schools, Secondary Schools, Comprehensive, Community and Vocational Schools, 1975/76. | 25 |
| 2.3 | Geographic Distribution of Private Primary Schools and Pupils, by Religious Affiliation, 1975/76. | . 28 |
| 2.4 | Secondary Schools by Ownership or Affiliation, 1973/74 (Unofficial Tabulation). | 28 |
| 2.5 | Estimated Windfall Transfers to Families of Secondary School Pupils Benefiting under 'Free Scheme'. | 35 |
| 3.1 | School Enrolments in Ireland (32 Counties), Census Years 1841-1901. | 53 |
| 3.2 | Number and Percentage of Persons Five Years Old and Upwards in Population in Ireland able to Read and Write, Census Years 1841-1911. | 53 |
| 3.3 | Attendance and Enrolment of Children Aged $6 - 14$ in Ireland (26 Counties), $1925/26 - 1963$ | 54 |
| 3.4 | Selected Statistics, Leaving Certificate Examinations, 1951-72. | 71 |
| 4. 1 | Estimated Population, 1974, and Predicted Population, 1976, 1981, and 1986: Selected Ages and Age Groups. | 78 |
| 4.2 | Comparisons of Forecasts of School-age Population, 1986 by Walsh, NESC, CSO and Present Study (x 1,000). | 79 |
| 4.3 | School Participation Rates, First, Second and Third Levels, Actual and Predicted. Selected Places and Years. | 91 |

| Table 4.4 | Full-time School Participation Rates, First, Second and Third Levels, Ages 15-18, OECD Countries, Selected Years. | Page 93 |
|--------------|--|------------|
| 4.5 | Full-time School Participation Rates, First, Second and Third Levels, Ages 19-24, OECD Countries, Selected Years. | 94 |
| 4.6 | Enrolments and School Participation Rates, 1966, 1971, and 1974, and Predicted, 1976, 1981, and 1986, by Age. | 96 |
| 4.7 | Predicted School Participation Rates and Enrolments, 1986, by Type of School and Age. | 99 |
| 4.8 | Required Number of Teachers, and Students per Full-time Teacher, 1974 and 1986. | 101 |
| 4.9 | Public Education Expenditures, Year Ending March 31, 1974 (x £1,000), Current (1974) Prices | 103 |
| 4.10 | Public Education Expenditures, Year Ending March 31, 1974 (x £1,000), Constant (1970) Prices. | 104 |
| 4.11 | Predicted Public Education Expenditures, Calendar Year 1986 (x £1,000), Constant (1970) Prices. | 107 |
| 5.1 | Education Expenditure as Reported in National Income and Expenditure 1974 (x \pounds 1,000). | 119 |
| 5.2 | Household Education Expenditure as Reported in Household Budget Survey 1973. | 121 |
| 5.3 | Summary of Results of CPSMA Surveys of National Schools, Selected Areas, Years. | 125 |
| 5.4 | Per-Pupil Expenditures, Results of Survey of Non-Aided Primary Schools, by Catholic and Protestant Affiliation, 1974/75. | 128 |
| 5.5 | Estimated Per-Pupil and Total Expenditures, Non-Aided Primary Schools by Catholic and Protestant Affiliation, 1974/75. | 128 |
| 5.6 | Summary of Results of Survey of Protestant National School Income and Expenditure, 1974/75. | 130 |
| | | : |

| Table 5.7 | Public, Private, and Total Primary School Current Expenditure, Nine Months Ending December 31, 1974, and Twelve Months Ending December 31, 1975, with Public and Private National School Expenditures, Twelve Months Ending March 31, 1974. | <i>Page</i> 131 |
|--------------|--|-----------------|
| 5.8 | Summary of Results of Survey of National School Income and Expenditure, 1975. | 133 |
| 5.9 | Regressions of 1975 National School Income and Expenditure, Selected Concepts on Selected Independent Variables. | 137 |
| 5.10 | Secondary School Expenditure, Various Concepts, 1974/75, from Sample of Audited Accounts, and Supplementary Survey by Type of School, Totals for the State. | 141 |
| 5.11 | Secondary School Expenditure, Various Concepts, 1974/75, from Sample of Audited Accounts, Day Schools, Free Scheme Only, by Area. | 144 |
| 5.12 | Regression Results: Secondary School Expenditures, Various Concepts, Regressed on Selected Independent Variables, 1974/75, All Secondary Schools. | 145 |
| 5.13 | Regression Results: Secondary School Expenditures, Various Concepts, Regressed on Selected Independent Variables, 1974/75, Catholic Schools Only. | . 149 |
| 5.14 | Regression Results: Secondary School Expenditures, Various Concepts, Regressed on Selected Independent Variables, 1974/75, Catholic Day Schools Only. | . 153 |
| 5.15 | Regression Results: Secondary School Expenditures, Various Concepts, Regressed on Selected Independent Variables, 1974/75, Schools Accepting Boarders Only. | 155 |
| 5.16 | Regression Results: Secondary School Expenditures, Various Concepts, Regressed on Selected Independent Variables, 1974/75, All Secondary Schools. | 158 |
| 5.17 | Total Secondary School Expenditures, Various Concepts, 1974/75, Estimated from Estimating Equations in Table 5.16, Together with Numbers of Students, Total and Selected Categories. | 159 |

Table Page 5.18 Public, Private, and Total Secondary School Current 160 Expenditure, Nine Months Ending December 31, 1974 and Twelve Months Ending December 31, 1975. Figure IV.1 School Participation Rates, Ages 4 and 5: Actual, 1963-74. 81 and Predicted, 1976-86. IV.2 School Participation Rates, Age 14: Actual, 1963-74, and 82 Predicted, 1976-86. IV.3 School Participation Rates, Age 15: Actual, 1963-74, and 83 Predicted, 1976-86. IV.4 School Participation Rates, Age 16: Actual, 1963-74, and 84 Predicted, 1976-86. IV.5 School Participation Rates, Age 17: Actual, 1963-74, and 85 Predicted, 1976-86. IV.6 School Participation Rates, Age 18: Actual, 1963-74, and '86 Predicted, 1976-86. IV.7 School Participation Rates, Age 19: Actual, 1963-74, and 87 Predicted, 1976-86. IV.8 School Participation Rates, Age 20+: Actual, 1963-74, and 88 Predicted, 1976-86. V.1 Sources of Local Contributions to National School Funds, as .136 Per Cent of Total, 1975.

General Summary

This study examines educational expenditure in the Republic of Ireland. It reviews the structure of the Irish school system, with principal attention to questions of finance. It reviews the origins and recent history of the system, to uncover likely sources of changes to come. It examines the future of enrolments and expenditures, and makes recommendations for ways to deal with anticipated pressures and events. It looks for causal determinants of differences in educational expenditures as among regions of the country; the rich and the poor; pupils attending schools under Catholic and Protestant management, etc., and it makes estimates of private (i.e., non-State-financed) expenditure, on first and second-level schooling.

In our review of the structure of the system, four points stand out. First, the system is essentially an "aided" one. That is, relatively few schools are owned and operated by public authorities. Most schools are denominational in ownership and management. However, the State, through the general exchequer, provides the vast bulk of funds. Second, in terms of participation, the system is a pyramid, with a very large base and a very small peak. Considering first, second, and third-level pupils combined, two-thirds are in primary schools; roughly thirty per cent are in second-level schools; and fewer than four per cent are in third-level institutions.

Third, the school system in the Republic of Ireland is broadly and deeply developed. Participation is approximately one hundred per cent in the primary age groups, and very nearly so in the junior cycle of second level. Participation rates in second-level education exceed those in England and Wales, in Scotland, and in Northern Ireland.

The fourth point regarding structure is of a different order. In our study of financing, we reviewed the effects of the 'free scheme' for second-level schools, introduced in 1967/68. Our analysis shows that the main effect of the scheme was to relieve a great many parents from paying fees which they would otherwise have had to pay and, hence, to permit windfall increases in the levels of living of these families, rather than to provide the schools with more income. That is, most of the funds did not find their way into the educational system. Between the initiation of the 'free scheme' and 1974/75, we estimate that £31.7 million had gone in windfall benefits to the parents of Secondary School pupils.

No nation in Europe has a more firmly and deeply established educational tradition than Ireland, and the highly developed system of today undoubtedly owes a large part to this history. The structure of the modern system, which took shape in the nineteenth century, reflects two principal historic influences. One is the devout Catholicism of the large majority of Irish people, and the still larger majority of the population of the present day twenty-six county Republic. The other is the fact of centuries of foreign rule. The schools were for several centuries an important arena for struggles over religion, language, and nationality. These two influences, devout Catholicism and the struggle over foreign rule, individually and in combination, produced today's aided, 'managerial' system. It has been remarked that the aided, denominational character of the Irish school system, which is, to a considerable extent, a reflection of the Roman Catholic hierarchy's traditional suspicion of the State and its institutions, serves today, perhaps ironically, more to protect the interests of the five per cent of pupils who are Protestants than to protect Catholic education from State influence.

When Ireland was still essentially a peasant society, every parish had at least one primary school, and some had more than one; the vast majority of children attended these schools, and learned something. Today, with the lowest income levels in the European Economic Community, Ireland has an educational system which is highly developed and, as noted, one whose participation rates are higher than those in Britain. These are remarkable achievements. The study examines how they came about.

Socially, the system was not developed for the purpose of encouraging economic growth. Individually, its main function was not seen as preparing pupils for careers. Rather, the major function of primary education, and of secondary education also, was religious, moral, and intellectual instruction. One implication is that education of girls has always been considered to be as important as education of boys, thus accounting for higher female (and hence total) participation rates than in most other countries Another implication is that instruction has traditionally been overbalanced in favour of such subjects as English literature, religion, Irish and Latin — highly verbal subjects, with low technical and quantitative content.

How could Ireland *afford* such a highly developed system? A review of the development of the system suggests four kinds of answers. First, Ireland has operated the schools in a spartan and frugal manner, especially at the primary level, in the National Schools, with historically extremely large classes, taught by poorly paid teachers, in ill-equipped and poorly maintained schools. To say this is not necessarily a criticism; the alternative, in many cases, was not to have schools at all, or (as in other countries) to educate only an elite. Second, the Catholic Church, with its great interest in education, has had a major role in marshalling the nation's resources for educational purposes. An extremely large (but unknown) amount of money for education has come from (or through) the Church.

A third way of financing Irish education beyond ordinary expectations has been the contribution made by members of religious orders — brothers, nuns, and clerics — both in terms of actual cash contributions (they have built most of the Secondary Schools, largely from their own resources, for example), and, more important, in the contribution of their services. Religious teachers return their salaries to their communities, which (with qualifications) means 'ploughing back' their salaries into the schools themselves. In some schools, religious work more hours per day and days per week than lay teachers. In a number of cases, especially in convent schools, religious have served as wholly unpaid assistant teachers.

And fourth, the Irish system of education has economised by having emphasised in the curriculum subjects with low technical content or which for various reasons have required little in the way of costly equipment.

It is worth considering these four sources and resources, first, because they have permitted extensive development of the system; second, because they have vitally influenced the character of Irish education, in nearly every non-financial respect as well; and third, for the very important reason that each of the four appears today either to be fading or to be entirely disappearing. As they go, Ireland will be left with a high developed — and expensive school system.

Teachers' salaries are now almost in line with the rest of Irish incomes. Class sizes are falling, and the standard of amenity in National Schools is rising. There are signs of Church acceptance of an increasing State role in education, and of a corresponding decline in relative Church financial contribution. The most dramatic as well as the most easily quantified changes concern the decline in relative numbers of religious teachers. In Secondary Schools, religious as a percentage of all teachers was about 50 per cent as recently as 1966; the figure fell to 31 per cent by 1974 (and still lower if Comprehensive and Community School teachers are included): and, we predict, this number will fall to about 20 per cent by 1986, in the absence of a sudden and sharp reversal of existing trends. In National Schools, the percentage falls from about 23 per cent in 1966 to 15 per cent in 1974, and (we predict) to 9 per cent by 1986. And, finally, the 'arts' component of Irish education, and in particular the classical component, appears to be in decline, while the technical component is rising. A more costly education, per pupil, is implied.

These are important trends, which will affect not only the financing of education but its content and character as well. In the study, we point to still further reasons, arising out of special historic events, for increases to occur in per pupil educational costs in the coming decade. All of these changes appear to be coming at about the same time as enrolments are expected to rise very sharply, thus implying very considerable increases in overall educational expenditures.

This study forecasts enrolments in first, second, and third level institutions through 1986 by separately predicting population change in the relevant age group and predicting participation rates by year of age. Our population predictions rely primarily on the work of others, but three adjustments were found necessary. One reflected the high rate of child immigration, larger than elsewhere estimated, which has had the effect of increasing school populations. The second was the shortfall in births, beginning in 1975, as compared with predicted numbers, presumably a consequence of the severe world-wide recession. Third, we rejected the assumption, made in several population forecasts, that zero net emigration for the State as a whole implies zero net migration in each age group as well. Instead, our predictions are based on the assumption that recent migration patterns, by age group, will persist in the future, though with moderating force. The results of these assumptions show Ireland to be one of the few States in Europe with a rapidly rising youth population.

If adequate school places are provided, we expect school participation rates to grow the fastest in senior cycle, second level, and in third level. Between 1974 and 1986, we predict an increase in enrolment in all three levels of 188,700, or a gain of 23 per cent over the twelve years. This breaks into growth of 80,200 (14 per cent growth) in the group aged through 12 years; 36,800 (23 per cent) in the group aged 13 through 15; 35,400 (50 per cent) in the group aged 16 through 18; and 36,300 (159 per cent) in the group aged 19 and over.

The expected increases in per-pupil costs, together with the predicted increases in enrolments, point to explosive growth in educational expenditures. First- and second-level total public expenditures, when adjusted for inflation, are expected almost to double in the twelve years, 1974-1986. In addition, it appears likely that third-level expenditures, also adjusted for inflation, will far more than double over the same time period.

Because these enrolment and expenditure pressures are liable to precipitate efforts to economise wherever possible, the study calls for a major national debate on educational priorities. It is suggested that careful consideration be given to the differences between public and private benefits from education, and that public funds go primarily for public purposes. A scheme based on this and other principles, and which involves extensive changes in the financing of senior cycle, second-level, and third-level education, is offered, mainly as background to the hoped-for national debate.

The study concludes with estimates of private expenditures on education, as well as some analysis of the determinants of differentials in per pupil expenditures. In 1973/74, we estimate total locally (i.e., privately) financed current expenditures of National Schools to have been approximately £1.4 million, or 3.0 per cent of total current National School expenditures. Our estimate for calendar year 1975 is only £0.4 million, approximately 0.5 per cent of the total, which we take to be a temporarily depressed, transitional figure, arising from the introduction of a new financing scheme. In the small Private Primary School sector, we estimate total expenditures of £1.3 million in 1973/74, and £2.4 million in calendar 1975, none of it from public sources. Per pupil expenditures in Private Primary and National Schools appear to be very similar. For example, our estimates for 1974/75 are £94 per pupil in National Schools and £95 per pupil in Private Primary Schools.

In the calendar year 1975, approximately £50.7 million in public funds were spent on Secondary School education. We estimate private expenditure at £4.6 million, or 8.3 per cent of combined public and private spending of £55.3 million. In these figures, boarding expense, estimated at approximately £3.0 million, is excluded. If it were included, the private component would rise to £7.5 million, or 13.0 per cent of total Secondary School expenditures.

Our study shows regional differences in per pupil expenditures in Secondary School, even when size of school, denominational affiliation, and other influences are held constant. Most striking, perhaps, is the difference within the high and low income sections of Dublin City and Dun Laoghaire. More is spent than elsewhere in the high income parts of Dublin, in Kildare/ Wicklow and probably in Laois/Offaly/Carlow. Less is spent in the low income parts of Dublin, in Louth/Longford/Cavan/Meatń/Monaghan, in Leitrim/Roscommon/Mayo/Sligo/Donegal, and (with respect to Catholic Day Schools only) in Tipperary and probably Clare/Limerick. In addition, significantly more is spent in respect of eduction of Protestant than Catholic Secondary School pupils, even where other variables – including the use of religious teachers, and the presence of a boarding component – are controlled for.

Our study also shows the influence on per pupil and per school current expenditures of a number of other variables. Of particular interest is the fact that the per cent of lay staff is strongly and positively related to both per pupil and per school expenditures, even where an imputed charge is included for the services of religious. We interpret this as a clear indication, if not measure, of the economic contribution of religious to education in Ireland. In addition, pupils per teacher, an indicator of class size, is positively related to both per school and per pupil expenditures, as regards Catholic day schools. We also find that schools not in the free scheme, i.e., fee-charging schools, spend very considerably more than schools in the scheme, on both per-school and per-pupil bases, even where other influences are controlled for.

Much of the information estimated and published in this study is more or less routinely gathered and published by the public authorities in many other countries. So should it be in Ireland. This would include periodic publication of a document explaining and describing the organisational and fiscal structure of the system; enrolment and possibly expenditure forecasts; and total (public and private) educational expenditures, and their distribution.

Chapter 1

Introduction

The purpose of this study is to examine educational expenditures in the Republic of Ireland. Educational expenditure data are routinely published in a number of places¹, and it is not our purpose to duplicate this effort. In recent years, two studies of education expenditures in Ireland have been published: the important two-volume survey, *Investment in Education*², which itself is a milestone in Irish educational history; and, more recently, The National Economic and Social Council, Report No. 12. Our purpose is to add to this literature, by focusing on unknown, unexplored, or ill-understood questions concerning educational expenditures in Ireland.

Among these questions are the following:

1. Structure of the system.

While the 26-county State of Ireland is small, as is its educational system, the system is none the less exceedingly complex. This is particularly true of the financing of Irish education. Essentially, the system is an 'aided' one. That is, the educational institutions are not themselves 'public,' in the usual sense, but they are provided with substantial amounts of public funds. One evident need in the literature of Irish educational expenditures has been a working out of the essential structure of the system, especially as regards finance, in sufficient detail as to be useful and realistic, and yet without excessive legalistic complexities.

In working out the structure, a number of anomalies — inefficiencies, inequities, and what can only be called peculiarities — came to light. These anomalies were not the object of the study; but they are reported also.

2. The future of enrolment.

The implications of population growth have been widely discussed in Ireland since the publication of The National Economic and Social Council Report No. 5. One major purpose of the present study is to investigate the implications of probable future patterns of population growth for enrolment in Irish schools, particularly first and second levels, as well as third level.

In the process of working up enrolment forecasts, we discovered other

2. Investment in Education, Dublin, Stationery Office, 1965. See Chapter 3, below.

^{1.} The Department of Education publishes expenditure data periodically in its Tuarascail Staitistiuil. Educational expenditure data are also found in the annual Estimates and in Appropriation Accounts (see Chapter 4, below); and in National Income and Expenditure (annually) and the Household Budget Survey (see Chapter 5, below). Information on grants from the Higher Education Authority, to the universities and certain other third-level institutions are contained in the Authority's 1974 Progress Report and in its Annual Accounts. All the foregoing are published by the Stationery Office, Dublin. See also Kevin McDonagh, "The Way the Money Goes," Oideas 17, 1977, which was received after the present manuscript had been completed.

interesting phenomena, such as the (then) unexpectedly large occurrence of child immigration,³ which has obvious significance for enrolments.

3. The future of educational expenditures.

The large expected growth in child population in Ireland has obvious and important implications for educational expenditures. Our considerations of the subject of the future of educational expenditures turned up a number of other important sources of rather explosive potential future growth in the demand for resources.

To put the matter most succinctly, Ireland has had, for a variety of historic reasons, an unusually cheap school system. Its cheapness, in turn, encouraged a deep and broad expansion of the system. But the historic reasons for its cheapness are, one by one, falling away, threatening to leave Ireland with a highly developed, and extremely costly, education system, at just the same time as its largest enrolment growth ever is occuring. For example, the decline in religious vocations of the recent past is now catching up with the school system, implying higher expenditures in a variety of ways. For another, Irish education has traditionally been high on verbal and low on technical content; but today's labour markets require school leavers with at least technical potential, if not technical skills.

On top of these forces, there are special circumstances applicable only in the next few years, which will account for additional cost rises. Among these are the coming of equal pay in education, and a 'bulge' in the age distribution of second-level teachers, which will travel up the increments structure and bring disproportionately large salary increases.

4. Determinants of expenditure differentials.

It was decided to use as much cross-sectional data on the schools as was available, in order to see whether statistical methods would indicate sources of differences in educational expenditures. Our questions were such as these: Are there regional variations in per-pupil school expenditures? If so, are these explained by region (and, by implication, by regional income differences); or are they explained by other variables, such as school size, religious affiliation, extent of boarding function, etc.? Is more spent per pupil in Private Primary Schools than in National Schools? In Protestant than Catholic Schools? And so on. Our regression analysis also permits us to examine the net economic contribution of religious teachers.

Our purpose in the regression analysis is not hypothesis testing *per se*. Instead, it provides us with a more useful and rigorous way of reading the data. The method permits us to look at the relationship between expenditures

^{3.} The Population Office of the Central Statistics Office makes intercensal population estimates by age group. The Department of Education collects data on school enrolment annually, by year of age. We discovered that in many age groups, actual reported enrolments exceeded estimated numbers of children in the State; and, moreover, annual increases in school enrolments in age cohorts were fairly substantial. This information led to CSO revisions of population estimates and forecasts.

and each of a number of other variables, whilst holding constant ('controlling for') the influence of the other variables. Thus it is a high-powered alternative to tables with endless cross-tabulations. As there is no record of any prior 'determinants' studies of Irish educational expenditures, the results are of considerable interest. The data include some heretofore unavailable information on private sources of funds, adding further novelty and interest to the material. We examine not only the determinants of per-school and per-pupil school expenditures, but also the determinants of per-pupil and per-school expenditures out of non-public funds, which reflects the differential ability of schools in different circumstances to raise and spend funds from other sources.

5. Private educational expenditures.

There are virtually no data on private educational expenditures. All education expenditure data published on a regular basis occurs as a byproduct of budgeting for and operating the system. None of it is specially collected for purposes of publication. As a consequence, *public* education expenditure becomes a proxy for *total*, social educational expenditure. Of course, this is true not only in the Republic of Ireland, but in many, perhaps the majority, of other countries. What is different is that in most of these other countries, public expenditure is a good proxy for total expenditure, as their school systems are primarily public. In Ireland, publicly-owned, operated, and supported schools are exceptional; as noted, the system is an 'aided' one. This means, at least potentially, that public expenditure is not a good surrogate for total expenditure.

We set out to gather information about private expenditure of three sorts: the local contribution to the operating costs of National Schools; the total costs of operating Private Primary Schools, which receive no public funds; and the total expenditures, less State grants, of Secondary Schools. We were provided access to the audited accounts of Secondary Schools affiliated with the Catholic Secretariat of Secondary Schools. In addition, valuable survey data were given us by the Catholic Primary Schools. These data were supplemented by our own surveys of income and expenditures of Catholic National Schools, Protestant National Schools, Private Primary Schools, lay Catholic Secondary Schools, and Protestant and Jewish Secondary Schools.

What is not Covered

There still remains much primary work to be done on education expenditures in Ireland. Limitations on time, resources, and data, together with our own set of priorities, precluded examination of a number of important subjects.

We do not deal, for instance, with the issue of the 'returns' to 'investment' in Irish education. Such a study would require, in our view, a large national

survey, or a considerable improvement in household income data in conjunction with educational attainment data. Some such kind of survey is well worth doing, to focus primarily on the question: What are the educational determinants, controlling for background (socioeconomic status, sex, region, etc.) variables, of success in the labour market? For example, is the recent criticism of the 'arts' emphasis in Irish education justified? That is, what is the differential return to different kinds of education in Ireland?

We have not been able to develop estimates of future educational requirements in the labour force, though we devoted some time and effort to the question. In order to develop such estimates, one requires fairly accurate forecasts of the pattern of future economic growth; and no such forecasts were available, though we were led to believe that they exist, in the Department of Finance. Moreover, in spite of the criticism we heard repeatedly of the curricula of the Irish educational system, we arrived at the judgement that product demand, both domestic and export, together with macroeconomic problems, constitute the present limits to Irish economic growth, and that educational requirements scarcely enter in. That is, restructuring of Irish education along lines of forecast requirements seems unlikely to influence growth or employment significantly.

None the less, an improved use of both educational and labour resources could be brought about by an improved monitoring of current labour market developments, and perhaps anticipation of specific, near-term developments, and communication of these to the schools and universities, and through them to the students. The openness of the Irish economy, especially as regards migration, has permitted education and the labour market to develop somewhat independently. Where the education system provides trained personnel not needed by the economy, they can simply emigrate. Where the economy needs skills not provided by the education system, they can be attracted from abroad. While some such cross-emigrational pattern is inevitable in any case, and perhaps even desirable, there is very considerable latitude for bringing the education system and the economy into harmony. Complete success in this endeavour may require educational institutions to surrender some of their independence; but some progress could be made simply by the Department of Labour providing all second- and third-level institutions with quarterly or semi-annual reviews of prospects and problems in the labour market-areas with inadequate or excessive supply, relative to demand.

Finally, we have focused mainly on first- and second-level education, at the expense of third-level institutions, and to the complete exclusion of adult education and other programmes outside the conventionally bounded 'education system'. A survey of the Irish education system in its entirety, including, for example, all those institutions for which the Department of Education has no responsibility and over which it has no authority, would seem a useful thing to do. It would amount essentially to an assessment of

Irish educational resources, something which has never been done (though *parts* of the system have been subject to very searching examination); and such an assessment would be invaluable to the makers of educational decisions and policies, not only in the government, but in the educational institutions, and in the country at large.

Plan of the Work

An attempt has been made to make this study useful to specialists, and accessible and, possibly, even interesting to non-specialists. Thus an effort is made to explain the analytical and statistical techniques used. Similarly, the study is addressed primarily to the Irish reader, but an effort has been made, in terms of the discussion of the history, curriculum, institutional structure, etc., to make the study accessible and meaningful to the non-Irish reader.

The plan of the work is as follows:

Chapter 2 is titled, 'The Structure of the System.' It discusses each major type of educational institution, at each level, with particular attention to the role of public and private funds. Chapter 2 also includes an analysis of the distributional consequences of the 'free scheme' in second-level education.

Chapter 3 concerns 'Sources and Resources of the System.' It is argued in Chapter 3 that the characteristics, resources, and circumstances which made Irish education cheap and which, hence, induced its very considerable development, are in the main disappearing. The historical roots of a number of the characteristics of the unique Irish school system are also surveyed in the process. Recent educational policy changes are also discussed.

Chapter 4 deals with 'Enrolments and Expenditures: the Coming Explosion.' In order to forecast school enrolments in all three levels, Chapter 4 examines and modifies population projections prepared by others, and by predicting a future pattern of school participation rates, produces predicted enrolments. An illustrative plausible pattern of teacher requirements and expenditures is also set forth. As the title of the chapter suggests, the next decade is seen to be one of explosive growth in per pupil costs, in enrolments, and in the product of the two, educational expenditures.

Chapter 5 is titled, 'Private Current Expenditures on Education.' It presents estimates of total current expenditures on private, non-aided, primary schools; total and private National School current expenditures; and total and private Secondary School expenditures. This chapter also examines educational expenditure data as published in *National Income and Expenditure*, and the *Household Budget Survey*. And Chapter 5 also presents the results of cross-sectional multiple regression analysis of Secondary School expenditures, under the following headings: all Secondary Schools; Catholic Secondary Schools; Catholic day Secondary Schools; and all boarding Secondary Schools.

Chapter 6 presents a 'Summary and Concluding Remarks.'

Chapter 2

The Structure of the System

T he Irish system of education has points of difference and similarity to systems in other countries. In this chapter the system will be described and discussed, with special focus on economic aspects, particularly sources and use of funds.

Like most education systems, the Irish system can be described as consisting of three levels:

First level (or primary), beginning at the age of six (the age of compulsory attendance) and consisting of six years of study (in addition to which there is, usually in the same schools, a non-compulsory pre-primary or infants' division beginning at the age of four attended by some 85 per cent of children aged 4 and 5);

Second level (or post-primary – the term 'secondary' is reserved in the Irish system for a particular type of second level institution), consisting of two sub-levels or 'cycles', viz:

Junior cycle normally requiring three years of study (compulsory schooling ends at age 15), culminating in examinations for the Group Certificate after two years, and/or the Intermediate Certificate after three years' study; and

Senior cycle, normally requiring two years' study beyond the Intermediate Certificate, and culminating in the Leaving Certificate;

Third level (or post-secondary), whose time requirements vary, depending on the degree or qualification sought.

The schools making up these categories are listed, with their 1973/74 enrolments⁴ and relative importance in the system, in Table 2.1. While a great many types of institutions are listed, it is evident from Table 2.1 that only five types merit extended attention on grounds of size alone: National Schools, which enrolled 62.8 per cent of the full-time pupils at all levels in the State⁵; Secondary Schools, which between junior and senior cycle enrolled 20.4 per cent; Vocational Schools, which enrolled 7.4 per cent in junior and senior cycles; Private Primary Schools, which enrolled 2.9 per

^{4.} As this is written, the most recent published data on enrolments, etc. is for 1973/74 as contained in the Department of Education's *Tuarascail Staitist:..d*, which ceased annual publication in 1967/68 and has been published (with annual data) only twice in the subsequent six years.

^{5.} If Secondary Tops (second-level courses offered in National Schools) are included, the figure rises to 63.1 per cent.

Table 2.1: Number of persons receiving full-time education on 1st February 1974, by type of institution, and number as per cent of total.

| Type of Institution | Number attending | As per cent of students, at same level ^e | As per cent of students, all levels ^e |
|--|------------------|--|---|
| First Level: | | | |
| National Schools | 512,461 | 94.1 | 62.8 |
| Special Schools | 7,018 | 1.3 | 0.9 |
| Special Classes | 1,689 | 0.3 | 0.2 |
| Private Primary Schools | 23,260 | 4.3 | 2.9 |
| Others | 11 | 0 ^a | $0^{\mathbf{a}}$ |
| Total first level | 544,437 | 100.0 | 66.7 |
| Second Level: | | | |
| Secondary Schools, Junior Cycl | e 114,916 | 47.5 | 14.1 |
| Secondary Schools, Senior Cycl | e 51,680 | 21.4 | 6.3 |
| Secondary Tops, Junior Cycle | 1,924 | 0.8 | 0.2 |
| Secondary Tops, Senior Cycle | 325 | 0.1 | $0^{\mathbf{a}}$ |
| Comprehensive Schools, Jr. Cy. | 4,322 | 1.8 | 0.5 |
| Comprehensive Schools, Sr. Cy. | 1,374 | 0.6 | 0.2 |
| Community Schools, Jr. Cycle | 3,154 | 1.3 | .0.4 |
| Community Schools, Sr. Cycle | 956 | 0.4 | 0.1 |
| Vocational Schools, Jr. Cycle | 48,927 | 20.2 | 6.0 |
| Vocational Schools, Sr. Cycle | | | |
| General | 5,757 | 2.4 | 0.7 |
| Secretarial | 3,997 | 1.7 | 0.5 |
| Technical | 353 | 0.1 | 0^{a} |
| Regional Technical Colleges ^b | 526 | 0.2 | 0.1 |
| Other Aided | 1,416 | 0.6 | 0.2 |
| Non-Aided ^C | 2,030 | 0.8 | 0.2 |
| Total second level: | 241,679 | 100.0 | 29.6 |
| Third Level: | | | |
| Universities | 20,360 | 68.7 | 2.5 |
| Aided Teacher Training ⁰ | 2,368 | 8.0 | 0.3 |
| Vocational Technological | 2,907 | 9.8 | 0.4 |
| Regional Technical Colleges | 1,600 | 5.4 | 0.2 |
| Other Aided | 1,491 | 5.0 | 0.2 |
| Non-Aided Teacher Training | 232 | 0.8 | $0^{\mathbf{a}}$ |
| Non-Aided Religious | 787 | 2.7 | 0.1 |
| Total third level: | 29,640 | 100.0 | 3.6 |
| Grand Total: | 815,763 | · · · · · | 100.0 |

Source: Dept. of Education, Statistical Report, 1972/73, 1973/74.

Notes:

less than 0.05 per cent a:

b:

General, 201 pupils; secretarial, 165; technical, 160. Commercial, 1,738 pupils; religious, 20; radio schools, 272. c:

National, 1,847 pupils; vocational, 248; domestic science, 173. d:

Detail may not sum to 100 per cent because of rounding. e:

cent: and Universities, which enrolled 2.5 per cent. In addition, it is appropriate to attend somewhat to Comprehensive and Community Schools, and Regional Technical Colleges, because of their rapid rates of recent (and anticipated future) enrolment growth. The other types of institutions will not receive much discussion below. In their relationship to public authorities. the institutions listed in Table 2.1 can be described as in three categories: public. aided: and non-aided. The non-aided institutions in Table 2.1 are the Private Primarv Schools and the handful of institutions listed as 'non-aided' in second and third levels. All of these combined enrolled only about 3 per cent of all pupils, and their share is declining. Comprehensive, Community, and Vocational Schools and Regional Technical Colleges, among them enrolling fewer than 10 per cent of all students, may be described as public institutions. More than 87 per cent of pupils, then, attend 'aided' institutions. It is fair, then, to describe the system as essentially a State-aided one. In important respects, however, the structure of the system has been such as to blur the distinction between public and private, and this blurring regrettably has also affected the nature of statistical data available. The complex interweaving of public and private is summarised in Table 2.2 and is suggested by a description of the system.

National Schools: With minor exceptions, the National Schools are owned and managed either by Church bodies or religious orders. In spite of their names, they are not public educational institutions in the usual sense. As shown in Table 2.2, their principals and teachers, however, are paid directly by the State. The theory is that the State acts as 'agent' for the Management Committee (see below) in paying teacher salaries. In a practical, if not nominal, sense, the teachers are State employees. As Roman Catholic, Church of Ireland, and other Protestant National Schools are distributed around the country in proportion to the distribution of pupils of these faiths. the vast majority of National Schools are under Catholic management. In 1973/74, 92.6 per cent of schools, enrolling 97.5 per cent of pupils, were under Catholic management. Each School's Manager has been nominated by the 'Patron' of the school i.e., the Bishop, (There are trivial but interesting exceptions to the rule that the Patron is the appropriate Bishop of the Diocese. A few Protestant National Schools are under 'lay Patronage', where the Patron is a member of the landed gentry, e.g., an Earl. And the Minister for Education is Patron for Model National Schools) and traditionally the Manager has been the Parish Priest or Rector. In 1975, Managers were replaced by seven-member Committees of Management, four of whose members are appointed by and represent the Patron, one of whom is the principal teacher, and the other two of whom are elected by and from among the parents. The Chairman of the Committee of Management is usually the Parish Priest or Rector. In larger schools, committees can be as large as ten members, the Patron always maintaining an overall majority of direct appointees. To muddy the waters still further, the teachers — public employees in private institutions — are appointed by the Committees of Management. The Department, in negotiation with INTO (The Irish National Teachers' Organisation) sets salaries. It also establishes the curriculum and must approve all text books used.

The Constitution (Article 44.4) guarantees that "The State shall provide for free primary education". Current provisions, however, require substantial amounts of expenditure on National Schools from 'local sources', i.e., raised by Committees of Management from sources other than the Exchequer. Since 1975, the State provides a grant of £6.00 per pupil, provided that there is also a contribution of at least £1.50 per pupil from local sources. At time of writing the scheme has been in effect only one school year, little is known about how the £1.50 is raised; but from our own survey it is indicated that many schools ask parents to 'contribute' £1.50 per pupil.⁶ Moreover, local sources must pay about 15 per cent on average, of school construction costs and 100 per cent of site acquisition costs. There is nothing in the-Constitution or statute which limits organisation of National Schools to religious bodies; but these burdensome local costs of establishing a school make it difficult for an unaffiliated group of local parents or neighbours to organise a school.⁷ In addition, it is not automatic that a new school is recognised as a National School by the Minister. (Prior to this scheme, the State paid grants toward cleaning, heating, and painting of schools, and any shortfall in covering the costs of running the school was made up by local sources; the new scheme required a local contribution that had in many cases already regularly been made.)⁸

There are no centrally-collected data on privately-raised funds for current National School expenditures or on actual school expenditures (as opposed to State expenditures); and the data on privately financed portions of capital expenditures are limited to the 'matching' portion of construction expense. These gaps in the published data are particularly serious because the system is an aided rather than a public one. Prior to 1975, National School Managers were not even required to keep separate books as between school and parish (or religious community) accounts; the new requirement that separate accounts, subject to inspection and audit, be kept, will make it possible in

6. The survey, which is discussed in Chapter 5, indicates that just under 20 per cent of local funds are from parents' or pupils' contributions. 7. There might seem to be antagonism between these required forms of local contribution and the

7. There might seem to be antagonism between these required forms of local contribution and the constitutional guarantee of free primary education. Evidently the interpretation of the statement, 'The State shall provide for free primary education', is that the State will assure that every child *can* without charge receive a primary education, not that the State will necessarily pay for that education. So long as a free primary education is available to all children, the constitutional guarantee may be said to be fulfilled, at least according to this interpretation.

as a free primary education is available to all children, the constitutional guarantee may be said to be fulfilled, at least according to this interpretation. 8. According to a 1973/74 sample survey of Catholic National Schools, government grants under the old system covered only about 20 per cent of the year's expenditures for medium and large schools (six to nineteen classrooms), about 25 per cent for smaller schools and about 16 per cent for larger schools. The survey was conducted by Rev. Leo Quinlan, Secretary of the Catholic Primary School Managers' Association and the results are published under the heading, 'The Cost of Running a Primary School', in the *Education Times*, July 25, 1974.

Table 2.2: Types of expenditure financed from public and private sources, national schools, private primary schools,

secondary schools, comprehensive, community, and vocational schools, 1975/76

| Tuto of others! | Curre | nt expense | Capital expense | | |
|--|---|--|---|---|--|
| Type of school | Public | Private | Public | Private | |
| Classification National schools | Teacher salaries. £6 per pupil per year toward other current expenses. | Current expense in excess of grant aid, equal to at least £1.50 per pupil per year. | 84 per cent of construction costs ^a | 16 per cent of construction costs ^a ; 100 per cent of site acquisition costs | |
| Capitation National schools | Same as above in all respects except additional capitation grant (see text) paid to school instead of direct payment of teacher salaries. | | Same as above | | |
| Private primary schools | None | All private | None | All private | |
| Free secondary schools (Catholic) | 'Incremental' teacher salary and responsibility allowances. Capitation Grant of £18-£24 per pupil in attendance previous year; plus Supplemental grant of £50 per pupil in current year (amount payable for boarding students in low-fee (£265 boarding fees) boarding schools). | 'School salary', £400 p.a. All current expense in excess of grant aid. | 80 per cent of building equip- ment costs. | 20 per cent of building, equipment costs; 100 per cent of site costs; 100 per cent of costs of boarding facilities | |
| Fee-charging secondary schools (Catholic) | Incremental salary and Capitation grant, same as preceding, but no Supplemental grant. | Same as preceding | These schools eligible for building grants same as preceding, b practice less often receive them. | | |
| Secondary schools (Protestant) | Incremental salary and Capitation grant, same as Catholic schools, above. Block grant equivalent in total funds to Supplemental grant, given to inter-church committee which distributes aid according to pupil means. | Same as Catholic schools | These schools eligible for building g design specifications and hence do r same as Free secondary schools (Ca | rants but do not always agree to not always receive aid. Otherwise, tholic) above. | |
| Comprehensive and Community schools ^C (State schools) | All current expenditures paid by State | No private expenditures ^d | All capital expenditures are by State. | Nominal private expenditures by religious orders | |
| Vocational schools (Vocational Education Committee schools) | Of all current expenditure, for salaries, administration, main- tenance, or whatever purpose undertaken by VEC's, the sources are as follows: | Fees — collected only from post- secondary students and evening adult students, about 2 per cent | All capital expenditures are by State | · | |
| | Rates, approximately 13 per cent Grant from Department of Education to VEC's, 85 per cent | | | | |

Note: 'Public' expenditures are those financed by taxes, rates, or borrowing by the state or by local authorities; 'private' expenditures are those financed by fees, contributions, fund-raising events, etc., and by grants, subventions, etc. from church bodies, religious orders, etc.

Footnotes: a: Figures given are effective average; State share can vary between 66 per cent and 99 per cent.

b: Depending on number of students, number of attendances per year

c: There are differences in ownership, organisation between Comprehensive and Community schools but they are similar with regard to matters contained in this Table.

d: Comprehensive, Community, and Vocational schools are free to receive contributions from parents and others, and to undertake fund-raising campaigns to pay for items not covered in the Department's expenditure programme.

future to collect and publish data on school expenditures and on local sources. There has been no announcement that such data collection is contemplated, but it is most desirable that the State begin to collect and publish data on school expenditure. In 1978 it is none too soon to begin to report such fundamental type of data. Our own estimates of private expenditure on first and second-level education are reported in Chapter 5.

In 1973/74, 150,482 pupils or 29.4 per cent of the total attended convent or monastery National Schools and of these 52,227 (or 10.2 per cent of all National School pupils) attended institutions in which the State paid a capitation grant to the school, rather than paying salaries to the teacher. (In that year the grant ranged from £41.80 to £43.27½ on the first 100 pupils and from £27.92½ to £29.40 on the remainder.) The capitation option arose originally out of the needs of religious whose vow of poverty prevented them from taking salary cheques; in addition, these schools have been able to employ teaching religious who lack full qualifications. The 140 teachers in capitation monastery and convent schools in 1973/74 are presumably not even *de facto* State employees; that is, the State does not, either as employer or 'agent' give them salary cheques.

The State bears the full expense for operating a small number of schools. There are eight 'Model National Schools' managed just as others are, but fully financed by the State.⁹

Table 2.1 also indicates a number of pupils attending Special Schools and Classes: these are for such handicapped as deaf, blind, emotionally disturbed, physically handicapped, and mentally impaired.

Private Primary Schools receive no State funds of any kind. As a consequence, the State collects no data concerning them, except their enrolments. Thus, there are no available data on either their current or their capital expenditures.¹⁰ And, accordingly, there is no State control or supervision over curriculum, teacher qualifications, number of hours per day or days per year in instruction, etc. Even in these most private of private-sector schools, however, there is a certain blurring with the public and aided sectors, in that the vast majority of private primary schools are operated jointly, and share school grounds, with National Schools or aided Secondary Schools. Moreover, in many cases parents are said to enrol children in these Private Primary Schools, rightly or wrongly, to be more certain of securing a place in a Secondary School when the child is older. Thus the fees they pay are viewed by them as an entry charge which will subsequently gain the child access to a second-level education. Not only are Secondary School places limited in

10. See, however, Chapter 5, where we report our own estimates of their expenditures.

^{9.} These eight Model Schools are remnants of the system established here between 1831 and 1861 by the Board of National Education, which were the first public primary schools established anywhere under the Crown, and are regarded as a 'pilot' for the British system. They were to be multi-denominational in character. The Irish Catholic Hierarchy opposed the Model Schools, which eventually numbered 28, both on the grounds of their multi-denominationality and on the grounds that they were State schools. They were boycotted by Catholics and ultimately the National Schools' system under denominational management was established. See Chapter 3.

some areas, but some Secondary Schools, even in the 'free scheme' will not admit National School pupils. Pupils from lay National Schools have the lowest rate of entry into Secondary School.¹¹ In some cases, religious orders, who run most Private Primary Schools, are said to use superannuated teachers from other schools in their Private Primary Schools, thus implying another link.

| | Nı | mber of schoo | ls | 1 | Number of pupils | | | |
|-----------------|---------------------|-----------------------|------------------|---------------------|-----------------------|-----------------|--|--|
| Place-County | Catholic Schools | Protestant Schools | Total Schools | Catholic Schools | Protestant Schools | Total Pupils | | |
| Dublin | 70 | 9 | 79 | 15.070 | 1.158 | 16.228 | | |
| Louth/Kildare/ | | | · · · · · · | ····· | _, | 10,440 | | |
| Wicklow | 12 | 1 | 13 | * | * | 9 259 | | |
| Cork | 11 | 1 | 12 | * | * | 1,858 | | |
| Rest of Country | 15 | 2 | 17 | * | * | 1,348 | | |
| Total | 108 | 13 | 121 | 20,355 | 1,338 | 21,693 | | |

| Tabl | e 2. | 3: | Geograp | hic distrib | ution of | [°] private | primary | schools | and | pupils, |
|------|------|------|---------|-------------|------------|----------------------|---------|---------|-----|---------|
| ÷., | | ς, ' | 2 | by religior | ıs affilia | tion, 19 | 75-76 | | | |

*Withheld to preserve confidentiality. Source: Department of Education

| Ownership, affiliation | No. of schools | Ownership, affiliation | No. of schools |
|--|----------------|--|---------------------------------------|
| Orders of Nuns | | Orders of Priests | · · · · · · · · · · · · · · · · · · · |
| Sisters of Mercy | 108 | Society of Jesus | . 4 . |
| Presentation Sisters | 53 | Holy Ghost Fathers | 6 |
| Loreto Sisters | 22 | Others, fewer than 5 each | 24 |
| Irish Sisters of Charity Holy Faith Sisters | 14 12 | Total | 34 |
| Dominican Sisters | 10 | | |
| Others, fewer than 10 each | | Diocesan Colleges, total | 35 |
| Total | 300 | and the second sec | |
| | | Lay Catholic, total | 32 |
| Orders of Brothers | | - | |
| Christian Brothers | 85 | Dustanting total | 0.0 |
| De La Salle Brothers | 20 | Frotestant, total | 26 |
| Presentation Brothers | 11 | | |
| Others, fewer than 10 each | | Jewish | ` |
| Total | 130 | Total | 558 |

Table 2.4: Secondary schools by ownership or affiliation, 1973/74 (unofficial tabulation)

Source: Compiled from information furnished by Catholic Secondary Schools' Secretariat.

11. A significantly greater proportion of Secondary School entrants come from Catholic religious, Protestant, and Private Primary Schools than from lay Catholic National Schools. (Gearey (1973)).

As Table 2.1 indicates, the Private Primary Schools enrolled only 4.3 per cent of first level pupils, and but 2.9 per cent of all pupils, in 1973/74. Moreover, both percentages have been, and seem destined to continue, dropping. But they are none the less important institutions. Table 2.1 also shows that they enrolled more students in 1973/74 than the Universities, a somewhat striking statistic. Table 2.3 which is based on 1975/76 data, shows the Private Primary Schools to be clustered in two main areas: County Dublin and its immediate environs. Counties Louth, Wicklow and Kildare, and to a lesser extent in County Cork. Indeed, 65 per cent of the schools and 75 per cent of the pupils are in County Dublin alone, and 76 per cent of the schools and 85 per cent of the pupils are in Dublin, Louth, Wicklow and Kildare, When County Cork is added, the figures rise to 86 per cent and 94 per cent of the pupils. Three reasons might be offered for this concentration. First, these areas contain more wealthy persons who might be able to afford fees more easily. Secondly, certain middle-class families, especially in Dublin City, might not wish their children to mix with economically disadvantaged and educationally deprived children, who would be found in a relatively greater concentration in National Schools in those areas. Thirdly, there is more demand for Secondary School places relative to supply, and hence more competition for entry, in these areas than in the rest of the country, and hence more motivation for parents to seek secure places by enrolling children in affiliated Private Primary Schools. Whatever the reason, these schools obviously enrol a much larger percentage of all first level pupils in Dublin, its environs, and Cork, than would be suggested by the 4.3 per cent figure for the State.

Secondary Schools: Until the fairly recent introduction of Comprehensive and Community Schools, second-level schooling in the Republic of Ireland had been (with small exceptions) either in voluntary Secondary Schools, which have emphasised academic or 'arts' education, similar to that in grammar schools in other countries, and Vocational Schools, which have emphasised trade and technical education. In 1963/64, Secondary Schools (with Secondary Tops) enrolled 73 per cent of all second-level pupils; in 1973/74, the figure was 70 per cent.¹² Since Secondary School leavers have traditionally been stronger in verbal areas (English, Irish, Latin, history, nonlaboratory science) than in technical areas, the domination of second level by these schools has meant an 'arts' emphasis greater in Ireland than elsewhere in Europe, and one which has been the object of considerable criticism. Critics contend that the system has traditionally produced ample numbers of able candidates for such positions as civil servant or bank clerk but too

^{12.} Most of the data in the study are taken either from *Investment in Education*, Stationery Office, 1965, or from the Department of Education *Tuarascail Statistiul* various years. The 1963/64 data cited are from the former and the 1973/74 from the latter.

few technically and technologically competent leavers. The curriculum also strongly reflects University matriculation requirements.

In the event, second-level education began to change in a number of respects, essentially beginning in the late 1960s. Vocational Schools' curricula were expanded to include Intermediate and Leaving Certificate courses (q.v., below), and in theory a vocational pupil can today take a programme of study similar to that found in Secondary Schools. The examinations themselves, as well as the Secondary School curriculum, began to be altered, to provide for more technical content, a point to which we shall return in the next chapter. And Comprehensive and Community Schools, incorporating both 'arts' and 'vocational' education, were established for the first time.

Secondary Schools are denominational with minor exceptions - e.g., three schools are regarded as 'non-denominational Protestant' - and are of five general types. The most numerous are owned and maintained by Catholic religious orders, who have long dominated Irish second-level education. In 1973/74, 443 of a total of 534 Secondary Schools accounted for were operated by religious. Second, there are parochial or diocesan schools, called "Diocesan Colleges" of which there were 31 in 1973/74, which were originally established to prepare boys for seminary, and still serve that purpose. Third, there were 33 'lay Catholic' Secondary Schools, owned and operated, in effect, as businesses by individuals.¹³ Fourth, there were 26 Protestant Secondary Schools, many of them associated with the Church of Ireland. And fifth, there is one Jewish Secondary School, in Dublin.

An unofficial tabulation indicating ownership or affiliation of these Secondary Schools is provided in Table 2.4, which shows that three orders the Sisters of Mercy, the Presentation Sisters, and the Christian Brothers, among them maintained 246 Secondary Schools, or 46 per cent of the total. Accordingly, these three orders have had, and continue to have, an exceedingly important educational and cultural impact in Ireland. Table 2.4 indicates far more schools run by orders of nuns than of brothers. The former, unlike the latter, often tend to accept both boys and girls. Moreover, most of the schools operated by orders of priests and the overwhelming majority of Diocesan Colleges serve only boys.¹⁴ As might be expected, there is a degree of controversy over the domination of secondary education by religious orders. The issues raised include alleged or real religious contribution to sectarian nationalist traditions, to excessive 'arts' (as contrasted with technical) emphasis in the curriculum, and to Church domination of Irish Society. We do not treat any of these issues in this study. There has been a certain amount of economic significance of the heavy participation of religious, not only in Secondary Schools, but in primary education as well, on which we

13. That is, they took a proprietary form; no invidious connotation is intended. 14. Of §1 Diocesan Colleges accounted for in Table 2.4, 27 served boys only, and the other four were mixed. None served girls only.

do comment. On the one hand, the religious as individuals have had what might be called privileged positions with regard to certain matters. There is a built-in bias in their favour in recruitment, not only in Secondary but in National Schools, and the least qualified religious entering teaching is typically less able than the least qualified lay person. In time of recession and depression, this bias has meant that religious are fairly certain of employment, while lay people have far greater difficulty finding teaching posts. In Secondary Schools run by religious, the highest position available to lay teachers is that of Vice Principal, as Principalships, with higher pay and responsibility, almost always are reserved for religious.¹⁵ On the other hand, there is no doubt that the religious, and in particular members of the three large teaching orders already referred to, have made very substantial personal sacrifices in order, in effect, to 'subsidise' the nation in the education of its children, particularly in second level. Since the number of teaching religious are today in decline, meaning in effect that this 'subsidy' is being withdrawn, we will have occasion further to consider the economic contribution of the religious.

It is impossible safely to generalise about schools operated by the religious. In some, members of religious orders live what appear to the outsider to be very comfortable lives indeed, with extensive grounds and well-cared for gardens, and buildings of great beauty and character. Other religious evidently have lived frugally, both out of conviction and to devote the saved resources to education. Some have endeavoured to serve the poor. Others, especially the orders of priests, have concentrated on the relatively élite, and their schools, in the main, have not joined the 'free scheme' to which reference is made below.

While it can be stated that the exchequer contributed to Secondary Schools in five clearcut defined ways, the extent to which finance of Secondary Schools is 'public' or 'private' is an exceedingly complex matter, and there is no generally applicable answer. Each teacher receives a 'school salary', in recent years set at £400 per year, paid by the school and not the State. In addition, the State directly pays recognised, qualified teachers a so-called 'incremental salary', ranging in 1973/74 from £911 to £1,821 for women and unmarried men, and £1,150 to £2,341 for married men, with additional allowances for children, special qualifications, and duties. These added responsibility allowances loom large in arriving at total salaries. Obviously, the 'school salary', while not trivial, is far smaller than the socalled 'incremental salary'. The State has historically provided the Schools with aid in the form of grants, as discussed presently, from which the 'school salary' is paid, so both parts have a State origin. The reason for dividing the salary in this fashion and for calling the larger State share 'incremental', is to support the theory that Secondary teachers are private not public employees.

15. Access to full Principalships has come to be a vocal demand of ASTI, the Association of Secondary Teachers in Ireland.

State grants to Secondary Schools for running expenses are in two parts. The first is the so-called 'capitation grant', paid to all recognised Secondary Schools, which in effect means all Secondary Schools. In 1975/76, as shown in Table 2.2, this was set at £18-£24 per pupil (slightly more where instruction is in the medium of Irish). The grant is received in the year following their attendance. All types of schools listed in Table 2.4 are eligible to receive this aid, irrespective of form of ownership (e.g., including 'lay' or proprietary schools), and irrespective of whether they charge fees. Thus some rather élite institutions which charge fairly high fees and which would be wholly private in many countries do receive public funds in the Republic of Ireland, as well as being favoured by State payment of incremental salaries. In 1967/68 the Department of Education introduced the so-called 'scheme for free post-primary education', whereby a supplementary grant (originally £15/£25 per pupil; £50 in 1975/76) would be paid to all schools willing to forgo fees. (Grants are also paid on behalf of boarding pupils at so-called 'low fee' boarding schools, defined in 1975/76 as those with boarding fees of no more than £265 per year.) In 1975/76, 72 of the 534 schools accounted for in Table 2.4 charged fees. These included all Protestant Schools, which, however, had a variant on the 'free scheme' discussed later. Unlike the capitation grants, the in-lieu-of-fees grants are payable on this year's, rather than last year's, enrolments.

It will be noted that neither the capitation grant nor the supplemental grant in lieu of tuition is based in any way on outlays or on a matching 'local' contribution. Thus it is possible that some schools, with higher than average costs, cannot function without additional private sources of funds. Others may do fairly well, and indeed may, in a number of years, operate at a surplus.¹⁶ It is hence impossible to generalise concerning the mix of public and private funds in secondary education.

The situation involving the religious is one step more complex. Until 1974/75, most of the schools run by religious orders (with the notable exception of Christian Brothers' schools) operated on a financial system which Father John Hughes, S. J., until recently Director of the Secretariat of Catholic Secondary Schools, has called 'the bucket': all funds coming in were seemingly poured into one bucket, and all payments came out of that. same bucket, irrespective of whether they were on the account of the school or the religious community. No line was drawn between the community and the school. While agreement among the Association of Secondary Teachers' Ireland (ASTI), the Secondary Schools, and the Department of Education called for a school salary of £400 per teacher, that salary was only 'deemed' to be paid to religious. Moreover, as members of religious communities, these teachers routinely turned their incremental salary cheques over to their orders, i.e., to the "bucket". Since at that point community and school accounts were co-mingled, it cannot with any certainty be estimated to what 16. See Chapter 5 for further discussion.

extent these teachers 'ploughed back' their salaries into their schools. The extent to which that is so depends on the extent of community expenditures, which in turn depends very considerably on the level of living of the community, as well as on their other commitments. It seems virtually certain, however, that many nuns and brothers over a period of many decades have lived simply, frugally, and efficiently in order to turn over a substantial fraction of their incremental salaries to the education of the nation's children. This seems most certainly true, again, of the three large orders to which reference has already been made: the Christian Brothers, Sisters of Mercy and Presentation Sisters (as well as of a number of other orders). In 1973/74, the Secretariat introduced a uniform system of accounts, and in effect abolished the "bucket". In future, there will be three local parties to actual transactions: the teachers; the religious community; and the school. School salaries are actually to be paid to all teachers, lay and religious. Religious may then turn these salaries, together with incremental salaries, over to their communities. Separate records will be kept for community and school. Consequently, where the community in turn shifts funds to the school, there will be an explicit record. Likewise, where other members of the community perform services for the school (cleaning, administrative duties, boarding school duties, etc.), a value is to be placed on such services and a record kept. Thus in future it will be possible to determine how much, in money terms, the religious are contributing to secondary education, though some transactions are not 'arms-length' and must be regarded as nominal. It is regrettable that this practice was initiated only in 1973/74, in what one takes to be the twilight of this system. While there is no evidence that continued substantial domination of secondary education by religious orders is coming to an end, the decline in the number of religious as a percentage of all secondary teachers means a decline in this special form of financial or economic contribution, which regrettably was never measured in its heyday.

For a variety of reasons, a somewhat different scheme to that of supplemental grants in lieu of tuition was established for Protestant Secondary Schools when the so-called 'free scheme' was introduced. When that scheme first appeared, £25 was a rather typical annual fee in Catholic day Secondary Schools, thus when schools were told that they would be given an added grant of £25 if they were to forgo fees, most of them 'broke even', at least at first. (This was not the case for all, or for all time, as will be noted below.) At the same time, the typical fee in Protestant Secondary Schools was £50, a difference attributed by the Protestant Schools to the presence of religious orders in the Catholic schools.¹⁷ Protestant Schools were consequently

^{17.} A handout distributed by the (Protestant) Secondary Education Committee in 1969, taking Question/Answer form, says, 'Why is there a different scheme for Protestant Schools? Because very few of our schools can afford to charge fees of £25 or less and so are ineligible for the Government Scheme. Roman Catholic schools can do it because they are usually based on Religious Orders, and even then much sacrifice is entailed.' A spokesman for the Church of Ireland (Irish Press, January 9, 1975) is quoted as saying that the main reason for higher costs in Protestant schools at the time of

unable to give up fees altogether in exchange for a £25 grant. Instead, an arrangement was worked out under which the Department gives a bloc grant to an inter-Church committee representing the Church of Ireland, the Presbyterians, the Methodists, and the Society of Friends. The amount of the grant is determined by first calculating the percentage of Catholic schools which have adhered to the free scheme (usually about 92 per cent); applying that percentage to the total number of Protestant pupils; and multiplying that number by the current size of the supplemental grant. To this amount there has, in recent years, been added an additional amount to take account of the special boarding needs of Protestant pupils who in many rural areas live a very long way from a 'suitable' post-primary school. The inter-Church Committee, styled the 'Secondary Education Committee', then distributes these funds strictly according to means, and Protestant school children can receive anywhere from no aid at all to full payment of fees.¹⁸

The 'free scheme' is open to a number of criticisms, of which two stand out. First, while presented as a technique to improve access to education by the less advantaged, it distributed resources in ways often favouring the more advantaged, at least as compared with alternatives available, some of which evidently were considered. To a considerable extent the Protestant scheme is exempted from this criticism. Secondly, it failed to marshall the maximum of education resources per $\pounds 1$ of public funds available. The basis for these two criticisms lies in the following analysis.

Estimates have been made of the net effect on second-level enrolments of the introduction of the 'free scheme', and of the extension of the vocational curriculum to include courses toward Intermediate and Leaving Certificates (Tussing, 1976). These estimates appear in Column 2, Table 2.5 under the heading, 'Attributed Enrolment Gains'. (Figures for 1975 are extrapolated on the basis of trend.) While there are difficulties in allocating these estimated gains among Secondary Schools, Comprehensive Schools, and Vocational Schools,¹⁹ for present purposes we report (in Column 3, Table 2.5) as the 'Secondary Schools Share' a number proportional to Secondary Schools' enrolment as a fraction of second-level enrolment. In Column 3, we report

the introduction of the free scheme 'is that religious orders do not exist in the Church of Ireland'. The implied contribution of religious orders to education in Ireland is set for 1967/68 at an equivalent of, or near, £25 per pupil per year, or in excess of £2,500,000. Even a substantial scaling down of this estimate would leave a very substantial contribution indeed. There are other reasons, however, for this difference. Teachers in Protestant Secondary Schools are often paid a higher 'school salary' and these schools often provide 'extras' not found in Catholic Schools. Consequently, as is shown in Chapter 5, substantially more is spent per pupil in Protestant than Catholic schools, even when other influences (location, size, boarding and use of religious) are controlled for.

18. A number of relatively advantaged Protestant parents were vociferously resentful of the scheme worked out, as it meant they had to pay full fees, while equally advantaged Catholic parents next door paid none. The difference between the two was not only one of a difference in costs as between the two types of schools, but also the Protestants' decision to use the funds made available in a way that was progressively redistributive. 19. See Tussing (1976b), for a discussion. The 'free scheme' meant a change in the relative prices

19. See Tussing (1976b), for a discussion. The 'free scheme' meant a change in the relative prices and other structural differences as between different types of schools, and led to differential changes in enrolment (presumably shifts between types of schools) in junior and senior cycles.

| (1) School Year | (2) Attributed enrolment gain (x 1000) | (3) Secondary schools' share (x 1000) | (4) 'Free secondary pupils (x 1000) | (5) Estimated windfall beneficiaries (x 1000) | (6) Supplemental grant per pupil | (7) Total windfall (x 1000) |
|----------------------|--|---|---|---|--|-----------------------------------|
| 1967/68 | 8.4 | 6.0 | 114.1 | 108.1 | £25 | £2,702.5 |
| 1968/69 | 13.5 | 9.7 | ₹22.9 | 113.2 | £25 | £2,830.0 |
| 1969/70 | 18.8 | 13.5 | 132.8 | 119.3 | £25 | £2,982.5 |
| 1970/71 | 19.1 | 13.7 | 138.6 | 124.9 | £25 | £3,122.5 |
| 1971/72 | 19.2 | 13.7 | 144.6 | 130.9 | £25 | £3,272.5 |
| 1972/73 | 19.5 | 14.0 | 148.7 | 134.7 | £25 | £3,367.5 |
| 1973/74 | 19.6 | 14.0 | 153.3 | 139.3 | £45 | £6,268.5 |
| 1974/75 ^a | 20.0 | 13.9 | 157.8 | 143.9 | £50 | £7,195.0 |
| Eight years' to | otal | | | | | £31,741.0 |

 Table 2.5: Estimated windfall transfers to families of secondary school pupils benefiting under 'Free Scheme'

^aEstimated by extrapolation

Source: Figures in Column 2 are taken from A. Dale Tussing (1976). 'Labour Force Effects of 1967/68 Changes in Education Policy in the Irish Republic' The Economic and Social Review. Vol. 7, No. 3, April, 1976, pp. 289-304. Other data from Department of Education or calculated as described herein, see text.
THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

the estimated number of Secondary School pupils, or the equivalent in Protestant schools, who benefit from the 'free scheme' estimated as 92 per cent of the whole number of Secondary School pupils. If the Secondary Schools share in enrolment gain (Column 3) is subtracted from the beneficiaries of the free scheme (Column 4), the remainder (Column 5) is an estimate of the number of 'windfall beneficiaries' of the scheme, i.e., those who benefit from the free scheme, but who would have attended Secondary School, and whose parents would have paid fees, had the scheme never been introduced. It should be noted that the attributed enrolment gain (Column 2) is a conservative estimate.²⁰ In Column 6, we report each year's figure for the amount of the Supplemental Grant. Finally, in Column 7, the product of the last two previous columns is obtained, as an estimate of the total windfall gain to families of Secondary School pupils. For example, we estimate that in 1973/74, these families gained £6,268,500 - money they would have spent on school fees, had fees been continued, but which was paid instead, on their behalf, by the State. Between the initiation of the 'free scheme' and 1974/75, we estimate that £31,741,000 has gone in such windfall benefits to the parents of Secondary School pupils.

The figures in Column 7 obviously refer only to Secondary Schools. It will be recalled that the 'free scheme' applied as well to Comprehensive Schools (of which there were three in 1967/68) and Vocational Schools, both of which abolished their fees effective 1967/68. In light of this, $\pm 31,741,000$ must be taken to be a minimal estimate.

This £31,741,000, though labelled 'education' in the eight budgets involved, acted in effect as an increase in the disposable incomes of the families involved, and presumably went for housing, clothing, recreation, etc., rather than for education. While there may be important exceptions, these families are, in the main, among the more advantaged members of society. Moreover, to the extent of these funds, society has failed to marshall the maximum available amount of resources for education. For example, had there been introduced instead a free scheme only for low-income persons (i.e., for those qualifying by a means test), the remission of fees might have been supplemented, for those with especially low incomes, by a subsistence grant (NESC, 1976, Report No. 12).²¹ Thus far more students might have been brought into the system. Alternatively, had the funds been devoted to primary education, their educational impact would certainly have been greater, and more equitably distributed as well.

Where a segment of the community is evidently able and willing to pay

20. See Tussing (1976b). Regression equations were estimated with school participation rates for each year of age as dependent variable, using dummy variables to measure the effect of the policy. The coefficient for the dummy variable was accepted with a fairly weak test, viz. that the coefficient be larger than its standard error. Thus the estimated enrolment gain, while a conservative estimate, is more liberal than usual significance tests would require.

21. Such a scheme was evidently considered at the time, and rejected, though the history of the 'free scheme' suggests extraordinarily little consulation on this matter by the Department with people in the field of education.

for a benefit such as Secondary education, especially where for reasons other than fees the benefit is far from universally available, it appears foolishly wasteful (particularly from the perspective of budgetary stringency of the late 1970s) for the State to assume the costs.

Another, unrelated criticism is that since the grants bear no necessary relation to costs, the effect among schools has been capricious, and was particularly so at first. Moreover, schools were induced to forgo fees when a grant of approximately the same amount as most schools' fees was introduced; but then the grant was held constant, during a period of rising costs, for the next six years (see Column 7, Table 2.5) which brought many schools near or beyond the brink of closure, according to spokesmen for these schools.

It is widely believed (though obviously it cannot be documented) that this failure to keep the grants' structure in line with operating costs over time was in a sense deliberate. According to this argument, the Department of Education thought that the original free scheme had gone too far — that as argued above, it provided an excessive windfall to families at the expense of providing more educational resources. Holding down the grants in the face of rising costs amounted to *de facto*, a partial repeal of the scheme. Eventually, however, the pressures for higher grants became irresistible.

Whatever the reason, it seems fair to comment that the authorities did not foresee, or ignored, the probable *incidence* of the new grants (in the sense analogous to tax incidence), and that this error was followed by a policy on grant amount which brought undeserved hardship to many schools.

In addition to paying teachers' incremental salaries, and paying capitation grants to all recognised schools and supplemental grants to all free schools (and an equivalent amount to fee-charging Protestant schools), the State aids Secondary Schools in two further ways. One of these is fairly trivial: there are additional grants to equip classrooms, for science and other purposes, and toward the expense of choirs and orchestras. In 1974 the total amount involved was only £159,000.

The other is far from trivial. All recognised Secondary Schools, whether fee or free, Catholic or Protestant, are eligible for building grants toward the cost of new or enlarged schools. The State pays nothing toward the acquisition of sites, and 80 per cent of building costs. While in principle schools of all types are eligible for such grants, in practice they go only to free Diocesan and Religious Catholic Secondary Schools. Lay schools are permitted to continue to receive current grants, but their expansion is not being encouraged, i.e., funded. Fee-charging Catholic Secondary Schools are formally eligible, but are less often approved for building grants than free schools. Protestant Secondary Schools are eligible for grants, but not all their school buildings built since the introduction of the current grants' scheme have fit within the maximum specifications for schools eligible to receive grants. Boarding facilities are not eligible for capital grant aid, a rule which affects Protestant more than Catholic Schools.

State control over Secondary Schools is more indirect and hence somewhat weaker than for National Schools. The schools must employ at least a minimum number of qualified (Higher Diploma in Education) registered teachers, and only such teachers are eligible for incremental salaries. There is also a maximum number or quota of eligible teachers, related to the number of pupils.²² Otherwise the State has no hand in teacher selection. Curriculum and textbooks are influenced indirectly but powerfully through examinations, and through publication by the State of a syllabus corresponding to each year's examination work.

Vocational Schools: Vocational Schools are wholly public institutions. They are owned, operated, and maintained by Vocational Education Committees which are organised in 38 jurisdictions: four cities; seven urban districts; 25 of the 26 counties, and the two divisions of the 26th.²³ (The VECs, as they are called, also offer third-level education, and some second-level education, through Regional Technical Colleges and, in Dublin, through Colleges of Technology.) Since 1930, when the present arrangement was introduced,²⁴ the system has subtly changed from one in which vocational education was viewed as essentially a local function, obtaining half their funds from the rates and from nominal student fees, to one in which vocational education, like other second-level education, is viewed as a national function, though still operated by the local VECs. In 1973/74. in excess of 85 per cent of current expenditures, and all of capital expenditures, were from State funds.

Effective in the 1967/68 school year, with the introduction of the 'free scheme', tuition fees for second-level students were abolished, and the State grant to VECs was increased commensurately. At the same time, there were far-reaching curriculum changes which require some mention here. Until that year, vocational schools were charged with providing two types of education: 'continuation' and 'technical'. The former consisted essentially of the continuation of some or most of the subjects offered in National Schools: Irish, English, and sometimes commercial arithmetic, along with such courses as woodwork, metalwork, mechanical drawing, etc.; and led after two years' study to the 'group certificate' examination. The latter consisted of specialised. advanced work oriented courses, including apprenticeships.²⁵ Thus there was minimal overlap between subjects as studied in Vocational and in Secondary Schools. When the 'free scheme' was introduced, Intermediate and Leaving

25. Ibid., p. 13.

^{22.} As this is written, a school is permitted one teacher for each twenty pupils, plus a Principal, a

^{22.} As uns is written, a school is permitted one teacher for each wenty pupils, public a rincipal, and Vice Principal, and, if the school has more than 250 pupils, a guidance teacher.
23. The four cities are Dublin, Cork, Limerick, Waterford; the urban districts are Bray, Drogheda, Dun Laoghaire, Galway, Sligo, Tralee, Wexford and; the County VECs exclude the preceding cities and urban districts; and County Tipperary is divided into North and South Ridings. Members of the WEC VECs are appointed by the local rating authority.

^{24.} Previous to that, Vocational Schools were operated directly by the State (Investment in Education, p. 12).

Certificate courses were introduced in Vocational Schools, as, in effect, extensions of the 'continuation' programme. (At the same time, as already noted, some more technical subjects were introduced in these examinations, and in the Secondary curriculum.) Thus a very substantial overlap came to exist between the curricula of Vocational and Secondary Schools. Some have characterised the resulting Vocational Schools as, in effect, 'second-rate Secondary Schools', a description given even by some employed in the VEC system, who regret the relative decline of the vocational function. While we are not in a position to judge the characterisation, and in particular the 'second-rate' ranking, the characterisation does have some significance for the future, and will be referred to in Chapter 4.

Of all the types of Schools at first and second level, only Vocational Schools are inherently non-denominational. (Comprehensive and Community Schools, while they may be regarded as State schools, and indeed are the only schools about which that might be said, are effectively denominational in management.) None the less, it would be wrong to conclude that Vocational Schools are *secular*. Religion is an important part of the curriculum of Vocational Schools. And they have been subject to a good deal of ecclesiastical influence. A high proportion of VEC members are in fact clerics.

The influence of the Department of Education over Vocational Schools is somewhat greater than over Secondary Schools, as the Department must sanction staff appointments made by the VECs.

Comprehensive and Community Schools. For our purposes Comprehensive and Community Schools are essentially similar, and for the most part they will be dealt with together (as they are, in general, by the Department of Education). They are effectively State schools which offer both 'arts' and vocational courses, and they have features of both Secondary and Vocational Schools. The plan to build Comprehensive Schools was first announced in 1963, and the first school opened in 1966. They follow the pattern of Comprehensive Schools in Britain, except that they are specifically denominational. They are operated by three-person Committees of Management, representing the Department, the VEC in the area, and the Protestant or Catholic Bishop, or Religious Superior, as the case may be. This last member is the Chairman. In the case of schools managed by a Protestant Church or a Religious Order, two additional members are named by the Bishop or Superior, to assure the appropriate majority on the Committee.

Since 1974, no additional Comprehensive Schools have been opened, and there are no plans for further schools of this type. Instead, 'comprehensive education' is to be provided through Community Schools, the first three of which were opened in 1973. For the most part, these represent a joint venture between VECs and religious orders (one or two, depending on whether the school is to be co-educational). Some Community Schools, in

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

fact, have resulted from amalgamation of existing Secondary and Vocational Schools. The amalgamation may, or may not, involve common buildings. That is, the Community School may consist of separate school buildings, but with a common registration, and mutual shared use of facilities. More often, new Community Schools have been organised in newly settled or expanding areas, which do not involve the amalgamation of existing schools, but rather the co-operation of the Religious order and the VEC. Community Schools are controlled by Committees of Management consisting of two members appointed by the appropriate Religious or Diocesan authorities, two by the VEC, and two elected by the parents from among their numbers.

Though the Community School is evidently the leading edge of change at second level, expanded or re-built Vocational and Secondary Schools continue to receive building grants. New schools are supposed to take the Community School form.

Third Level: Third level education is not a central concern of this study, and is mentioned mainly for the sake of completeness. Moreover, third level education is complex in structure, and is in flux at the moment. Any adequate description would have to be lengthy; indeed, third level education deserves a study of its own. Our review here will of necessity be brief. As is clear from Table 2.1, the Universities dominate third-level education. Like Secondary Schools, they have in the past often been criticised for excessive concern with 'arts', as opposed to technical, subjects, as well as with the 'old professions' such as law and medicine. In the mid-1960s, the only other third-level institutions of significant size were the Colleges of Education or the teacher training colleges (to train National School and Vocational School teachers - Secondary teachers are trained in the Universities) and the two technological colleges operated by the Dublin VEC. Wide appreciation that Ireland was lacking in advanced technical training led to the development, with World Bank loan assistance, of a system of Regional Technical Colleges, as well as a National Institute of Higher Education, (in Limerick), which constitute the most rapidly growing sector of third-level education.

The two Universities²⁶ are empowered to award degrees. If and as other third-level institutions offer degree-level courses, the degrees must be 'validated' by one or the other University. A national Council for Education Awards exists to make other, non-degree awards.²⁷

26. The two universities are the National University of Ireland and Dublin University. The latter has only one constituent college, Trinity College. The former is divided into three University Colleges, in Cork, Dublin and Galway, and seven 'recognised colleges' including St. Patrick's at Maynooth. It has been announced (July 30, 1976) that each of these is to become a separate university. The other six recognised colleges are St. Patrick's College, Drumcondra; Our Lady of Mercy College, Carysfort; Mary Immaculate College, Limerick; the National Institute for Higher Education, Limerick; Thomond College of Education; and the Royal College of Surgeons in Ireland.

27. In 1977, the new government announced plans to return the power to award degrees to the NCEA.

The Universities²⁸ must be described as 'aided' institutions. The State funds a statutory body, the Higher Education Authority (HEA), which in turn allocates funds to the Universities. The University Colleges in Dublin and Cork also receive a substantial amount of funds from the Department of Agriculture. Approximately 85 per cent of their income for current expenditures is derived from these two kinds of State grants. Most of the remainder is received as fees. However, a fraction of fee income, too, arises indirectly from the State. Roughly one quarter of University students are recipients of Higher Education Grants, which are set at a level so as to cover fees and a minimal, subsistence level of maintenance. Grants are awarded on a combined means/merit/location basis which, in effect, imposes more stringent standards on less economically advantaged students. That is, the academic requirements for obtaining a grant are higher than those for entry into an institution of higher education. Funds for higher Education Grants come in part from the State (the Department of Education), and in part from Local Authorities. Higher Education Grants also go to students enrolled for certain courses in the Colleges of Technology, the National Institute for Higher Education, the Kings Inns (Law School), the National College of Art, and the Regional Technical Colleges. But these exceptions are minor; the grants serve primarily University students.

Expenditures by Colleges of Education, as well as a programme of loans and grants for students, are funded by the State, in the Primary Education vote, rather than the third-level vote. These too, are aided institutions and prepare teachers along denominational lines, i.e., Catholic Colleges of Education in general prepare teachers for service in National Schools under Catholic patronage, etc., though there are no rigid sectarian lines.

The Colleges of Technology in Dublin are under the Dublin VEC. The RTCs are nominally under the aegis of the VECs for the areas in which they are located, but funds and a substantial amount of direction come from the Department. As one commentator noted, 'As controller of funds, the Department of Education has immense influence. It has a direct say in the appointment of Principals, Heads of Departments, and Staff. Courses, equipment, and other facilities are more directly its concern — as holder of most of the purse strings — than that of the Vocational Education Committees' (Corcoran 1973). RTCs do charge fees, but these cover but a small fraction of their current expenditures. RTC students are covered by a system of Scholarships administered by the VECs.

The education system in the Republic of Ireland takes a form unique in the world because of the special history and character of the country - its

^{28.} In this paragraph the discussion of 'the Universities' also applies to the Royal College of Surgeons in Ireland; the College of the Pharmaceutical Society of Ireland; the National College of Art and Design, Dublin, the National Institute for Higher Education, Dublin (as this is written in 1977, still only a planning body); and the National Council for Educational Awards, Dublin (not a teaching but a validating body, as discussed earlier).

past as part of the United Kingdom, its economic structure, the homogeneity and heterogeneity (religious, racial, and cultural) of its people, and the traditionally high concern with religion as at the heart of the educational process. In the next chapter we will review the historical roots of some aspects of the system, not out of an intellectual, historical curiosity (though there is nothing wrong with that) but to determine if any seeds of future change are contained within the past and present.

Chapter 3

Sources and Resources of the System

The school system in the Republic of Ireland is broadly and deeply developed. Participation is approximately 100 per cent in the primary age groups, and very nearly so in the junior cycle, second-level age groups as well. Participation rates in second-level education exceed those in England and Wales, as well as those in Scotland and Northern Ireland.

In this chapter, we will look at a number of reasons why the Irish school system is so highly developed, and why it takes the particular shape and character it does. In particular, we will examine the special characteristics, resources, and circumstances that have combined to keep down money cost (whether total or public funds), and thus have encouraged full development of the system. We will also examine how some of these characteristics, resources, and circumstances may be in the process of changing.

This chapter is not intended as a history of Irish education. Rather, our purpose is fairly narrow. It is argued here that the characteristics, resources, and circumstances which made Irish education cheap and which, hence, induced its very considerable development, are in the main disappearing, and that per pupil costs and total expenditures are very likely to rise at an unprecedented rate in the coming decade. If the widely-quoted, and well-known forecasts of rapid population growth (Chapter 4) are borne out by events, then the growth in expenditures will be further aggravated.

The historical roots of a number of the characteristics of the unique Irish school system will be surveyed in the process. Several of the forces and trends that are altering costs are also altering the character and structure of the system, perhaps radically.

In the next chapter, we essay some quantitative estimates of the growth in population, participation rates, enrolments, per-pupil costs, and overall expenditures. In this chapter, we concentrate on the backgrounds and likely future characteristics of the problem. The discussion is organised into three time periods: the 'early period', which extends from the earliest times through the 1950s; the recent past, which consists of the 1960s and, to date, the 1970s; and the 'foreseeable future', which extends through 1986.

THE EARLY PERIOD

The Irish educational tradition is one of the oldest in Europe. Prior to Christianity, there existed the bardic schools which carried on a highly developed culture, a system of law, and medicine, history, poetry, and literature. The system of fosterage was an educational technique which began in these Celtic, pre-Christian times and which was evidently still in use in the eighteenth century. One explanation for the well-developed and articulated Irish school system is that there has been a concern with education from the earliest times, extending back into pre-history. After the beginnings of Christianity, Ireland became the 'land of saints and scholars' by virtue of the development and thriving of monastic schools during Christianity's darkest days on the continent of Europe. The bardic schools and the monastic schools existed side by side for more than ten centuries.

From the time of Henry VIII until the Treaty and partition, two themes recur in the history of Irish education, having significance for the present structure of the system. One is the use by the British and by the Irish Protestant ascendancy of their control over the education system for the purposes of suppression of Catholicism and Gaelic nationalism, and for political and sectarian proselytism. The other, a reaction to the first, is the tendency (in the schools and outside) of Gaelic Catholics to associate nationalism with Catholicism, and to associate both of them with hostility to an active State role in education. In the present Republic of Ireland, with a population which is 95 per cent Catholic, it is the minority 5 per cent who by rights should fear a State system of education, since such a system could hardly help being heavily Catholic-oriented. But instead, the Catholic Church evidently opposes State influence as much as, if not more than, do the Protestants. One reason is surely the experience of Catholics during the long years of British rule and Protestant ascendancy. It would hardly have been surprising that Irish Catholics would be suspicious of State education in a land in which the State for centuries represented the religious interests of a minority, even had there been no history of efforts to suppress Catholic education.

There was, of course, such a history. Henry VIII and Elizabeth I, in suppressing the monasteries, also supressed the monastic schools. There followed in the sixteenth and seventeenth centuries efforts to promote Protestantism and the English language in Ireland.²⁹ The Dublin Parliament passed the Act for the English Order, Habits and Language in 1537, and 'thus began a definite English State policy in Irish education, namely, the Anglicisation of the Irish people and the suppression of Catholic ideals in education.' Council of Education (1954, p. 12). Two years later Henry VIII suppressed the monastery schools. In 1570, in the reign of Elizabeth I, the Dublin Parliament passed the Act for the Erection of Free Schools. The purpose of the Act was to provide a proper education in order to deal with 'the rude and barbarous states' of the Irish people. The Act provided that 'there shall be from henceforth a free school within every diocese of this realm of Ireland, and that the schoolmaster shall be an Englishman or of the English birth of this realm'. The peak in efforts to suppress Catholicism came, of course, with the Penal

29. Council of Education, Report of the Council of Education (Dublin, the Stationery Office), 1960, p. 8. These two reports, of 1954 and 1960, contain useful brief histories of the school system, and will herinafter be cited as Report, 1954, and Report, 1960.

Laws, which (among other things) from 1691 through 1782 forbade any Catholic acting as a schoolmaster. Although Catholic education in any sense was outlawed in Ireland, it continued by virtue of the so-called 'hedge schools', so named because they were often conducted under the protective cover of hedges and in other outdoor spots. These schools were usually proprietary; the schoolmaster took fees from his pupils, and presumably had other sources of income as well. In addition to the hedge schools, there grew up a system of Irish colleges attached to Catholic universities on the continent, in Spain, France, and Belgium. As the enforcement of the Penal Laws was relaxed in the mid-1700s the hedge schools began coming out into the open (or rather, began going indoors), and they constituted an important part of the school system after the Relief Acts of 1782 and 1792.³⁰

After the ending of the Penal Laws, Irish Catholics were in a difficult position with respect to education. They distrusted the educational initiatives of the State, and equally so those of the various charitable trusts and foundations which had been established in the country to provide education for children. At the same time, though some parish and diocesan schools were started, and the hedge schools continued, Catholics were not really in a position to start and maintain a system of education equal to that of the Protestants.

McElligott has commented:

The conflict between the two main rival religious groups acted as a brake on all educational progress at a national level and made unanimity of approach to any educational question impossible. It can be argued that a system of education could ultimately have been created, acceptable alike to Catholic and Protestant, had Catholics not kept so completely aloof from all schools which accepted aid from charitable and proselytising agencies. They did so at a time when they were not in a position to provide schools from their own meagre resources and when, indeed, many Catholic religious orders had no thought of founding schools. History has shown how accurate was their assessment of the position which led them to await a time when the material conditions of the people was to give them unconditional control of their schools. Most of the orders were missionary in aim but the thought that education might become a State monopoly and, again, the anxiety to shape youthful minds after a desired pattern made them enter the field. They were supported by the mass of the people who saw in this work an opportunity of fulfilling their own hopes and, at the same time, embarrassing the government. As the schools had no endowments for their support, the religious orders were obliged to

30. Hedge schools came to be called pay schools because the children had to pay the schoolmaster. (Akenson, 1967, p. 4) (*Report*, 1960, pp. 26, 30) (Clarkin, 1969, pp. 92-93). The Penal Laws were also directed at Presbyterians (Dissenters) who also suffered educational disabilities (McElligott, 1966, pp. 56-57).

make many sacrifices to provide education on as widespread and cheap a scale as possible. (McElligott, 1966 p. 57)

In this setting there came the establishment of the first large religious orders devoted specially to teaching, and particularly the education of the poor: the Presentation Sisters, in 1800, and the Christian Brothers in 1802. A number of other orders followed, and a quarter century later the Sisters of Mercy were established. To this day the three are important in primary education. The vast majority of monastery National Schools are Christian Brother schools, and between them the Sisters of Mercy and Presentation Sisters also maintain the vast majority of convent National Schools, and, as discussed in Chapter 2, dominate secondary eduction.

The Managerial System

The system of National Schools had its origins in an effort, announced in 1831 in the House of Commons (which voted £30,000 in support), to provide what today might be called a multi-denominational, non-sectarian 'system of National Education'. Catholic and Protestant children were to attend school side by side, and the school would be used certain hours each week for religious instructions in turn by-Catholics, Anglicans, and Dissenters. The funds were disbursed by an Irish 'Board of National Education', on a basis not unlike the present system. 'Aid would be given . . . for the maintenance of the school, the payment of the teacher's salary, and the purchase of books and school requisites at half-price. For a grant to be given towards school building, at least one-third of the estimated cost was to be contributed locally and, in addition, a site approved by the Commissioner was to be provided, and the school house was to "be vested in trustees, to be approved by them"" (Clarkin, 1969, pp. 97-98); (see also Atkinson, 1969, pp. 93-94). The Board also initiated a system of Model Schools, whose purpose was mainly to assist in the training of teachers. The significant aspect of the Model Schools is that they were financed wholly by the Board, and hence constituted a system of free public schools, one of the first such systems anywhere, and the first ever in the United Kingdom. A handful of these Model Schools still exist: see Chapter 2. It is somewhat ironic that Ireland was, in the nineteenth century, a leader in the public schools movement, in light of the fact that the public sector of the education system is so small today.

There is no doubt that the development of a national system of education was a singular accomplishment for any country at that time, but in particular for Ireland. Akenson (1967, p. 15), historian of the early National School system, comments,

. . . After having seen that the English and the Scots obtained State systems of mass education only after undergoing economic and social revolutions, we can only be surprised to find Ireland in possession of a

State system of schools almost a full four decades before either of its neighbours. Ireland before the Famine was an 'underdeveloped country', even if no one had yet thought to use the term... The majority of its people were farmers, barely surviving by subsistence farming. ... Industry was only a minor sector of the economy. The average income of the people was low, and they lived under extremely poor housing conditions.

Akenson attributes Ireland's early development of a national system to five factors: (1) the fact that 'whatever its formal status in the eighteenth and nineteenth centuries' Ireland was a crown colony, well used to State intervention; (2) that in particular there was a history of State intervention in education, largely for Anglicisation and religious proselytism; (3) 'the Irish peasantry showed a striking desire for their children to be schooled'; (4) an official consensus had developed in the early nineteenth century on educational development; and (5) particular individuals on the scene at the time were instrumental in establishing such a system (Akenson, 1967, p. 15). The system was an aided one, as today. Before long it was determined that schools operated by Catholic religious orders would be eligible for aid just as any other school might, so long as the rules set down by the Board were complied with. These included scrupulous non-sectarianism. The Christian Brothers, who evidently had maintained aided schools briefly in the system, withdrew in 1837. But the system had been established, in effect, in response to a request from the Catholic Bishops of Ireland to Parliament (Clarkin, 1969, p. 95), and in 1841 Pope Gregory XVI urged all Catholics to participate (McElligott, 1966, pp. 3-4).

At the outset, the most vociferous criticisms of the new system came from the Presbyterians. Catholics, too, had a number of specific complaints and grievances against the system; but most complaints (though not all) of both groups were accommodated over time. Catholics sought to have schools vested in local trustees, rather than the Commissioners (i.e., the Board); and this was achieved in 1861. They evidently objected to the non-denominational design of the system, especially in that they often detected Protestant and/or English bias in the choice of textbooks. At first it was the Presbyterians more than the Catholics who objected to its non-denominational character, and they were instrumental in altering it. (Akenson, 1967, p. 95) (Atkinson, 1969, p. 97). "Whereas the textbooks had a heavy English bias, the books Sacred Poetry and Lessons on the Truth of Christianity had a heavy Protestant bias" (Clarkin, 1969, p. 101) (Report, 1960, pp. 42-44) (Akenson, 1967, pp. 225-ff.). The system evolved into one of denominational management under parish priest or Protestant clergyman. By 1860, 'the vast majority of schools ministered mainly to the needs of one denomination alone' (Atkinson, 1969, p. 99). The Powis Commission, a Royal Commission, chaired by the Earl of Powis, commissioned to examine the primary system in Ireland, appointed in 1868 to study education in Ireland, declared the attempt to establish a non-denominational system a failure, and declared the system to be, de facto, a denominational one (McElligott, 1966). Third, the Catholies objected to the non-denominational character of teacher training, in a Departmental training college, together with the use of the Model Schools. In response, a system of providing liberal aid to Catholic teacher training colleges was established, though not before 1883. Fourth, they objected strongly to the State-operated system of Model Schools on the grounds of improper State intrusion in a sensitive area. The model Schools were, in effect, boycotted by Catholics, and in 1863 became de facto Protestant Schools (Akenson, 1967, pp. 377-378). And fifth, they objected to the fact that the seven-man National Board consisted of five Protestants and two Catholics: the Duke of Leinster, as president of the Board; the Archbishop of Dublin and the Provost of Trinity College, to represent the Church of Ireland; the Archbishop of Dublin and the Chief Remembrancer, to represent the Roman Catholic Church; one Presbyterian and one Unitarian clergyman. (McElligott, 1966, p. 3). By 1860 the composition had been altered to ten Protestants and ten Catholics.³¹

The Managerial system has built within it centrifugal and centripetal tendencies. The name 'National School' is obviously a misleading one, based on a historical accident: there was an attempt to create a national school system, but instead a parish, or at best, diocesan system was erected. There is considerable latitude for local initiative and flexibility. On the other hand, the examination system came to be the means of central control. The Powis Commission just referred to was responsible for the introduction of the 'results system' by which teachers' salaries would be, in part, determined by pupils' results on examinations conducted by the Board's inspectors. The 'results system' (also adopted in the secondary system, as will be discussed below) was subsequently dropped.

The resulting system, with small changes to be noted below, is in structure essentially the system in operation today. In *substance* as opposed to structure, the system continued to attract strong criticism. Padraig Pearse considered the education system to be a 'murder machine'. While Pearse attributed the 'murder machine' to the English, it is a fact that much of what he inveighed against in this essay continued in the system after the establishment of Saorstat Eireann. Certainly nothing like the system Pearse described as 'ideal' was ever introduced in Ireland.

A limited form of compulsory attendance was introduced in 1892. At

31. Initial Catholic resistance to what amounted to a secular public school system may have had parallels in the USA at about the same time. Bruck, in his case study of the establishment of public schools in Lowell, Massachusetts in the 1840s, finds resistance among Irish immigrants to efforts, led by employers in the area, to establish free and compulsory primary schools. 'Irish parents and children evidently did not share the employers' enthusiasm for schooling,' note Bowles and Gintis, drawing on Bruck's work. 'Though the precise causes are obscure, the Lowell School Board reports document a sustained school boycott by the Irish community, and a number of attempts to burn down the school in the Irish neighborhood.' (Bruck, 1970, p. 164).

least 75 days attendance in each half year was made compulsory for children between the ages of six and fourteen. The provision applied, however, only in municipal boroughs and townships; rural authorities could apply it where they chose. In addition, the acceptable excuses for non-attendance included fishing work and harvesting operations. Historian Dorothy McArdle (1965) describes the resulting system as one organised

... to obscure the consciousness of a separate nationhood as far as possible, in the Irish people.... In these schools the Irish language was forbidden; the literature and history and legends of Ireland were not taught. The courses were designed with meticulous attention to detail for the Anglicisation of the rising generations. The children were taught to regard the English language, English history and culture as their own inheritance and England as the Mother country to which their whole allegiance was due.³²

And Akenson, summing up the achievements of the system, notes, 'The national system of education . . . was the chief means by which the country was transformed from one in which illiteracy predominated into one in which most persons, even the poorest, could read and write'. But, '. . . if it was through the national system that the Irish nation was given the blessings of literacy, it is important to note that the system taught the nation to read and write English, not Irish' (Akenson, 1967, pp. 377, 379).

The establishment of Saorstat Eireann answered the complaints concerning curriculum and textbooks, but there were few, if any, changes in structure or organisation of the primary system. This appears to have been even more the case of the First and Second Dáil (Ó Buachalla, 1977, pp. 57-75). The 'results' system was abandoned. Compulsory school attendance was extended to virtually all children aged six to fourteen. The Irish language became compulsory. But the primary system maintained its structure as an aided system, denominational in character, managerial in control. It was seen, as the Bishop of Clonfert described it prior to World War II, as 'perhaps as good a system as human ingenuity can devise to meet the rights and interests of Church and State'. The managerial system itself, Bishop Browne of Galway said in 1945, 'had given Ireland the most satisfactory state of Catholic school control of any country in Christendom'. It was a system which was a legacy of the British and of the struggle over schools of the nineteenth century.

^{32.} While the quotation from Dorothy MacArdle's book is a fair representation of the kind of complaints which nationalists have made against the National Schools, the point of view offered has been criticised *inter alia* by F. S. L. Lyons (1971) on the grounds that (i) Irish could be taught in the schools from 1879 onwards, though out of regular school hours and on a voluntary basis; and (ii) commencing in 1904 the National Board permitted schools in Irish-speaking or bilingual districts to teach Irish to all classes and to teach other subjects through the medium of Irish. In Lyons' view, the National Schools were not used to implement 'a coherent policy for the extermination of Irish' (pp. 49-50).

The secondary system, too, is little changed from that handed over in 1922. Officials in the Department of Education were kept at a good distance from the establishment of State schools, but they came to have considerable authority via the examination system. An effort was made in 1835 to establish an 'academy' in each county and an agricultural school and a 'college' in each province. (Clarkin, 1969, p. 100). This plan was unsuccessful because of the opposition of the Catholic Church. In 1878 the outlines of the subsequent aided system were established. Previous to that, the secondary system consisted primarily of voluntary schools and of secondary tops of National Schools.

The Intermediate Education Act of that year established an Intermediate Education Board to distribute State funds on the basis of written examinations. However, the funds were small, and operation of Secondary Schools required either fees of a substantial amount (for those days) or the personal sacrifices of members of religious orders. In spite of the decentralised nature of the system, the introduction of examinations made striking uniformity out of what had been diversity:

Before the year 1878 each school stood apart without any thought of, or direct concern for, what was being done in other schools. After that year schools had to comply with a common set of rules and regulations and accept a common curriculum and syllabus if they wished to have any share of the money provided by Parliament under the Act. The peculiar needs of a district or the strength of a headmaster's personality no longer counted for anything. As if to rivet the system more securely on the Irish people came the schools of the Christian Brothers, and it is arguable, if there was any discernible difference between any of the scores of schools opened by them during the lifetime of the Board. It was this levelling down of the standard of money which was most destructive of good teaching (McElligott, 1966, p. 60).

One significant effect of this 'results system' was to turn Secondary Schools away from practical and technical subjects, such as navigation, rural science, and woodwork, which were not covered in the examinations. This but reinforced a bias in favour of a curriculum seen as leading to desirable, clean, pensionable white-collar jobs. In 1902 the Act was amended to include reports of government inspectors along with examination results in determining the amount of funds going to each school. The nominally decentralised secondary or 'intermediate' system was in fact highly centralised. Largely for this reason, Padraig Pearse described 'the present [1916] intermediate system' as 'the most evil thing that Ireland has ever known' (Pearse, 1916). Under Saorstat Eireann, the system was substantially decentralised and, for a time at least, liberalised, by changing the basis of aid from examination results and inspectors' reports to a straightforward capitation basis, and freeing up the curriculum. McElligott reports that an 'open course' system commenced with the First Report of the new Department of Education in Saorstat Eireann, and lasted for fiteen years, which abolished required texts, and 'provided the teacher with a wide opportunity of developing some genuine literary taste in the pupils'. It represented one of the few swings in the direction of more progressive education to occur after the Treaty. It was eventually sacrificed not because of any educational objections but because of 'the difficulty experienced in devising suitable examination papers' under such a system (McElligott, 1966, pp. 61-69).

The present vocational system dates to 1930, but its antecedents go back to 1898 and 1899. In 1898 the Local Government (Ireland) Act empowered local authorities to levy rates for the purposes of technical education: and in 1899 the Department of Agriculture and Technical Instruction for Ireland was established. The result was a system in which local committees, under the local rating authorities, planned and built schools called Technical Schools. In 1925, under Saorstat Eireann, administration went over to the Department of Education. The income in 1925 of the statutory Technical Instruction Committees was 27 per cent from rates, 6 per cent from student fees, and the remainder, approximately two-thirds, from the State (Table 2.2). The Vocational Education Act of 1930 changed this structure very little. Vocational Education Committees were established in place of the Technical Instruction Committees. The Revenue sources remained virtually unchanged, though the State share has gradually grown since 1930. The most important change in 1930 was the broadening of the curriculum from 'technical' to 'vocational' education, where the latter was defined so as to include 'continuation' as well as 'technical' subjects (McElligott, p. 105).

Until the 1960s, third-level education was practically synonymous with University education.³³ There was considerable flux in the structure of higher education in the second half of the nineteenth century (comparable to that of the 1960s and 1970s), but the locus of university campuses has changed little since 1854. There are now, and in the foreseeable future will be, five university campuses:³⁴ two in Dublin, and one each in Maynooth, Cork and Galway. The oldest by far is Trinity College, Dublin, which opened its doors in 1594 as what was then seen as the first (and in 1977 still the only) constituent college of the University of Dublin. It is formally non-denominational, but its roots are English and Protestant. The other Dublin institution, University College, Dublin (as it is called in 1977), is a descendant of the Catholic University of Ireland (1854), whose first rector was Dr John Henry (later Cardinal) Newman. UCD (as it is known) was

33. In addition, there existed the Training Colleges, and the Colleges of Technology under the Dublin VEC.

34. On July 30, 1976, it was announced that the three University Colleges (Dublin, Cork, Gaway) making up the National University of Ireland, along with one recognised college, St. Patrick's Maynooth, are to become independent universities. Some expect the National Institute for Higher Education in Limerick, with the other institutions in Limerick, eventually to form another university campus, the sixth in the State.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

organised in large part out of the Catholic University as the base institution in the National University (1908), along with University College, Cork, and University College, Galway. These latter two began life (as did Queen's University, Belfast) as three Queen's Colleges, of the Queen's University of Ireland (1850), which were specifically and, evidently, sincerely nondenominational, organised 'on the principle of perfect religious equality. There would be no interference, positive or negative, with religious convictions. but religion would not be neglected. . . .' (McElligott, 1966, p. 136). In spite of the effort at non-sectarian University structure, the three Queen's Colleges became *de facto* Protestant institutions, as they were boycotted by Catholics, at least until the organisation of the National University. Queen's College, Belfast, 'well supported by the Protestant and Presbyterian middle classes, guickly acquired a reputation for the excellent work of the medical and science faculties. . . . The student body grew rapidly, so much so that many Ulster Presbyterians, failing to win a scholarship place in Belfast, went either to Galway or Cork, where, owing to the scarcity of Catholic students, the competition was less keen' (McElligott, 1966, p. 138). Thus the four university campuses, now all nominally non-denominational and funded by the State, began life respectively as a Protestant University, a Catholic University, and as two colleges of a de jure non-denominational, de facto Protestant university. The National University has received substantial State aid since its organisation; Trinity College, Dublin, has received an annual grant from the State since 1947.

How Could Ireland Afford the System?

By most standards, the Irish system of schooling has developed more rapidly and is today more highly developed than would have been predicted, on the basis of worldwide experience, for a country whose income levels, state of development, urbanisation and industrial structure have been those of Ireland. In primary education, participation rates approximate 100 per cent, and thus there are no states with higher participation rates. Moreover, these are actual participations, and not the fictitious enrolment-withoutattendance one occasionally finds in rural sectors. In the post-compulsory segment of second-level education, participation exceeds that of a number of more highly developed economies with higher *per capita* incomes. For example, it is not uncommon that British economic achievements are used as targets in Ireland; it is still more common that income and welfare benefit levels in the Republic of Ireland are compared (almost invariably unfavourably) with those of Northern Ireland (NESC, 1976, Report No. 12). It is instructive that school participation rates in the post-compulsory ages exceed those of England and Wales, and of Scotland, and of Northern Ireland.³⁵

In Tables 3.1, 3.2 and 3.3, the progress of the development of this system is shown. These Tables are all taken from T. J. McElligott, *Education in* 35. See Table 4.3

52·

Ireland. Tables 3.1 and 3.2 show, respectively, school enrolment and literacy in the period prior to compulsory school attendance.³⁶ It is noteworthy that in 1911, only 12 per cent of the population (32 counties) could neither read nor write. (A critic might argue that this measure examines the education system at its strong point — verbal ability — rather than its weaker points — quantitative, manual, or technical facility — a point to which we shall return presently.) Table 3.3 shows attendance as a percentage of enrolment, which is a fair proxy, after the start of compulsory education, for participation.

| Census period | In primary schools | In superior schools | In universities | | |
|---------------|--------------------|---------------------|-----------------|--|--|
| 1841 | 475,559 | 27,391 | | | |
| 1851 | 485,880 | 18,502 | . <u> </u> | | |
| 1861 | 443,433 | 21,674 | 1,711 | | |
| 1871 | 615,785 | 21,225 | 2,945 | | |
| 1881 | 675,036 | 20,405 | 4,288 | | |
| 1891 | 685,074 | 24,271 | 3,498 | | |
| 1901 | 636,777 | 38,565 | 1,598 | | |

Table 3.1: School enrolment in Ireland (32 counties), Census Years, 1841-1901

Note: In the years 1841 and 1851, universities were included under 'Superior Schools'. The returns for 1861 are for one day only, the numbers in the other years being the attendance for a week; hence the apparent decrease in the number in primary schools. In 1891, two establishments which were classed as 'colleges' in 1881 were included under 'Superior Schools' in 1891. The figures for 1901 are from the Census returns for that year, and the figure given under 'Universities'' is for the Queen's Colleges and Trinity College only.

Source: T. J. McElligott, Education in Ireland, Dublin, Institute of Public Administration, p. 13.

 Table 3.2: Number and percentage of persons five years old and upwards in population

 able to read and write, census years 1841-1911

| | Read and write | | Read only | | Neither read nor write | |
|-------------------|----------------|------------------------|--------------|------------------------|------------------------|------------------------|
| Census perio d | Total No. | Proportion per cent | Total No. | Proportion per cent | Total No. | Proportion per cent |
| 1841 | 1,966,000 | 28 | 1,413,000 | 19 | 3,766,000 | 53 |
| 1851 | 1,939,000 | 33 | 1,203,000 | 20 | 2,766,000 | 47 |
| 1861 | 2,106,000 | 41 | 1,023,000 | 20 | 1,973,000 | 39 |
| 1871 | 2,349,000 | 49 | 822,000 | 17 | 1,588,000 | 34 |
| 1881 | 2,726,000 | 59 | 714,000 | 16 | 1,158,000 | 25 |
| 1891 | 2,990,000 | 71 | 467,000 | 11 | 777,000 | 18 |
| 1901 | 3,187,768 | 79 | 276,580 | 7 | 551,715 | 14 |
| 1911 | 3,329,015 | 84 | 154,291 | 4 | 471,212 | 12 |

Source: Ibid.

36. Taking 1926 rather than 1892 to be the effective date of compulsory school attendance.

| School-year | Average number of pupils on rolls | Average daily attendance of pupils on rolls | Percentage of average daily attendance to average number on rolls | | |
|-------------|-----------------------------------|---|---|--|--|
| 1925/26 | 493,382 | 362,588 | 73.5% | | |
| 1933/34 | 502,661 | 422,266 | 84.0% | | |
| 1943/44 | 454,647 | 373,414 | 82.1% | | |
| 1953/54 | 472,536 | 404,618 | 85.6% | | |
| 1963 | 485,464 | 427,765 | 88.1% | | |

Table 3.3: Attendance and enrolment of children aged 6-14 in Ireland (26 counties),1925/26-1963

Source: Ibid.

When Ireland was still essentially a peasant economy every parish had at least one primary school, and not a few had more than one; and the vast majority of children attended these schools, and learned something. This is a remarkable achievement. It is fair to ask how it could have come about.

It can be observed that the system was not developed for the purpose of encouraging economic growth. Neither was the main function ever seen as preparing individuals for careers. Rather, the major function of primary education, and for those who went on to secondary education, was religious, moral and intellectual instruction.

One implication of this is that education of girls has always been as important as education of boys. This has meant higher overall participation rates than in countries where the education system was viewed more functionally, and where boys were favoured, especially in second level, because of their higher labour force participation.³⁷

Another implication is that education was overbalanced in favour of such subjects as English literature, religion, Irish and Latin — highly verbal subjects. If Ireland in effect inherited the beginnings of a developed school system, it also inherited a banking system and a civil service system developed beyond the general state of the economy; and the school system, fortuitiously one suspects, fed the banks and the civil service with competent paper handlers — people both literate and verbal.³⁸

How could Ireland, a poor country from all accounts, whose main economic activity was non-commercial agriculture, and whose housing conditions, infant mortality rates, and cash incomes were among the worst in Europe, *afford* such a highly developed system? Indeed the question might be changed from past to present tense. How can Ireland, by a long chalk the poorest country in the EEC, afford such a highly developed system?

37. This point was called to my attention by Ms Joy Rudd. 38. This point was called to my attention by Liam Ebrill.

A review of the development of the system suggests four kinds of answers.

55

First, Ireland has managed to operate a spartan and frugal school system, especially in the National Schools, which have had historically extremely large classes (some of today's generation of young adults tell of attending classes of 70 or more a generation ago), with poorly paid teachers, in illequipped and poorly-maintained schools. To say this is not a criticism; the alternative, in many cases, was not to have schools at all, or (as elsewhere) to educate only an élite. Thus the average annual public expenditure per National School pupil in 1950, including teacher salaries, school maintenance, etc., was under $\pounds 12$ per pupil. The Investment in Education³⁹ team found in 1963 that 1,979 of 4,779 responding National Schools (or 371/2 per cent) did not have electric current; 3,008 (or 63 per cent) were heated by open fires; and 2,411 of 4,358 (or 55 per cent) did not have flush toilets. One assumes that many of the schools lacking flush toilets were identical with those lacking electric light and using open fires; a quarter to a third of all schools must have had all three characteristics - in 1963! While figures have improved since 1963 (in large part because of publication of the findings of the Investment in Education team), a decade later things had not improved dramatically. The Catholic Primary School Managers' Association routinely surveys its member schools about a number of matters, and in 1973 schools in one Western diocese (not an extreme example) reported no electric light in 18 per cent of the schools, no power points in 22 per cent, no drinking water in 43 per cent, and no flush toilets in 32 per cent.⁴⁰

Impressionistic evidence bears out this finding. As one tours the country, one finds four sorts of quasi-public buildings in virtually every village: churches, banking offices, public houses, and National Schools. This writer, in casually examining these structures, has found few examples where the National School does not suffer by comparison with the other three, in terms of comfort, warmth, cleanliness, colour and facilities.

A romantic image may be conjured up by the thought of children carrying slates to school, crowding around an open turf fire, or reading by lamplight. But the reality may not always have been so picturesque. 'Medical opinion has vehemently condemned the sanitary conditions prevailing in many schools, but it can do no more than make a report to local authority', wrote McElligott in the 1960s. '. . The most unsatisfactory schools are to be found in remote areas from which the population has ebbed, where poverty is endemic and where there is no strong body of public opinion to support a demand for better conditions.' Of the general position, the Report of the Council of Education has this to say: 'The sanitary arrangements and equip-

39. Loc. cit.

40. Material furnished by Rev. Leo Quinlan, Secretary of the CPSMA.

ment in a number of schools are regarded as unsatisfactory, the heating in some cases insufficient, the school surroundings often badly kept, and the furnishing, woodwork and windows defective.^{'41}

Controversy has flared from time to time about reliance on corporal punishment in the National Schools. The present study does not deal with such issues except to note that whatever the ostensible, surface rationale for the practice of corporal punishment, its use has almost certainly been dictated on occasion by the need to maintain strict order, which while sought in classes of all sizes, was essential in very large classes. It is possible that the same may be true of sex-segregation: teachers report that it is easier to maintain order in classes of all boys or all girls than in mixed classes, particularly after the age of nine or ten.

The Investment in Education team in 1962/63 found that 46.5 per cent of all pupils attended classes of 40 or more pupils, and 23.3 per cent attended classes of at least 50.⁴² Very few of these were in rural areas. The Department of Education has published data on class size only in more recent years. In 1968/69, 31 per cent of all National School pupils were reported as attending classes of at least 45 pupils, the largest reporting classification. It is not known how many of these were in classes of 50 or more, but a number must have been, as the average size of the classes in this group was 48.7 pupils. Since then, the percentage attending classes of 45 or more, and the average class size in this group, have continuously declined. They stood, respectively, at 28 per cent and 48.0 pupils in 1971/72, and at only 14 per cent and 46.5 pupils in 1973/74.⁴³ But National School class sizes remain very large.

In rural areas quite another pattern existed. Since at least one Catholic school, and often at least one Church of Ireland school, have existed in each parish, rural areas have often had very small schools. In 1955/56, for example, of 4,871 National Schools in the Republic, 781 were one-teacher schools, and 2,686 were two-teacher schools. In two-teacher schools until fairly recently, a common pattern was to employ one trained principal teacher and one 'junior assistant mistress', an untrained teacher with a significantly lower salary than a trained 'junior assistant teacher'.

In sum, one reason it has been possible to educate so many Irish children for so many years is that the education has been provided at such a low cost, and at such a low level of amenity.

A second reason Ireland has been able to 'afford' its extensive school system has been the very great interest in education on the part of the Catholic Church. It is probably fair to say that no other organised religious body in the world has taken such an interest in schooling. Where it is possible, Catholic teaching requires that the schools be in the hands of the Church. Moreover, 'Catholic children may not attend non-Catholic, neutral,

41. Op. cit., p. 40.
 42. p. 234.
 43. Department of Education.

or mixed schools, that is, those which are open also to non-Catholics.' (Canon Law, 1374) (Rome, 1918). Exceptions are up to the local bishop. Our review of the history of education in Ireland would support the view that this doctrine has been fairly rigorously applied in Ireland. It has meant that State schools, whether under the Irish or the British, could not successfully be introduced; and it has meant that Church resources, as well as Church-encouraged private resources, have gone more into the schools in Ireland than would have been the case in other circumstances.

Unfortunately, the financial contribution of the Church to education, though certainly substantial, is unknown, even within the Church. The Department of Education has gathered and published data only (or mainly) where statistics have arisen as a by-product of administering the system. (Thus there have been copious statistics on examinations, as these are a Departmental function; but far less on in-school matters.) In general, the only financial data available deal with State (or VEC) income and expenditure. Within the Church, historically there has rarely been a separation of parish (or community, in the case of religious orders) accounts, since schooling has been regarded as an inseparable function of the Church (and the order), rather than a side activity or a 'business'.

Some might want to argue that the funds provided by and through the Church really come from the Irish people, and that therefore one need not 'thank' the Church for its 'generosity' to Irish education. This is an argument on which this paper need not take sides. Our point here is only that the Church's great acknowledged interest in education has meant that the economic resources devoted to that activity, and especially non-State resources, have been significantly greater than otherwise would have been the case. And the nature of the Church's interest has led to use of those resources in such a way as to reach every boy and girl in the State, and not only an upper stratum (at least with respect to primary education).

A third way of 'financing' Irish education beyond ordinary expectations has been the contribution made by members of religious orders — brothers, nuns, and clerics — both in terms of actual cash contributions (they have built most of the Secondary Schools, largely from their own resources, for example), and, more important in the contribution of their services. A very substantial fraction of religious in Ireland — estimated at about two-thirds have taken up teaching vocations. Their contributions can take a number of forms:

(1) They return their salaries to their communities. With the qualifications noted earlier,⁴⁴ this means (in convent and monastery schools) 'ploughing back' salaries into the schools themselves. In Secondary Schools, in addition to returning the incremental salary to the school, religious also have traditionally waived their 'school salaries'. The amount returned in Secondary

44. See Chapter 2, p. 32.

Schools also always includes the principal's salary, as lay teachers have not usually been permitted to hold principalships.45

(2) In some schools, religious work more hours per day and days per week than lay teachers. They are available for work after school hours (as with recreation activities) and even after evening meals. This has meant most, apparently, to boarding schools at secondary level. In a number of cases, especially in convent schools, religious have served as wholly unpaid assistant teachers.

We distinguish, in these last few paragraphs, between the interest and contributions of the Church and its congregations, on the one hand, and the personal contributions of religious on the other. Obviously, this is a somewhat arbitrary distinction with which some will quarrel, at least on philosophical grounds. We make the distinction for two reasons. First, Church contributions can rise at a time when teaching vocations are declining, and vice versa; that is, the two can vary independently. And second, the contributions made by religious are comparable to the presence of voluntarism in supporting education, which can be found in many other countries, manifested in forms other than religious vocations.⁴⁶

The economic value of the contributions made by religious is hard to estimate, both because underlying data are hard to come by, and because of methodological problems. A limited effort is made, however, elsewhere in this study.⁴⁷

Fourth, and finally, the Irish system of education has economised by having emphasised in the curriculum subjects with low technical content or which for various reasons have required little in the way of costly equipment. (Teachers have referred to the emphasis on 'chalk-and-talk' schooling.) There are a number of reasons for this emphasis, in both first and second level education. First, as noted earlier, until fairly recently the schools have not been viewed principally in terms of their role in preparing youth for employment; rather, their role has been more moral, intellectual, and religious. Second, also until relatively recently, there has been little employment outside of the banks and insurance companies, the civil service, and other services, for which any special schooling or technical instruction was required. Third, at the time of the First Economic Programme in 1957, it could fairly be said that the Republic of Ireland was, if not a 'pre-industrial State,' clearly a 'non-industrial State'. It is impossible to work out patterns of causation, but it seems likely that the curricular emphasis of the schools was both a cause and a consequence of the economic structure. Fourth, for the reasons set forth above, primary education and most of second-level education has been in the hands of the Church – of bishops and parish

45. Principals' allowances in 1974/75 range from £499 to £2,321, and averaged £1,100. 46. A number of these other aspects of 'voluntarism' seem also to be present in Ireland. Parents are often called upon to contribute repair work or materials and equipment. School personnel – mainly nuns, evidently - call on merchants for reduced prices for services and merchandise.

47. See below, Chapter 5.

priests, in the case of primary education, and also of religious, in some primary and most Secondary Schools. The subject matter emphasised was that in which these persons had *competence*, almost irrespective of the merits of one or another type of curriculum.

Whatever the reasons, and whatever arguments might be offered both for and against the 'arts' emphasis, there can be little doubt that it was virtually the least expensive curriculum that could have been devised. Latin and Greek cost a good deal less to teach than physics and chemistry, and in fact less, as a rule, than modern languages. And religion, Irish, English literature, and music (singing), the main elements in the primary school programme, are also relatively cheap subjects.

The End of Cheap Education

These four sources and resources — the frugal and spartan nature of the system, especially in National Schools; the abiding interest of the Church in education; the personal contributions of religious; and the low technical content of the curriculum — have given Ireland a more developed educational system than the country could have otherwise afforded. They have also vitally influenced the character of Irish education, in nearly every respect. These other respects — curriculum, the role of the Church, corporal punishment, etc. — will have to be debated elsewhere. The principal purpose for noting each of the four in turn is that all of them appear either to be fading or entirely disappearing, to a greater or lesser degree. As they go, Ireland will be left with a highly developed — and expensive — school system. And all of this is happening just as enrolments are about to grow at rates unprecedented in recent decades, and just as costs are on the verge or rising for other reasons.

Our discussion of enrolments - and of population and participation rates. - is postponed to the next chapter. In this section, we will attempt to justify the statement that all four sources of financial ability are, or soon will be, fading or disappearing. Later in this chapter, there will be a discussion of still further reasons for increases in costs.

Spartan and frugal system. There are two reasons why the spartan and often primitive character of schooling, especially at National School level, seems to be doomed. One is that Ireland is a developed and increasingly urban society, in which housing and work accommodation, while still lagging behind need, have progressed sufficiently that school standards have risen accordingly. While few would accuse managers of newly-built National Schools of extravagance, there is a great distance between the newest schools being opened in growing areas, and the older schools being closed in rural areas for want of pupils.

Whatever may be thought of as fit conditions for children attending schools, the recent condition of National Schools has not been regarded as fit or adequate working conditions for *teachers*.⁴⁸ And, as will be noted momentarily, even religious teachers are less willing than heretofore to suffer deprivation.

The second reason is that the complex of conditions which have made possible classrooms with fifty or more pupils are themselves disappearing. These conditions include an old-fashioned authority relationship in the classroom, expected by teachers and accepted by children (still obtaining in rural areas but in decline nationwide); public acceptance (as noted earlier) of the use of corporal punishment; and 'subject-centred' as opposed to 'child-centred' educational approaches. In the emerging social environment of the late 1970s, it is difficult enough to teach a straightforward, subjectcentred curriculum in urban or suburban National Schools, with large classes and without the aid of the cane. It is virtually impossible to do so with a child-centred curriculum. Large v. small classroom size reflects resource issues and limitations, but it is not simply a matter of choice, or of differing philosophies of education. Large class size may require certain objective environmental circumstances, and these appear to be disappearing.

The interest of the Church. Our second source of Ireland's ability to afford its extensive system is the interest and contributions of the Church. A statement that these are waning could and should draw objections, for it is more of a prediction based on straws in the wind than a description of empiricallyderived data. In fact, no one is in a position to state whether the Church's financial contribution, either relative or absolute, is diminishing, increasing, or constant, since no one, inside or outside the Church, is known to have measured it. What can be said is, first, that there are signs of Church acceptance of an increasing State role in education, including some State schools, a development heretofore unheard of; and second, that there are signs of a general interest in a public (as distinct from either a State or Church) voice in educational policy.

The newest and most rapidly growing sectors — the Comprehensive/ Community School movement and the Regional Technical Colleges — make the education system more public. Similarly, the Vocational Schools — for long the one significant public element in the system — have enlarged their scope, offering Intermediate and Leaving Certificate courses. Vocational Schools are increasing their share of second-level education. Thus it is no longer necessary to attend a school maintained by a religious order in order to take a Leaving Certificate.

The new system of boards of management for National Schools⁴⁹ previous-

48. According to McElligott, the 'sharpest controversies' over the managerial system 'have arisen . . . over the material conditions within the schools;' he describes a campaign by INTO, the Irish National Teachers Organisation, in 1964 for improved heating and cleaning. Op. cit., p. 38. And on October 7, 1968, INTO threatened to strike named schools on the following day if conditions in them were not improved immediately. Clarkin, op. cit., p. 196. INTO still in the late 1970s periodically threaten industrial action over the issue of classroom conditions.

49. See Chapter 2 above.

60

IRISH EDUCATIONAL EXPENDITURES – PAST, PRESENT AND FUTURE

61

ly under exclusively ecclesiastical management represents a step in the direction of greater public and parental control of primary schools. There are reports that parishes in middle-class and privileged areas are finding it increasingly difficult to raise funds for new schools, not because of poverty, but because of declining loyalty to the Church in these areas. Pressures for multi-denominational National Schools in Marley Grange, Tallaght, and Dalkey may or may not yield fruit; but they constitute additional evidence for at least interest in, and a tendency toward, a reduced role for the Church.⁵⁰ Ultimately, these trends must mean a lessened Church influence over education and an undiminished Church economic contribution at the same time.

Contributions of the religious. The most dramatic as well as the most easily quantified changes concern the decline in numbers of religious. The decline in new vocations seems to date from the early 1960s. The effect on costs and on the character of education in Ireland - has been delayed, because new teachers are a relatively small percentage of total teachers. Statistical series on religious as a percentage of secondary teachers⁵¹ goes back only as far as 1961, when it stood at approximately 50 per cent. Five years later, in 1966, it still stood at roughly 50 per cent. In absolute numbers, the series 'peaked' in 1970, and then the numbers of religious teachers began its present decline. The decline in religious as a percentage of secondary teachers 'peaked' and began to decline somewhat earlier, in 1967 and 1968, when the introduction of the 'free scheme' led to a rapid expansion in the number of pupils, and hence of teachers. So the decline in the relative importance of religious results both from a decline in their numbers and an increase in the number of lay teachers. By 1974, the percentage religious had fallen to 31 per cent (and even less if Comprehensive and Community School teachers are included, as well they might be). It will continue to fall. Even if there were a gain in number of new vocations, it would take a long time before that gain could show up in religious as a percentage of secondary teachers, once again because new teachers are but a small fraction of total teachers. Taking only Secondary Schools and not Community/Comprehensive Schools, the percentage will stand at 20 per cent in 1986, unless there is a sudden and sharp reversal of existing trends.⁵²

50. As this manuscript is prepared for press, the Dalkey multi-denominational school has been approved in principle by the Minister for Education.

51. Department of Education.

52. This is calculated as follows. During the period 1965-74, the net absolute number of religious teachers in Secondary Schools dropped by .4 times the numbers shown as reaching retirement age, i.e., 65 (*Investment in Education*, Vol. 2, p. 188). Applying this ratio to the number reaching retirement age through 1986 yields an average decline in numbers of religious teachers of 47 per year, 1975 through 1986, or a total of 517. This is subtracted from a stock of religious teachers of 3,739 to yield a predicted number for 1986 of 3,222. In Chapter 4 below (see Table 4.8), it is predicted that there will be 15,841 Secondary School teachers in 1986. This implies religious as a percentage of all Secondary School teachers of 20.34 per cent, or, avoiding the spurious accuracy that suggests, 20 per cent.

The same trends also influence the National Schools, though less dramatically. Our series on religious as a percentage of National School teachers, which goes back to 1956,⁵³ is stable at around 23 per cent through 1966. It falls below 15 per cent in 1974, and looks like falling to about 9 per cent by 1986.⁵⁴

From our previous discussion, it is apparent that a decline in numbers of religious, influences educational costs in a number of ways, especially in the Secondary Schools. To the extent that forgone and returned salaries have been a source of school finance, new sources must be found. While the State takes no *official* notice of these practices, in fact they have profoundly influenced the amount of State support in past years, and their decline must necessarily mean an increase in State funds. To the extent that religious work longer hours in order to provide for extra-curricular activities, school maintenance, and boarders, either the activities must be cut back, or lay personnel must be specifically employed, in addition to regular teaching staff. In some cases, the unavailability of religious for boarding schools may mean a decline in their number, meaning more transport expense for pupils living long distances from day schools.

Three other aspects of this change are worth mention. A decline in the numbers of religious may mean a rise in teacher militancy. Religious teachers' attitudes toward trade unions are inherently ambivalent at best. As members of the order operating the school, they are in a sense both employer and employee. Moreover, many of the benefits won by trade union activity are unavailable to them, at least directly. Increased laicisation of the profession, particularly in the Secondary Schools, cannot but increase the strength of trade unionisation. This effect probably accounts, however, more for salary growth and other gains in the 1960s than in the future. The ASTI demands for full access to principalships, referred to earlier, is an important indicator of this trend.

The second point also pertains to attitudes of religious teachers. Some of them have told us they are no longer willing to make the kind of sacrifices they made as little as five years ago. When wearing clothes repaired over and over again, living without adequate heat, and eating simple and cheap foods permitted children to go to school who otherwise would not have had the

54. The method is the same as that reported in footnote 52, supra. The net decline in absolute numbers of religious in National Schools was .75 of the number shown as reaching retirement age (Investment in Education, Vol. 1, p. 189); applying this to the number reaching retirement age through 1986 yields a decline, 1975-86, of 636 teachers, leaving 1,768 religious teachers in 1986. In Chapter 4 below (see Table 4.8), it is predicted that there will be 19,707 full-time National School teachers in 1986. This implies religious as 8.97 per cent of all National School teachers, rounded up to 9 per cent.

^{53.} The Department of Education series actually begins in 1961; we have been able to estimate the numbers for previous years on the basis of what appears in the early 1960s to be a stable relationship between numbers of religious employed in the National Schools and numbers of religious employed in convent and monastery National Schools. The latter series extends back much further, but wary of the dangers of extrapolation, we have used it back only so far as 1956. In the event, the figure is approximately stable at about 23 per cent for our period of estimated values, 1956-60, and for the first seven years of Department of Education reported values, 1961-66.

opportunity, they were willing to make the sacrifices. But today's Ireland is not an impoverished country, and if such sacrifices only permit reduced tax bills for families and companies, the motivation to sacrifice must be less.

Finally, decline in the numbers of religious teachers represents the sort of situation where a quantitative change becomes a qualitative one. The Irish system of secondary education has *always* been identified with religious orders; but that system is gradually changing into another type of system. The purpose of the present study is to focus on the economic implications of trends whose significance for education and for society is much more than economic. A drastic decline in the role of the religious in Irish secondary education represents not just an explanation for rising costs; it also implies a sharp break in tradition — a disappearance of one type of education system, and the appearance of another.

At what point, for example, does a 'religious school with lay teachers' become merely a 'lay school with a religious principal'? The decline in numbers and proportion of religious had led religious educators to re-examine their role. In February of 1973, a 'Working Party on the Future Involvement of Religious in Education' met in Dublin, under the Chairmanship of Rev. Paul Andrews, S.J. Their report, known as the 'FIRE' Report', was confidential but large sections of the report were published in the press.⁵⁵ The Working Party considered a number of strategies, and recommended that "Religious should begin, in a carefully phased way, to concentrate their forces into a small number of schools, which would generally be of the order of 400 pupils.' The report suggested favourable consideration of co-educational schools as a result of such mergers. The FIRE report has since been superseded, and the strategy indicated has not been pursued; instead, no radical shift in approach is now anticipated. But the report is a reflection of the problem still faced by religious, and one effort to deal with it.

Technical content. Finally, among our four factors, the 'arts' component of Irish education, and in particular the classical component, appears to be in decline, while the technical component is rising. A more costly education, per pupil, is implied.

Table 3.4 illustrates these and a number of other interesting changes in the second-level curriculum. The percentage of boys who take the Leaving Certificate examination in Latin fell from 91 per cent to 79 per cent between 1951 and 1966, and then to 39 per cent in 1972. Increases are shown in technical/scientific subjects – physics, chemistry, and honours-level maths – between 1951 and 1966. These percentages fell again in 1972, because of the introduction (in the examinations, mainly in 1971; in the curriculum, some years earlier) of a number of new, professional and tech-

55. Working Party on the Future Involvement of Religious in Education, FIRE Report, Dublin 1973. Education Times, September 13, 1973, and subsequent issues.

nical subjects, which immediately attracted a large number of students.⁵⁶ Further changes in the curriculum make data for subsequent years not directly comparable,⁵⁷ but it is evident that these patterns continued. For example, the percentage of boys electing to be examined in Latin fell further to 20 per cent by 1973/74. The pattern for girls is somewhat different, and shows weakness in physics, chemistry, and honours maths throughout.⁵⁸ But there is, overall, a decline in the percentage taking Latin, from 47 per cent in 1951 to 29 per cent in 1972 (and, not shown, to 15 per cent in 1973/74), a decline in the percentage taking Domestic Science/Home Economics General⁵⁹, and a large number enrolling for the new technical subjects.

The process indicated in Table 3.4 has really only begun, and its impact on educating costs lies mainly in the future.⁶⁰ While less dramatic, there will be similar changes in instructional costs in the National Schools, as more and more subjects require the use of equipment. It is virtually impossible to predict the amount of implied cost increases, though their importance can scarcely be doubted. Once again, there are important non-economic implications of economic changes. One of these is the following. The introduction of technical subjects, and the use of even fairly simple equipment in traditional subjects inevitably increases the optimum size of a school,⁶¹ the larger is the number of pupils who share the use of a piece of equipment, the lower is the per pupil cost. Future schools must be expected to be larger – probably twice as large - as today's. The impact of rising costs due to the introduction of technical subjects has been softened somewhat by the sharing of facilities among boys' and girls' schools, and among Secondary, Vocational, and sometimes Comprehensive Schools. In the extreme, the 'sharing' takes the form of amalgamation into a Community School, a process that undoubtedly will continue.

THE RECENT PERIOD

The 1960s was a decade of remarkable innovation and institutional

56. The 'number taking new technical subjects' refers to the number of examination papers, rather than the number of pupils. Hence some pupils are counted more than once in the figure. 57. For instance, in addition to 'Physics' and 'Chemistry' there is 'Physics and Chemistry A' and

'Physics and Chemistry B'.

58. This should be interpreted as revealing a lack of equality of opportunity as between boys and girls, rather than a disinclination on the part of girls to study such subjects. For example, until fairly recently, it was virtually impossible for girls to study an honours mathematics course. The lack of equal access to education as between boys and girls requires a separate study in itself.

59. 'Domestic Science/Home Economics General' may be in a sense a 'technical' rather than an 'arts' course, but it evidently has not required much costly equipment. Moreover, from a curriculum standpoint, it does not represent an accommodation of the curriculum to the need for professional and technical training in anticipation of employment.

60. One cannot, of course, be certain that a student studying physics or chemistry necessarily has access to a laboratory.

61. The term 'optimum' is not used in any educational or community sense, but only in a cost sense; and its use does not imply advocacy of large schools.

growth in Irish education. No type of educational institution came out at the end of the '60s as it went in at the beginning, and a number of wholly new kinds of institutions were introduced.

In the late 1950s, in the Government of the Taoiseach, Seán Lemass, it was determined that the Republic should attempt to encourage economic growth, through developing modern manufacturing industry. It was decided to do so through a number of State institutions, employing a form of economic planning. The result was the White Paper published in 1958, A Programme for Economic Expansion, for which T. K. Whitaker, then Secretary of the Department of Finance, and later Governor of the Central Bank and Chancellor of the National University of Ireland, is given major credit.

The new approach required that the educational system be looked at anew, not as in the past in terms of religious, moral, and intellectual training, but in its contribution to the economic system. At the 1961 Washington conference of the OECD (Organisation for Economic Co-operation and Development), Ireland was the first of the member states to volunteer, in the words of P. J. Hillery, then Minister for Education, 'to carry out pilot studies of their educational systems in the light of their probable long-term economic and scientific needs' (Clarkin, 1969, p. 120). The result was the appointment by Dr. Hillery of the team which, over the next three years, produced the historic two-volume report, Investment in Education. 62

The significance of the report is two-fold. First, and most obviously, the report is important on its merits, as a 'long, hard look at the education system' (Clarkin, 1969, p. 121), in which a battery of research devices, particularly direct survey research, were used to unearth and lay out, usually for the first time ever, the most fundamental kinds of information about the system. While the report, in its 800 pages, makes only one specific recommendation,⁶³ this is probably because the force of its factual findings in many instances made specific recommendations superfluous. The report was, and remains, very influential. Perhaps its most important conclusion was its demonstration of an utter lack of correlation between the curriculum and the subsequent careers of pupils.

The other significance of the report is symbolic. The fact of its being commissioned and published, under joint Irish-OECD aegis,64 and especially the fact that it reviewed the whole school system from the standpoint of an

62. The survey team, together with the positions they then held, were: Patrick Lynch, Lecturer in Economics, University College, Dublin, Director of the Survey; William Hyland, Statistics Office, United Nations, New York; Martin O'Donoghue, Lecturer in Economics, Trinity College, Dublin; Padraig O Nuallain, Inspector of Secondary Schools, and, as Secretary to the team, Cathal Mac Gabhann, of the Department of Education.

63. viz, that a 'development unit' be established in the Department of Education. 64. The joint aegis is described as follows at the front of the report: 'This survey was initiated by the Minister for Education in October, 1962. It was organised in co-operation with the Organisation for Economic Co-operation and Development as a project under the Educational Investment and Planning Programme of the Organisation. The Organisation contributed 146,734 French francs toward the cost and provided technical support and information on related developments in member countries'.

'investment', i.e., in terms of its contribution to future production, meant re-orientation of Irish education.

In retrospect, it is not clear how successful the industrial development strategy was. In the second half of the decade of the 1960s, the Irish cconomy boomed; but so did that of Britain, the USA, and Europe in general. A large fraction of the Irish boom seems, moreover, to have been associated with tourism, which dropped sharply, however, at the end of the decade. In the event, whatever the reason the evident buoyance and optimism of the decade carried over into the education arena. We have already stated that the school system, especially at the primary level, was permitted and indeed encouraged to grow and develop in the early period by virtue of a number of unique circumstances and resources which combined to make schooling cheap; and that as these now disappear the State is left with a highly developed, but expensive, school system. With regard to the recent period, we can add that the evidence of rapid economic growth permitted and indeed encouraged costly structural innovation and institutional growth in the 1960s; and that economic reversals of the 1970s mean that the burden of these on the economy is, and in future will be, greater than anticipated.⁶⁵

Without going into detail, let us set out the main outlines of the events of the 1960s, and their carry forward into the 1970s.⁶⁶

- 1962: Investment in Education team appointed.
- 1963: Plans announced for Comprehensive Schools; plans announced for **Regional Technical Colleges.**
- 1964: Scheme of building grants for Secondary Schools announced.
- 1964: Investment in Education, published.
- 1965: Plans announced to close most one- and two-teacher National Schools.
- 1966: First Comprehensive School opened.
- 1967: 'Free scheme' announced; expanded building grants announced for Secondary Schools; Intermediate and Leaving Certificate courses introduced for Vocational Schools; free school transportation scheme introduced.
- 1968: Higher Education Authority established.
- 1969: First five Regional Technical Colleges founded; Higher Education Grants scheme introduced.
- Plans announced for Community Schools; sweeping revision an-1970: nounced of National Schools curriculum, away from 'subject-centred', and toward 'child-centred' approach.
- 1971: Primary Certificate examination abolished.

65. See, for example, the discussion in Chapter 2, above, concerning the windfall transfer of its excess of £31 million from the Education budget to families of Secondary School pupils. 66. See NESC, 1976, op. cit., pp. 30 ff, for a useful account of educational policy in the decade

1965-74, to which this chronology is indebted (see also Horgan, 1973, p. 35 ff).

- 1972: School leaving age raised from 14 to 15; National Council for Education Awards established.
- 1973: First Community School is opened.
- 1974: Last Comprehensive School is opened.
- 1975: New Scheme for expanded aid to National Schools announced for schools adopting Committee of Management system.

According to Sheehan, public education expenditures were 3.05 per cent of Gross National Product in 1961/62, and rose to 6.29 per cent of GNP by $1973/74.^{67}$ This is truly remarkable growth – a doubling, in fact, of the proportion in but twelve years. It is important to note, however, that a commensurate growth of total education expenditures as a percentage of GNP is not necessarily implied. Instead, much of growth in public expenditure occurred through shifting a substantial component of private expenditure, especially in the Secondary Schools, on to the public budget. The sector in the education system to grow most rapidly during the period was secondlevel. The number of Secondary School teachers doubled between 1967 and 1974; and over the same period the number of Vocational teachers increased by 90 per cent. Finally, during the period of the 1960s and early 1970s, teachers finally achieved rough parity, in terms of salaries, with Civil Servants of similar education, experience, and responsibility.

Looked at from another standpoint, the big change in the decade of the 1960s was that the Department of Education came to have a role in the making and handling of policy. Theretofore, because religious bodies (and VECs) operated all the schools, the Education Minister was sometimes referred to, half in jest, as 'Minister Without Portfolio'.68 A conscious objective (partly achieved) was 'to break down the barriers which had hitherto existed between the two systems,' the Secondary and Vocational Schools.69

Most of the innovations and policy changes announced during the period cost but little during that time. Their real significance in budgetary terms lies in the remainder of this century.

THE NEAR FUTURE

This chapter has concentrated on reasons to expect increases in costs in education in the coming decade, and beyond. There are three other foreseeable reasons to expect a rise in per pupil expenditures. Two of these are, in effect, historical 'accidents', at least with respect to timing. The third pertains to all countries at all times, but especially where personal incomes are high and rising.

67. Ibid.

68. Clarkin, op. cit., p. 124. 69. Sean O'Connor, 'Post Primary Education Now and in the Future,' *Studies*, Autumn, 1968, p. 234. O'Connor was then Assistant Secretary, later Secretary, in the Department of Education.

The first is the 'bulge' in the age distribution of second-level teachers. As noted above, the policy changes of the late 1960s brought a spurt in second-level enrolments and a rapid increase in the numbers of second-level teachers. One implication is that the second-level teaching force is unusually young, and a large number of these teachers have progressed relatively little in the salary increments structure. Their *average* salaries are depressed as a consequence; and that average will rise relatively more rapidly over the next decade as a consequence; even apart from agreed increases in the whole structure.⁷⁰

The second cost-increasing event is equal pay. Since the foundation of the State, married men teachers have received higher pay than women or single men.⁷¹ A result of Ireland's accession to the European Economic Community is that there will have to be equal pay. Sooner or later, a single scale will apply, not only to men and women, or to lay and religious, but to married and single. It has already been agreed that equal pay means 'levelling up', i.e., that the 'rate for the job' is the married man's rate, and that other rates will be abolished. Nowhere in the Republic of Ireland is equal pay anywhere as important as in the schools. Elsewhere, its effect can usually be mitigated or avoided altogether by giving women and men nominally different jobs; but in the schools, women and men patently do the same work. Women as a percentage of teachers is high and rising.

These first two points interact with each other. As the teaching force gets older and moves through the increments structure, it will do so at the married man rate. Since the difference between the married man rate and the woman/single man rate rises over time, the 'cost' of equal pay will rise as well.

In the past, a large proportion of women and single men in teaching have been the religious. We have already discussed the decline in numbers of religious, and the consequent laicisation of the teaching force. It is worth noting that equal pay and laicisation are similar in an interesting way. Just as the growth of the school system was, in very considerable degree, especially in the early period, 'subsidised' by the willingness of religious to staff the schools with little personal economic reward, so too was the system subsidised by the availability of lay women teachers and of single men who would work at considerably less pay than the 'rate for the job'. And just as changed social conditions now mean that religious are not available in large numbers to subsidise the schools, so too have changed social conditions meant that women and single men can no longer be required to help support and build the system as in the past.

'Baumol's Disease'

The third cause of rising costs — the productivity-differential effect — 70. This effect was called to my attention by William J. Hyland.

71. Until recently, there was a 'single man' rate in excess of the 'woman' rate. Cf. The Tribunal on Teachers' Salaries, *Report Presented to the Minister for Education* (Dublin: Stationery Office), 1968.

applies in all economies in which there are high and rising incomes. This effect, often called 'Baumol's Disease' after the economist who first set out its main implications, implies that costs of most public expenditures, and certain private ones as well, tend to rise exponentially. Education is especially likely to be affected. Space limitations prevent a full discussion here, and it is possible to review the problem only briefly.⁷²

The central argument can be summarised as follows:

- (1) Productivity output per unit of input, such as labour grows in all economies, but it increases at different rates in the various firms, industries, and sectors of an economy. Productivity tends to grow most rapidly where there are large numbers of uniform products produced, as in manufacturing and agriculture. It tends to grow least rapidly where direct personal services are involved, as in health care, 'live' entertainment, repair services, and most public services, including education.⁷³
- (2) In most western economies, factor incomes (wages, interest, profits) increase at about the same rate throughout the economy, especially over the long run. While from time-to-time there is a change in the wage structure (e.g., some workers' wages overtake others'), these rarely have anything to do with productivity. To simplify, we can assume that all real wages and other factor incomes rise together at a given rate, such as 3 per cent or 3½ per cent.
- (3) Those industries and sectors where productivity rises more rapidly than average can pay these higher wages and other incomes without price increases; indeed, prices can fall. This effect is often concealed by inflation. Where the average price level rises, prices in these industries may rise too, but they will tend to do so more slowly than the average, so that 'real prices' will tend to decline. Those industries and sectors where productivity rises less rapidly than average must raise prices more rapidly than the average in order to pay the higher wages and other incomes.
- (4) The extreme case is where productivity does not rise at all. In this case, prices will rise at the same 3 per cent or 3½ per cent rate at which incomes rise.⁷⁴ This is the case approximated by education and other public services.

72. For a statement of microeconomics of Baumol's Disease, see Tussing and Henning, 1974 (a).

73. While much of the State sector would be affected by Baumol's Disease, the semi-State sector, or State enterprises, to the extent that they produce homogeneous products (turf, electricity, etc.) are likely to be exempt. The differences in productivity discussed here are unlikely to reflect on the persons involved: there is no reason to believe that schoolteachers are less hardworking or innovative, for example, than farmers. The difference arises out of the differences in the types of products produced. Also, productivity as referred to here has little to do with the so-called 'productivity deals' arising out of collective bargaining arrangements, to permit deviations from norms under the National Wages Agreement.

74. This obviously ignores the effect on price of products bought from others; that is, it applies only to the value-added component of price. In the schools, this is a minor qualification.

In this discussion, our definition of "productivity" must be such as to ignore quality changes. In education, it is synonymous with numbers of child-years of schooling provided. We are not arguing that for most purposes this is an appropriate definition of productivity; indeed, for educational purposes, it ignores the most important kinds of matters, such as whether anyone learns anything. But when the issue is cost per pupil per year, then productivity must be defined in terms of child-years of schooling. The implication of Baumol's Disease is that costs per child-year in Ireland will rise at about the same rate as personal incomes rise — probably 3 per cent to 3½ per cent per year — even apart from, and in addition to, the effects of inflation, enrolment growth, reduction in National School class sizes, improved amenity in National Schools, reduced Church contribution, reduced numbers of religious, greater technical content in the curriculum, the 'bulge' in the age distribution of second-level teachers, and equal pay.

It is to be emphasised that changes in the price structure which result from productivity growth differentials represent true changes in relative resource cost and hence in social opportunity cost. They are not a form of 'inflation' which should be adjusted out in calculations by use of some kind of deflator.

Two problems tend generally to conceal the effects of Baumol's Disease. One is that those areas of the economy – services and public expenditure in general – in which productivity seems likely to grow least rapidly are also exactly the same areas where it is most difficult to measure, and often even to define, output in terms of 'quantity units'. In principle, expenditure on police can be decomposed into 'quantity of police service' and 'unit price of police service'; but in practice this is extremely difficult, if not impossible. When expenditures on police, or education, or other State functions tend to rise annually, we are inclined to treat that rise as a matter of governmental discretion – as an increase in scope of public sector activity, ignoring the distinction between price and quantity.

The other problem, one which faces the statistical investigator, is that since the tendency is for price or cost to rise in areas with little or no productivity growth at the same rate as personal incomes grow, it is hard to distinguish this 'supply side' effect from that of high income-elasticity of *demand*. None the less, Baumol's Disease has been shown to be an important and robust factor in explaining public expenditure growth in the USA.

In Ireland, it is virtually an inexorable force. The only escape is to find ways to increase productivity as defined above — using excess capacity in the system, increasing class sizes, introducing a shift system, operating schools in the summer, etc. Experience elsewhere suggests that these devices are of limited practical value.

Baumol's Disease affects any economy where incomes are growing. But it becomes important only where personal incomes are relatively high as well. It is classed as one of the "miseries of growth". Like the other forces

70

discussed in this chapter, high and rising incomes are, then, part of the stage setting in preparation for explosive expenditure growth in the coming decade.

| | 1951 | 1956 | 1961 | 1966 | 1972 |
|--|--|--|-----------------|---------------|-------------|
| Total Number of Boys Taking Leaving | | | | | |
| Certificate Exam. | 2,524 | 3,366 | 4,521 | 6,315 | 11,460 |
| Number Taking Latin As per cent of Total | 2,301 91 | 2,994 89 | 3,903 86 | 4,986 79 | 4,478 39 |
| Number Taking Honours Maths. As per cent of Total | 556 | $\frac{864}{26}$ | 1,215 27 | 2,037 32 | 1,925 17 |
| Number Taking Physics As per cent of Total | $\begin{array}{c} 415\\ 16\end{array}$ | 645 19 | $^{1,139}_{25}$ | $2,102 \\ 33$ | 2,505 22 |
| Number Taking Chemistry As per cent of Total | $516 \\ 20$ | $\begin{array}{c} 692 \\ 21 \end{array}$ | $1,214 \\ 27$ | 1,833 29 | 3,543 31 |
| Number Taking New Technical Subjects ^a As per cent of Total | _ | | | _ | 7,660 67 |
| Total Number of Girls Taking Leaving Certificate Exam. | 2,067 | 3,016 | 4,155 | 6,258 | 12,703 |
| Number Taking Latin As per cent of Total | 96) 47 | 1,261 42 | $1,559 \\ 38$ | 1,687 27 | 3,651 29 |
| Number Taking Honours Maths. As per cent of Total | 8 0+ | $17 \\ 1$ | $56 \\ 1$ | 172 | 361 |
| Number Taking Physics As per cent of Total | 5 0+ | 24 1 | $30 \\ 1$ | 90 1 | 259 2 |
| Number Taking Chemistry As per cent of Total | 21 | 36 1 | 102 | $261 \\ 4$ | 942 7 |
| Number Taking New Technical Subjects ^b As per cent of Total | | _ | | | 6,862 54 |
| Number Taking Domestic Sci./Home Economics General ^c As per cent of Total | $958\\46$ | 1,919 64 | 2,617 63 | 4,063 65 | 6,754 53 |

Table 3.4: Selected Statistics, Leaving Certificate Examinations, 1951-72

*a*Engineering Workshop Theory and Practice; Agricultural Economics; Technical Drawing; Building Construction; Mechanics; Accounting; Business Organisation; Economics; Economic History. *b*Same as for boys (note a), except Engineering Workshop Theory and Practice and Mechanics omitted,

Source: Department of Education, Annual Reports, 1950-51, 1955-56, 1960-61, and Department of Education, Annual Report, Statistical Tables, 1968/69-1971/72.

and Home Economics (Scientific and Social) included (see note c). cEffective 1970, 'Domestic Science' was discontinued as a subject for the Leaving Certificate examina-tion; in 1971, two new subjects were introduced, viz. 'Home Economics General', a course for future 'homemakers', which has been included above as a replacement for 'Domestic Science', and 'Home Economics (Scientific and Social)', a technical, pre-professional course which has been included above among 'New Technical Subjects'.
Chapter 4

Enrolments and Expenditures: The Coming Explosion

The purpose of this chapter is to examine present and recent data on school enrolments and expenditures in the Republic of Ireland, not only so that we can understand them better, but so that we can use this information, along with other data and methods, to make as accurate a set of forecasts as we can concerning near-future (through 1986) trends. Doing so is a matter of some importance, as our prediction is that both enrolments and per pupil public expenditures will grow extremely rapidly, if not explosively, over the coming decade. These two predictions are quite independent of each other. If only one or the other of them is correct, i.e., if only cost but not enrolments grow, or vice versa, it will be extremely difficult for the State to find the resources necessary to maintain existing educational commitments. If both are true, then there will be a crisis.

We discuss, in turn, enrolments and expenditures.

ENROLMENTS

To estimate future school enrolments, it is necessary first to estimate future school-age population, and then to predict school participation rates by age or age group. 'School participation rates' are the fraction or percentage of an age group attending school. Enrolments are simply the product of the population of the relevant age group and its respective participation rate.

There are two distinct steps in the process, then: prediction of population, and prediction of participation rates. A third step, prediction of actual enrolments, is but mechanical. After taking these steps, below, we will essay a plausible allocation of the predicted enrolments by type of school. Then we will attempt to estimate the number of teachers required by our plausible allocation. Finally (before turning to expenditures), we will compare the resulting figures with those released in 1976 by the National Economic and Social Council (NESC, 1976, Report No. 18).

Population

Population prediction is a highly inexact science at best. Demographers generally have not predicted the swings in birth rates, especially in developed countries, which have brought such variability in population growth rates, and have usually contented themselves with working out their implications. In Ireland, the problem is vastly more complex, because population has been profoundly influenced by emigration/immigration patterns, and in recent years these have been complex and difficult to understand. These migration patterns are not directly measured at all, and net (not gross) flows must be inferred from periodic censuses. Because these migration patterns influence the age/sex distribution of the population, they in turn influence birth and death rates, which are consequently more unstable (and still harder to predict) than they are in countries where migration is less marked.

Net emigration in the 1950s and early 1960s was unusually high, even for a country with an emigration tradition. According to estimates made by Brendan M. Walsh, there was net emigration of 469,800 persons in the twelve years 1950-1961, and 304,900 of these were in the seven years 1954-60. An extremely large percentage of first and recent entrants into the labour market (i.e., those in middle and late teen and through their twenties) left the country, in some communities, virtually the entire class of school leavers emigrated shortly after the end of the school year. In more recent years, net emigration has fallen to approximately zero, and indeed there is evidence in one or two years of net immigration. According to estimates made by J. G. Hughes of The Economic and Social Research Institute, net emigration fell from 15,000 in 1968 to 5,400 in 1969 and 4,700 in 1970, and then 'turned the corner' to net immigration of 700 in 1971; 1,800 in 1972; 300 in 1973; and 7,100 in 1974. These are provisional estimates and are subject to later alteration; our debt to J. G. Hughes is acknowledged.

This is a phenomenon that appears to be widely misunderstood by observers, who frequently take the fall in net emigration either to represent a disappearance of emigration *per se*, or who infer that the Republic is now able to provide an adequate number of jobs for its young people, who no longer must leave in order to find employment. Both of these are fallacious. There is very considerable evidence that emigration continues, and in large numbers; and that it is offset, more or less, by immigration of persons with quite different demographic and labour market characteristics, as we will discuss in more detail. See Hughes and Walsh, (1976) where data on gross flows for 1971 are discussed in detail. Similarly, while net emigration had declined very considerably, since, say, 1961, the number of persons employed in the Republic has not increased at all. It was approximately the same in April, 1961, as in April, 1974, and fell subsequent to that because of the world-wide depression. (Tussing, 1976a).

There are two explanations for why net emigration could have fallen to nil, and below, while employment failed to rise. First, there has been a very considerable expansion in school enrolments. In 1966, there were 91,100 persons 15 years and older in schools (at all levels); by 1971, this had risen to 125,000 persons, and by 1974, 139,300 persons, up 48,200 over 1966. Our own conservative estimate, published elsewhere, is that the important 1967/68 education policy changes discussed in Chapter 3 – namely, the 'free' secondary education scheme, by which most second-level schools were induced to forgo their fees, and addition of the Intermediate and Leaving Certificate courses to Vocational Schools – themselves accounted for at least an added 19,600 persons in school in 1974 (Tussing, 1976b).

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

The other explanation is understood qualitatively but its dimensions are not known. This is the fact that zero net emigration is achieved by offsetting flows of a number of groups. It appears that those leaving are still dominated by the same group as in the period of more massive outflow – those in their late teens through middle or late twenties. Those returning, at least on a net basis, seem to include persons in their thirties, and school-age children.⁷⁵ In addition, there is a substantial flow into the country of retired persons drawing pensions.⁷⁶ There are two types of evidence for these patterns. One is the change in numbers in relevant age cohorts between censuses, especially when actuarial death tables are taken into account. These reveal a clear pattern of net migration by age group. The CSO has provided us with a memorandum in which they predict future population trends. In commenting on a set of population projections in which annual net emigration of 5,000 persons is assumed to recur after 1976, the CSO states the following assumption regarding age distribution: 'For each five year step . . . the loss of population through emigration in each age group was decided by distributing the total . . , on the basis of the patterns observed between 1966 and 1971 both census years. This concentrates the loss through emigration in the age range 15 to 30 years, with gains in some other age groups, namely, under 15 years, 30-39 years, and 65-74 years, representing the return of families with young children as well as of elderly persons on retirement'. CSO also used the same patterns in estimating the age allocation of intercensal estimates of population for the years since 1971. The second type of evidence is school enrolment. For the ages in which school enrolment is approximately 100 per cent, i.e., ages 6 through 13, annual reported figures on school enrolment may be taken as estimates of the size of the relevant age groups. These figures have also shown regular annual increases in recent years – increases that can only be explained by child immigration.

A pattern is suggested: young people leave in order to find acceptable employment, which is usually taken up in Britain. Men and women born in Ireland marry, either in Ireland or abroad. They return some years after leaving, with school-age children. If a typical returning family consists of husband, wife, and two children (a wholly conjectural but plausible assumption), and if only the husband is planning to seek work, then every four persons leaving the country in search of jobs are matched, to achieve zero net emigration, with four persons entering the country, only one of whom is seeking a job. This means a net reduction of three in the labour force. With regard to retired persons, every person leaving in search of a job who is matched by a returning retired person means a net reduction of one in the Irish labour force. Thus the apparent emigration pattern has become one which maintains a fairly stable labour force by appropriate *structural* 75. According to Hughes and Walsh (op. cit., Tables 3 and 4), most of the immigrating adults were born in the Republic of Ireland; most of the children were not. The implication is that the children 76. Department of Education, Annual Statistical Reports, Dublin: Stationery Office. population adjustments. The consequence is also to worsen the dependency ratio (NESC, 1975 Report No. 7).

It is hazardous to attempt to project these patterns into the future. It is not really known who the returning emigrants are, or why they are returning. Some are evidently the same persons who left, as teenagers, in such large numbers in the 1950s and early 1960s. If so, one would expect return flows to decline in future. Large numbers can only return if large numbers leave at a previous time; and the numbers now emigrating, while presumably significant, probably do not compare with the outflows of the earlier period. There is also some indication⁷⁷ that a substantial portion of the returning emigrants come from the Six Counties of Northern Ireland, and that they have been seeking homes in the Republic since about 1970. If this is the case, one would hesitate to predict that such flows would continue indefinitely. In short, net emigration may resume in future more because of a decline in return flows than because of a rise in gross outflows.

Population data used in the Republic are based on periodic censuses (the most recent of which was taken in 1971), and on intercensal estimates. The Census of Population Division, Central Statistics Office (CSO), is responsible for both. In their intercensal estimates, the CSO evidently has underestimated the numbers of school-age children, particularly those under the age of 15, immigrating to the Republic, in the years 1973, 1974, and 1975. At least the numbers of children reported as in school by the Department of Education exceeded the numbers estimated as being in the country by the CSO. Beginning with 1973, CSO estimates of population by age began to fall short of reported school enrolments, by age. The former are estimates, based on a method of first estimating net population movements, and the second, allocating them among age groups on the basis of past experience (as revealed in census returns). The latter are actual counts, as reported by schools to the Department of Education. While there may be counting errors in the enrolment figures, there is no evidence that these are important. Enrolment data have been collected in the same way, by age, for more than a decade prior to 1973, and until 1973 (when official data began to indicate participation rates in excess of 100 per cent in some ages), enrolment and estimated population maintained a stable and predictable relationship with enrolment. Between 1972 and 1973, and between 1973 and 1974, reported enrolment gains exceeded estimated population growth in the school-aged population under 15 years of age. There are three possible interpretations of this error. One is that the CSO underestimated the number of families with small children returning to the Republic. This would mean that there would also be an underestimate of immigration in certain adult age groups, and an

77. While it is estimated that there is positive net *im*migration in recent years, presumably from Britain, British data show net *im*migration from the Republic of Ireland to England and Wales in some of the same years (HMSO, 1976). Assuming both sets of data to be correct, the difference could be accounted for by large migration to the Republic from Northern Ireland and/or Scotland. Assuming that Scotland is not an important factor in recent years, the implied rate of migration from Northern Ireland to the Republic is several thousand persons per year.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

underestimate of total immigration. A second interpretation would be that they estimated the number of returning families correctly but underestimated the number of children; this would mean that the underestimate among school-age children would be the only error, and would be equivalent to the error in the total. The third interpretation would be that CSO had correctly estimated the total number of immigrants but had incorrectly allocated these among age groups, so that an underestimate of immigrating children would be matched by a corresponding overestimate of returning adults. This last is the interpretation of the population office. For purposes of the present study, it does not really matter, since we are concerned alone with children. The two projections used here are by Walsh and the CSO. Walsh made two projections, one based on zero net emigration, the other based on a net emigration of 5,000 per year, between 1976 and 1986. These will be referred to hereinafter as Walsh High and Walsh Low, respectively. The Census of Population Office of the Central Statistics Office (CSO) has provided us with a copy of a projection, with accompanying commentary, which they made on an assumption of zero net emigration through 1976, and a net emigration of 5.000 per year thereafter. This is referred to hereinafter as CSO-5000. The Industrial Development Authority commissioned county-by-county population projection to be done by the CSO on an assumption of zero emigration and the IDA has shared a copy of this projection with us. This is referred to hereinafter as CSO-0.

Population projections freely available are used in this study, except that three kinds of adjustments had to be made.

First, the forecasts begin with understated base-year figures, because of the above-mentioned underestimate of schoolchildren. An adjustment is made to correct for this underestimate.⁷⁸

Second, births have been somewhat less than predicted in the depression year of 1975.⁷⁹ Our assumption is that this is a temporary phenomenon, related to economic conditions, and will affect births only in 1975, 1976, 1977, 1978 and 1979; we have reduced the predicted numbers of births accordingly.⁸⁰

Third, the available projections made very critical and implausible assumptions about future migration patterns. In two cases, these assumptions are that

. . . where zero net migration is assumed . . . net migration in all age groups is set equal to zero, although it is possible or even probable that

78. See Appendix 4.A for details.

79. A preliminary version of some aspects of this chapter was presented as a public paper, 'The Coming Explosion in Enrolments and Expenditures', in a seminar sponsored by the Association for Democracy in Education, in Dublin, June 27, 1976. Mr. Brendan Herlihy, of An Comhairle na n-Ospideal, who read press accounts of the seminar paper, was kind enough to call these recent short-falls in births as compared with predictions to my attention. As a consequence, predicted population and enrolments are somewhat less in the present paper than in the earlier, seminar version.

80. See Appendix 4.A for details.

a zero balance would be arrived at by a set of cancelling in and out-flows at different ages (NESC, 1975 Report No. 5).⁸¹

If it is assumed that zero net emigration for the State also implies zero net emigration for each age group, the only influence over time on the numbers in an age cohort is death. If today's immigration is some lagged function of previous emigration, then the assumption of zero net emigration is a complex and not a simple one, and would persist only as a consequence of some rather implausible or at least fortuitous circumstances. However plausible the assumption of zero net emigration for the State may be, the assumption of zero net emigration for each age group is inconsistent with all experience, and appears quite implausible. If past patterns persist into the future, then projections based on an assumption of zero net emigration in each age group will (1) understate future school-age population 15-64, and hence lead to over-prediction of the size of the future labour force; (3) under-predict the number of persons over the age of 65 in future; and (4) as a consequence of the first three, underestimate future dependency ratios.

While we must reject the assumption of nil net migration by age group, it, unfortunately, does not appear correct simply to assume that the migration patterns discussed above, and observed in recent years, will persist unaltered for another decade. As noted earlier, it seems more likely that return immigration will gradually decline in coming years. Instead, we assume that recent patterns of net emigration/immigration by year of age persist into 1976 only, and then that these patterns 'taper off' more or less in a linear fashion to zero net emigration for each age group by 1981, except that we expect small numbers of teenagers to continue to emigrate in a net sense. That is, if in 1976 it were assumed that there were net immigration of 5,000 in an age group, this would be reduced to 4,000 in 1977, 3,000 in 1978, etc., and to nil in 1981. Then it is assumed that there is no further net immigration in any age group, and reduced net emigration of teenagers between 1981 and 1986.⁸²

It must be emphasised that these are extremely conservative assumptions. Rather than projecting today's patterns into the future, we assume that today's highly significant pattern of child immigration will *not* persist, while assuming that emigration of older (13+) children will continue. While our assumptions are arbitrary, they seem more defensible, on intuitive grounds, than any others. But it must be pointed out that if today's immigration patterns *do* persist, our forecasts of youth population will fall short by a considerable margin, with alarming consequences. Thus demographers and analysts should be alert for evidence on this score throughout the remainder of the 1970s and 1980s.

Population estimates and forecasts for 1974, 1976, 1981 and 1986 are set

81. NESC, (1975) op.cit., p. 20. The same assumption is made in CSO-0.

82. See Appendix 4.A for the assumed pattern of child migration.

| ge | | 1974 | | | 1976 | • | •••• | al a A air | 1981 | • • | t s și G | 1986 |
|----------|---|-------|--------|-----------|-------|-----------|-------|---------------|-------|------------------|----------------|------|
| 0 - 4 | 1.1 | 337.0 | | | 340.5 | × | | | 373.2 | | | 408. |
| 4 ் | | 64.9 | | | 68.0 | | 1. A. | . 1 | 69.0 | | | 80.0 |
| 5 | 1 A A A A A A A A A A A A A A A A A A A | 62.1 | 1. A. | | 68.4 | (1, j, k) | | | 68.1 | 1.1 | | 79.4 |
| 6 | | 64.4 | | 5. j. j. | 65.1 | | . ' | | 68.1 | • | * ² | 77. |
| 7 | | 64.5 | | 1.1 | 62.9 | | | | 69.2 | 2 | | 75. |
| 8 . | | 65.8 | | | 65.4 | | | | 68.8 | | | 72. |
| 9 🦯 🖓 | · · · · | 66.3 | 2 | | 65.5 | | | | 68.6 | • . ¹ | | 69. |
|) | | 65.2 | | | 67.1 | | × . | | 69.4 | | | 68. |
| 1 142 | | 63.9 | | | 67.0 | | 2 | · · · | 66.3 | | | 68. |
| 2 | 1 | 62.6 | . • | | 65.6 | | | | 64.1 | | | 69. |
| 3 | | 61.8 | | | 63.9 | | 1. | | 66.4 | | | 68. |
| | · · · · · | 59.7 | 1 de 1 | Sec. 2. 1 | 62.1 | | , | | 66.1 | | | 68. |
| j | | 58.0 | | | 61.2 | | | | 66.8 | | | 69. |
| 5 | · · · · · | 57.6 | i. | | 59.1 | | | | 66.7 | | · . | 66. |
| | | 56.9 | 1.1 | | 57.2 | | | | 64.7 | | | 63. |
| 3 1 | | 54.7 | | | 56.7 | | | | 63.0 | | | 65. |
| | | 55.2 | 1.1 | | 56.2 | | | | 61.1 | | | 65. |
|) – 24 | | 233.1 | | | 242.8 | · ' | | | 264.0 | | | 297. |

Table 4.1: Estimated population, 1974 and predicted population, 1976, 1981, and 1986: selected ages and age groups

forth in Table 4.1. Figures for 1974 are based on CSO estimates, except that adjustments have been made, as discussed earlier, to account for a larger number of children at some ages than CSO had estimated, and except further that CSO figures, which are given for five-year age groups, have been interpolated according to past patterns to provide year-of-age estimates.⁸³ The remaining years -1976, 1981 and 1986 - are forecasts, as discussed earlier. In Table 4.2 the forecast for 1986 is compared with other available forecasts. The Walsh high and low are taken from his NESC paper.⁸⁴ NESC has released revised estimates of predicted population aged 15-19, on the basis of the high child immigration, on which NESC has based its own enrolment forecasts (NESC, 1976 Report No. 18). CSO-5000, already referred to, does not assume zero net migration within age groups; CSO-0 does. It will be seen that our own estimates lie between the Walsh/NESC high and low figures for 0-4s, 5-9s, and 10-14s; exceed somewhat the NESC high for 15-19s; and lie between the Walsh high and low (and between the CSO-5000 and CSO-0 as well) for 20-24s. Thus it seems unnecessary to discuss in any great detail points of difference between our estimates and others'. Predictions for 0-4s and 5-9s are the most conjectural, since none of these children have been born as this is written. Our estimate for the 10-14s tends toward the low end of the Walsh/NESC spectrum, because of the recent fall in births, which we have assumed is attributable to the depression, and which NESC may not have taken into account. Only in the 15-19 year range does our prediction lie

83. Estimated population by year of age, 14-19, based on such interpolations as these, have been published in Tussing, op. cit., p. 291. 84. NESC (1975) op. cit.

beyond any others' high estimate: it is slightly higher than the NESC revised version of Walsh's prediction. NESC adjusts Walsh's figures to take account of child immigration, but assumes an earlier end to it than we do, assuming that the '1975-76 recession and the very high unemployment levels should act as a brake on immigration'.⁸⁵ Our predictions for the 20-24 year age group lie between Walsh's original low and high.

Table 4.2: Comparisons of forecasts of school-age population, 1986 by Walsh, NESC, CSO andPresent Study (x 1,000)

| Age | Present | Walsh | Walsh | NESC | NESC | CSO | CSO | Keating | Keating |
|---|---|---|--|---|---|---|---|---|---|
| Group | Study | Low | High | Low | High | 5000 | 0 | Low | High |
| $0-4 \\ 5-9 \\ 10-14 \\ 15-19 \\ 20-24$ | 408.0 373.0 342.8 330.4 297.8 | $367.0 \\ 357.3 \\ 342.2 \\ 310.0 \\ 290.0$ | $\begin{array}{c} 457.1 \\ 403.6 \\ 354.7 \\ 310.0 \\ 320.0 \end{array}$ | $367.0 \\ 357.3 \\ 342.2 \\ 319.9 \\ -$ | $457.1 \\ 403.6 \\ 354.7 \\ 328.4 \\ -$ | 409.6 382.0 346.4 309.7 274.8 | $\begin{array}{r} 410.6\\ 370.0\\ 329.4\\ 312.3\\ 314.2\end{array}$ | 352.5 347.7 354.7 330.1 289.3 | 356.8 349.7 356.7 335.1 298.3 |

Sources: Present study: see text. Walsh: Brendan M. Walsh, Population and Employment Projections, 1971-86, NESC Paper No. 5 (Dublin: Stationery Office), 1975; the 'low' assumes zero net emigration 1971-76, and net emigration of 5,000 p.a. 1977-86; while the 'high' assumes zero net emigration 1971-86. NESC: Population Projections 1971-86; The Implications for Education, NESC Paper No. 18 (Dublin: Stationery Office), 1976: these are essentially NESC staff re-workings of Walsh's figures to take account of child immigration, and are in fact the same as Walsh's for most age groups. CSO-5000: This is a population projection carried out by the Central Statistics Office, and provided to us in a continued pattern of net emigration and immigration by age groups (see text). CSO-0: This is a population projection carried out for the Industrial Development Authority by the CSO, and provided to us by the IDA; it assumes no net emigration either for the State as whole or in any age group. Keating: W. Keating, Central Statistics Office, 'An Analysis of Recent Demographic Trends with Population Projections for the Years 1981 and 1986', paper read to Irish Statistical and Social Inquiry Society, March 3, 1977. The 'high' and 'low' estimates assume net emigration of nil and 25,000 respectively, over the five years, 1981-1986.

Most recently, W. Keating of the Central Statistics Office has developed a new set of population projections for 1981 and 1986. While these were published too late for us to take appropriate account of them in this paper, Keating's high and low 1986 predictions are given in Table 4.2. It will be apparent that Keating believes that the drop in births is a permanent, not a transitory, phenomenon, and that child immigration will persist longer than we have predicted. Accordingly, his high estimate shows 23,300 fewer 5-9 year-olds than does our single prediction. On the other hand, his low estimate for 10-14 year-olds is 11,900 higher than our single estimate. (Our predictions for 15-19 and 20-24 lie between his low and high estimates.)

^{85.} NESC, p. 10. NESC state as known fact what must be regarded as assumption or speculation. For example, they state, 'During 1971-75 there was substantial net immigration in the 25-39 age group, consisting mainly of trained and skilled workers drawn from the very large pool of young emigrants of the later 1950s and early 1960s. The children of these immigratis have caused substantial net immigration in the 0-14 age group'. While the pattern described is plausible and conforms with our own views, and while NESC notes in a footnote (p. 9) that the statement quoted 'must remain tentative for some time,' the NESC has given no evidence to support its statements about immigration during 1971-75, nor has it indicated where such evidence might come from in the future. The NESC does a disservice in presenting such speculation as fact.

Participation Rates

It is as difficult if not more so to predict participation rates as it is to predict population growth. Participation rates reflect, as well as dictate, State education policy; and they reflect, as well as influence, general economic conditions. Moreover, in an economy as small and as open as Ireland's, their growth is in large part a necessary consequence of the growth in participation rates elsewhere, a point upon which we will elaborate below.

Typically, nations' education systems grow in 'blocks', generally corresponding with levels or sub-levels within the system. First there will be a period mainly of primary level growth, followed in turn by growth in other levels or time blocks (e.g., junior cycle, or the equivalent, then senior cycle, or the equivalent), and concluded by third-level growth. Of course, all the while first level participation rates are growing, rates will also be growing at other levels. But evidently, the most rapid growth in other levels' participation rates begins after prior levels' growth rates have slowed.

In Ireland virtually all growth in first level participation rates occurred prior to partition and the establishment of Saorstat Eireann, and the rate stands today at approximately 100 per cent except for the infant division. We take to be a datum that participation rates for ages 6-13 will continue indefinitely at approximately 100 per cent. Growth of second level rates was given a boost by the education policy changes of 1968-69, but these rates were already in the process of growth, and the contribution of the policy changes was only marginal.⁸⁶ Third level participation rates evidently have not yet reached their main period of growth.

Growth in participation rates since 1963, in all school ages except for 6-13, is shown in Figures IV.1 through IV.8. (These graphs also show predicted future growth. These predictions, which are shown by broken lines at the right of the several graphs, will be discussed later in this section, and may be ignored for the moment.) Figure IV.1 shows participation rates for those aged 4 and 5, i.e., those in the infants' division, to be nearing saturation. The main period of growth has already taken place. The same may be said for 14-year olds, as is shown in Figure IV.2. The minimum school leaving age was raised from 13 to 14 in 1972, but Figure IV.2 shows that the principal growth at this level had already occurred by then: in 1971, 84.7 per cent of 14-year olds were in full-time schooling. As is shown in Figures IV.3, IV.4 and IV.5, major growth in participation rates of 15, 16, and 17-year olds occurred in the 1960s and 1970s. Indeed, rates for 15-year olds neared their saturation point in this period. Little growth has yet taken place, however, in rates for ages 18, 19 and 20+, i.e., in third level education, as is indicated by Figures IV.6, IV.7, and IV.8. Evidently, the period of most rapid growth is yet to come.

One cannot predict patterns of future participation rate growth simply by linear extrapolation of existing trends. Indeed, participation rates rarely grow

86. We have estimated that 19,600 of 306,800 enrolled in second-level in 1974, or 6.4 per cent of the total, are attributable to the policy changes. Tussing, op cit.



IRISH EDUCATIONAL EXPENDITURES – PAST, PRESENT AND FUTURE



82

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

a in a lar



IRISH EDUCATIONAL EXPENDITURES – PAST, PRESENT AND FUTURE





IRISH EDUCATIONAL EXPENDITURES - PAST, PRESENT AND FUTURE



86

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

Sources and notes follow Figure IV.8.





SOURCES AND NOTES FOR FIGURE IV.1 THROUGH IV.8

Sources: Actual participation rates: Enrolments, from Department of Euducation data; Population, from Central Statistics Office; Predicted participation rates; ESRI predictions are our own, see text; NESC predictions are from *Population Projections 1971-86: The Implications for Education*, National Economic and Social Council Report No. 18 (Dublin: Stationery Office), 1976.

Notes:

- (a) A small number of three-year olds is included in the numerator; the denominator is population aged 4 and 5.
- (b) NESC predicts a participation rate of 66.7 per cent for 4-year olds and 100 per cent for 5-year olds. Assuming the 4- and 5-year-old cohorts to be the same size, the two can be averaged to yield a combined participation rate of 83.4 per cent. For these ages, the NESC 'low' and 'high' have the same values.
- (c) NESC predicts participation rates for first and second levels only; third level students are omitted in their calculations. Our predictions, on the other hand, are for all levels. This difference affects comparisons only for 17- and 18-year olds. In Figure IV.5 we have tried to indicate what we believe NESC would have predicted, had they covered all three level's. Since the NESC low prediction is always the same as the 1974 participation rate, it is a simple matter to show a NESC-method low. The NESC-method high depends on what one assumes the NESC would have concluded about 17-year olds attending third level institutions. In 1974, these were 8 per cent of all 17-year olds. If one assumes the 8 per cent will remain constant through 1986, then one can add it to the 46 per cent that NESC predicts will be the 17-year old participation rate in first and second levels (i.e., by 20 per cent), then one can add the resulting 9.6 per cent to the same 46 per cent to yield a total of 55.6 per cent. We have taken 55 per cent, approximately the mid-point between these estimates, to be the NESC-method high rediction. As with all other predictions, the NESC high is reached in 1978, and then held constant through 1986.
- (d) The method is the same described in note (c). The low holds the 1974 rate constant. For the high, we sum the NESC value for 18-year old participation rate, 1986, for first and second levels (i.e., 16 per cent) with two versions of what might be the NESC prediction for 18-year olds in third level. In the first version, the present (i.e., 9 per cent) rate is maintained; in the second, it rises at the same rate as the first- and second-level rate (i.e., to 11 per cent). The two versions yield, respectively, 25 per cent and 27 per cent; we have taken 26 per cent to be the NESC-method high.
- (e) NESC makes no projections for 19-year olds.
- (f) All pupils aged 20 and over, as a percentage of the population aged 20 through 24.
- (g) NESC makes no projections for this age group.

in a linear fashion for very long. Normally, they exhibit typical S-shaped growth curve patterns. That is, they grow first at a relatively moderate, but increasing, rate. They then usually show a period of fairly rapid growth, in the midst of which growth participation rates begins to occur at a decreasing rate. The last phase is a return once again to growth at a moderate rate, or perhaps cessation of growth altogether. It can be taken that in Ireland, first-level age groups are in the third phase; that second-level age groups through age 15 are also in the third phase; that in the rest of second level, growth is currently in the second, or rapid, phase; and that in third level, growth is in the first, or moderate phase.

If the foregoing (which is admittedly impressionistic and speculative) is correct, then the main questions are, when will growth in second-level rates (ages 16 and above) slow down, and when will third level growth commence? Some idea of participation rate possibilities and patterns can be gleaned from international comparisons. Table 4.3 shows rates for a number of countries, including the United Kingdom and, separately, England-cum-Wales and Northern Ireland. Also shown are official forecasts of UK and Northern Ireland participation rates for 1980 and 1985. It will be noted that postcompulsory, second-level age-group projections for the UK and Northern Ireland show rates for 1985 lower than reported rates achieved for the Republic of Ireland for 1974.

Tables 4.4 and 4.5 show data recently published by the OECD (Organisation for Economic Co-operation and Development) on years of age that are, for most member countries, beyond the age of compulsory school attendance. While the data are somewhat out of date (they cover the period 1966-70), in general they bear out what was said above. In Table 4.4, which shows the ages roughly corresponding with senior cycle or senior high school, Irish rates compare extremely favourably with those of other OECD nations. Of 21 nations whose rates are reported in Table 4.4, Ireland's are eighth highest for 15 and 17-year olds, and seventh highest for 16 and 18-year olds. Irish participation rates are higher throughout than those of all EEC nations except Denmark. And the Irish rate used in these comparisons is that for 1967, prior to the boost in participation rates resulting from the policy changes of 1967/68.

By contrast, Table 4.5, which reflects the ages normally associated with third level, show Ireland in a less favourable light. The Irish participation rate is fourteenth of nineteen for nineteen-year olds, tied (with Italy) for seventeenth of nineteen for 20-year olds, and eighteenth of eighteen for 21-year olds. Most of the EEC countries are ahead of Ireland in all three years of age. The main significance is not that Ireland has done poorly, but that Irish rates are liable to grow fairly soon, and fairly rapidly, in third-level education.

As noted earlier, international comparisons are unusually important for Ireland. The reason is as follows. Growth in enrolment and in participation rates arises mainly from growth in demand for education, which is strongly associated with growth in per capita income. While to an extent this growth in demand reflects rising ability of a population to pay for their children's education, it also arises out of increases in so-called educational requirementsthe skills, abilities, certificates and qualifications required by employers throughout the economy. Increases in educational requirements in turn derive mainly from changes in the occupational structure (i.e., a relative increase in those kinds of jobs requiring more schooling), and from upgrading of educational requirements within occupations (i.e., those with particular job titles, such as journalist, secondary school teacher or bank manager, require more schooling than previously). For a small and open economy, these changes in general reflect changes in the larger world (or European) economy. Without a drastic reversal of present international economic policies, Ireland's future educational requirements are essentially beyond her control. As these grow

| 4 | | 170 A | Y T | D-11 | T | E | E | England- | | | United | Kingdom | | | Not | rthern Irelar | ıd | | | Repu | ıblic of Ire | land | , <u></u> , <u>_</u> , |
|--|---|--|--|--|---------------|----------------------|--------------|-------------------|---|---|--------------|--------------------|--------------------|---|---|---|--|--|--------------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|--------------------------------------|
| Age | 1960 | 1973 | 1970 | 1970 ISBN 1970 | Japan 1970 | 1962 | 1968 | 1973 | 1970 | 1973 | 1975 | 1980 | 1985 | 1970 | 1973 | 1975 | 1980 | 1985 | 1970 | 1974 | 1976 | 1981 | 1986 |
| 2 3 4 | | | | | | | | | } 11.2 | } | 20.0 | 27.8 | 29.3 | 16.5 | 21.7 | 26.5 | 30.0 | 33.4 | 76.7 | 84.7 | 86.0 | 90.0 | 90.0 |
| $\left. \begin{array}{c} 5\\6\\7\\8\\9 \end{array} \right\}$ | $\left. \begin{array}{c} 80.7\\ 99.6\end{array} \right\}$ | 92.5 99.1 | 65.4 95.3 97.8 98.5 98.4 | 94.6 98.3 98.7 98.8 | | | | 96.6 | 98.8 | } 100.7 ^d | | | | 99.7 | 97.9 | 97.9 | 99.8 | 99.8 | 98.9 | 99.9 | 99.5 | 99.5 | 99.5 |
| 11 12 13 14 15 16 17 | 99.5 } 97.8 } 82.5 } | 99.2 97.5 88.3 | 98.4 98.2 97.7 64.0 40.7 34.2 32.2 | 99.1 99.1 99.0 98.8 92.1 86.1 73.2 | 64.4 | <pre>99.0 49.7</pre> | 99.0 57.5 | 25.8 ^c | $ \begin{array}{c} 100.4^{d} \\ 100.8^{d} \\ 69.0 \\ 34.9 \\ 19.8 \end{array} $ | $\left.\right\} 100.1^{d}$ 73.0 36.6 21.0 | 51.7 23.4 | 57.6 29.8 | 63.1 \ 35.3 | 99.5 58.9 33.0 23.6 | 97.9 63.5 34.9 23.9 | 97.8 48.5 25.4 | 999.7 54.7 28.9 c | 99.7 60.8 32.5 | 96.7 85.7 71.2 56.9 39.3 | 99.4 92.5 77.6 60.2 43.1 | 99.5 94.0 80.0 63.0 48.0 | 99.5 97.0 26.0 67.0 52.0 | 99.5 97.0 92.0 75.0 55.0 |
| 18 19 } 20+a | ^{38.4} } 20.7 | $\left. \begin{array}{c} 42.9\\ 30.0 \end{array} \right\}$ | 32.2 ^b | 49.7 28.0 14.6 | 11.9 |) 9.0 |) 13.7 | J | 6.7 ^{ef} | 7.1 ^{ef} | 7.9 | 10.1 ^{ef} | 12.1 ^{ef} | $\left.\right\}^{10.3^{1}}_{1.7^{\text{fg}}}$ | $ \right\}^{10.2^{1}}_{1.4^{\text{fg}}} $ | $\left.\begin{array}{c}11.3^{1}\\1.5^{\mathrm{fg}}\end{array}\right.$ | $\left.\right\}_{1.7^{\mathrm{fg}}}^{13.8^{\mathrm{I}}}$ | $\left.\right\}_{2.0^{\mathrm{fg}}}^{16.4^{\mathrm{1}}}$ | 20.5 12.1 7.2 | 22.4 12.5 6.9 | 24.0 14.0 8.0 | 29.0 18.0 12.0 | 34.0 22.0 15.0 |

Table 4.3: School participation rates, first, second, and third levels, actual and predicted, selected places and years

(a) Rate figures on population aged 20-24. Notes:

•

.

.

(b) 18-24.

(c) 15+ figured on population aged 15-19.
(d) Reported school enrolment exceeds estimated population: see source.
(e) 18+ figured on population aged 18.

(f) Excludes 3rd level.

(g) Figured on population aged 19.

Sources: United Nations Demographic Yearbooks, 1971, 1973, UK Social Trends, 1974: Education Statistics for the United Kingdom, 1973; Northern Ireland Education Statistics, Feb. 1974; Rep. of Ireland Dept. of Education, Tuarascail, various years, with population estimates, see text, and participation rate projections, from present study.

| ·(per cent of population at | each age enrolle | d in acade | emic year | beginnin | g at date | cited) |
|-----------------------------|------------------|------------|-----------|----------|-----------|--------|
| | | 15 | 16 | 17 | 18 | 15-18 |
| Austria | 1969 | 54.8 | 32.6 | 23.6 | 16.4 | 31.9 |
| Belgium | 1966 | 75.1 | 61.3 | 47.0 | 33.2 | 54.2 |
| Denmark | 1970 | 85.2 | 66.8 | 31.8 | 23.2 | 51.7 |
| Finland | 1967 | (59.2) | 51.9 | 43.5 | • 35.2 | 47.4 |
| France | 1970. | 80.5 | 62.6 | 45.1 | 29.1 | 54.3 |
| Germany (a) | 1969 | 54.9 | 30.8 | 20.4 | 15.7 | 30.5 |
| Greece | 1969 | 56.8 | 49.1 | 45.8 | 26.1 | 44.7 |
| Iceland | | n.a. | n.a. | n.a. | n.a. | |
| Ireland | 1967 | 82.4 | 64.3 | 46.5 | 31.8 | 56.3 |
| Italy | 1966 | 42.1 | 33.6 | 27.4 | 20.2 | 30.8 |
| Luxembourg | 1970 | 67.9 | 56.5 | 42.3 | 31.2 | 49.5 |
| Netherlands | 1970 | 79.7 | 60.6 | 41.5 | 28.4 | 52.5 |
| Norway (b) | 1970 | 94.2 | 74.6 | 59.8 | 46.5 | 68.9 |
| Portugal | 1970 | 30.1 | 25.4 | 22.0 | 20.4 | 24.5 |
| Spain | 1970 | 35.0 | 29.6 | 22.8 | 19.0 | 26.7 |
| Sweden | 1972 | 96.7 | 74.0 | 60.8 | 40.8 | 68.1 |
| Switzerland | 1970 | 94.6 | 61.5 | 52.7 | 27.4 | 61.4 |
| Turkey | 1968 | n.a. | n.a. | n.a. | n.a. | (21.1) |
| United Kingdom | 1970 | 73.0 | 41.5 | 26.2 | 17.6 | 39.4 |
| Yugoslavia | 1968 | n.a. | n.a. | n.a. | n.a. | 36.6 |
| Australia | 1971 | 81.5 | 54.2 | 37.2 | 23.6 | 49.4 |
| Canada (c) | 1970 | 98.0 | 89.1 | 77.2 | 45.8 | 78.1 |
| Japan | 1970 | (83.8) | (79.0) | (74.8) | (29.9) | (65.8) |
| United States (d) | 1970 | 97.7 | 93.5 | 86.2 | 53.8 | 82.9 |

Table 4.4: Full-time school participation rates, first, second and third levels, ages 15-18,OECD countries, selected years

(a) Excludes part-time compulsory vocational schools.

(b) Includes a very minor number of part-time pupils.

(c) Excludes enrolments in trade schools and schools for handicapped.

(d) Including part-time students in higher education.

() Estimates.

Source: The Educational Situation in OECD Countries, Organisation for Economic Cooperation and Development, Paris, 1974.

in Europe, they must necessarily grow in Ireland; and where (as in third level) they lag behind the rest of Europe, they will be required to rise more rapidly than the European average. For similar reasons, the curricular structure of Irish education has been changing, and must continue to change, in the direction of more technical relative to arts subjects, and more 'ability' relative to 'knowledge'. We are not able to employ vigorous forecasting models here. Like NESC⁸⁷ we have had to use 'less formal methods of forecasting' based on the principles described above.

87. Op. cit., p. 14.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

Our predictions for the pattern of growth in participation rates for the period through 1976 are shown in Figures IV through IV, where they are labelled 'ESRI'. These predictions, which are repeated in Table 4.6, show participation rates continuing to rise at rates not much different from those of recent years in ages 15, 16 and 17; and at faster rates than have been experienced recently in ages 18, 19 and 20+. For age 14, they reflect effective

| Table 4.5: Full-time school participation rates, first, second and third levels, ages | 19-24, |
|---|--------|
| OECD countries, selected years | |

| | | 19 | 20 | 21 | 22 | 23 | 24 | 25 |
|----------------|------|--------------------|---------|----------|----------|------|------|-------|
| Austria | 1969 | 10.4 | 8.7 | 7.5 | 6.7 | 5.6 | 4.5 | 4.4 |
| Belgium | 1966 | 24.4 | 16.9 | 13.0 | 7.3 | 5.2 | 3.4 | n.a. |
| Denmark | 1970 | 18.0 | 14.7 | 14.8 | 14.6 | 12.5 | 10.9 | (9.3) |
| Finland | 1967 | 26.9 | 20.6 | 19.0 | 17.9 | 16.7 | 14.2 | n.a. |
| France | 1970 | 19.5 | 16.2 | 15.6 | 12.6 | 9.7 | 6.5 | 6.9 |
| Germany | 1968 | 12.6 | 10.6 | 9.5 | 9.1 | 7.5 | 6.9 | 5.8 |
| Greece | 1969 | (20.7) | (20.1) | (18.6) | (7.9) | 6.7 | 4.9 | 3.7 |
| Iceland | | n.a. | `n.a. ' | `n.a. ' | n.a. | n.a. | n.a. | n.a. |
| Ireland | 1967 | 16.7 | 10.0 | 6.8 | n.a. | n.a. | n.a. | n.a. |
| Italy | 1966 | 15.5 | 10.0 | 7.0 | 4.5 | 3.0 | 1.5 | n.a. |
| Luxembourg | 1970 | 22.2 | 14.9 | 11.8 | 9.2 | 6.8 | 4.5 | n.a. |
| Netherlands | 1970 | 20.6 | 15.1 | 11.1 | 8.4 | 6.7 | 5.1 | 4.6 |
| Norway | 1970 | 31.7 | 22.6 | 19.5 | 17.1 | 14.2 | 11.4 | n.a. |
| Portugal | 1970 | 16.3 | 13.0 | 9.5 | 7.5 | 6.4 | 5.4 | 5.7 |
| Spain | 1970 | 16.7 | 19.0 | 11.0 | 6.7 | 6.4 | 6.6 | 5.8 |
| Sweden | 1972 | 24.0 | 19.8 | 17.9 | 16.5 | 13.9 | 11.8 | 9.6 |
| Switzerland | | n.a. | n.a. | n.a. | n.a. | n.a. | n.a. | n.a. |
| Turkey | | n.a. | n.a. | n.a. | n.a. | n.a. | n.a, | n.a. |
| United Kingdom | 1970 | 14.3 | 12.4 | <u> </u> | 4. | 6 | | n.a. |
| Australia | | n.a. | n.a. | n.a. | n.a. | n.a. | n.a. | n.a. |
| Canada (a) | 1970 | 31.6 | 25.9 | 21.0 | 12.6 | 8.3 | 5.5 | 4.3 |
| Japan | 1970 | (22.0) | (13.8) | (13.7) | n.a. | n.a. | n.a. | n.a. |
| United States | 1970 | `40.9 [′] | (35.4) | (28.3) | ` | 14.9 | | n.a. |

(a) Excludes enrolment in trade schools and schools for handicapped.

() Estimates.

N.B.—It should be noted that not all the people in these age-groups are in higher education. In Sweden, a substantial number of 19 and 20 year olds are still at secondary school.

Source: The Educational Situation in OECD Countries, Organisation for Economic Cooperation and Development, Paris, 1974.

saturation in the very near future.⁸⁸ And for ages 4 and 5. they show continued asymptotic growth toward effective saturation.

Figures IV.1 through IV.4 also show the pattern of participation rate growth forecast recently by the National Economic and Social Council (NESC), and Figures IV.5 and IV.6 show the pattern we believe the NESC method would have shown for ages 17 and 18, had NESC figures included third- as well as first- and second-level rates.⁸⁹ Since NESC have recently published their own forecasts of a number of the series forecast here, and since in some respects NESC have arrived at different conclusions, it seems useful to compare NESC forecasts with our own (as indeed we have done above, in the section dealing with population), so that readers can arrive at independent conclusions. There can be no 'right' or 'correct' forecast of the future. Those who offer predictions can be asked to set out their methods and assumptions, so that readers can follow and make their own assessments, and, moreover, so that as conditions change, readers can adjust forecasts appropriately.

NESC presents two versions of predicted patterns of participation rate growth, a low and a high version: 'The first is that 1974 participation rates remain constant', thus giving the pattern we have labelled NESC-low in the graphs, with participation rate growth ceasing in 1974; and 'the second ... based on rising participation rates in the 1974-78 period and constant rates thereafter', thus giving the pattern we have labelled NESC-high, with participation rates continuing to grow (very rapidly, in most cases, as an inspection of the graphs will reveal) until 1978, and with growth ceasing thereafter. These are extremely conservative assumptions. Indeed, they are difficult to understand. There is no basis in the experience of Ireland, or, so far as we know, any other country on which to conclude that growth in participation rates might stop dead at all years of age at the same time. Moreover, there is no evidence that the demand for places in second level education in Ireland has stopped, or is about to stop, growing. These predictions are sufficiently dubious and sufficiently important to cast doubt upon projections of enrolments, teacher needs, and other variables flowing from them, in what is otherwise an extremely valuable publication.

Enrolment Numbers

It is simple arithmetic to combine the population projections with the predicted future participation rates to produce enrolment forecasts. Enrolments and school participation rates for 1966, 1971 and 1974 are shown in Table 4.6, together with predicted figures for 1981 and 1986, and 'predicted' or 'estimated' figures for 1976. The overall result is a predicted increase in school enrolment at all levels of 188,700 or 23 per cent, in a period of only

^{88.} NESC (1976, P.15) suggest a participation rate of 97 per cent for 14-year olds, as the minimum school leaving age was recently raised from 13 to 14, and 97 per cent was the rate for 13-year olds just prior to the raising of the school leaving age. We adopt this assumption in the present paper. 89. See the table of sources and notes following Figures IV.1 through IV.8; note (c).

| | 196 | 6 | 197 | /1 | 19 | 74 | 19 | 76 | 19 | 81 | 198 | 36 |
|--------------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|----------------|
| Age | Enrol- ment | Part'n rate |
| 4-5 ^a | 81.8 | .662 | 100.2 | .793 | 107.6 | .847 | 117.3 | .86 | 123.4 | .90 | 143.5 | .90 |
| 6-12 | 392.2 | .959 | 430.4 | .989 | 452.3 | .999 | 456.3 | .995 | 472.1 | .995 | 496.6 | .995 |
| 13 | 55.5 | .954 | 56.7 | .968 | 61.4 | .994 | 63.6 | .995 | 66.1 | .995 | 68.5 | .995 |
| 14 | 38.0 | .687 | 49.3 | .847 | 55.2 | .925 | 58.4 | .94 | 64.1 | .97 | 66.5 | .97 |
| 15 | 30.2 | .542 | 39.8 | .707 | 45.0 | .775 | 49.0 | .80 | 57.4 | .86 | 63.4 | .92 |
| 16 | 21.3 | .390 | 31.1 | .550 | 34.8 | .604 | 37.2 | .63 | 44.7 | .67 | 49.5 | .75 |
| 17 | 14.6 | .273 | 21.4 | .394 | 24.5 | .431 | 27.5 | .48 | 33.6 | .52 | 35.0 | .55 |
| 18 | 7.3 | .147 | 10.9 | .208 | 12.2 | .224 | 13.6 | .24 | 18.3 | .29 | 22.4 | .34 |
| 19 | 4.4 | .096 | 5.8 | .121 | 6.9 | .125 | 7.9 | .14 | 11.0 | .18 | 14.4 | .22 |
| 20+ ^b | 13.3 | .072 | 16.0 | .075 | 15.9 | .068 | 19.4 | .08 | 31.7 | .12 | 44.7 | .15 |
| Total ^C | 658.6 | .602 | 761.6 | .657 | 815.8 | .671 | 850.2 | .68 | 922.4 | .69 | 1,004.5 | .705 |

Table 4.6: Enrolments and school participation rates, 1966, 1971, and 1974, and predicted, 1976, 1981, and 1986, by age

(a) Participation rates are figured on the population aged 4 and 5; enrolment data also include a small number of 3-year olds.
(b) Participation rates are figured on the population aged 20 through 24.
(c) Participation rates are figured on the population aged 4 through 24.

12 years. This breaks into growth of 80,200 (14 per cent) in the group aged through 12; 36,800 (23 per cent) in the group aged 13 through 15; 35,400 (50 per cent) in the group aged 16 through 18; and 36,300 (159 per cent) in the group aged 19 and over.

Of the 188,700 enrolment growth forecast, 113,480 would occur with no increase in participation rates over 1974—a most implausible eventuality. This includes 71,750 of the 80,200 forecast for the group aged through 12; 23,720 of the 36,800 forecast for the group aged 13 through 15; 12,440 of the 35,400 forecast for the group aged 16 through 18; and 5,570 of the 36,300 forecast for the group aged 19 and over.

Thus, most of the growth forecast is attributed to indicated population growth. It is to be recalled that if present child immigration patterns persist, school age population, and hence enrolment, would be still higher. The participation rate increases forecast are quite moderate: for the school-age group as a whole (ages 4 through 24), and combined rate is predicted to rise from 67.1 per cent in 1974 to 70.5 per cent in 1986. The largest increases forecast are at the older ages, where numbers (both population and enrolment) are lower.

The results, then, are not very sensitive to individual assumptions. Even a sharp change in one or two assumptions will not yield very different results.

For example, employing the Keating population projections referred to earlier (see Table 4.2) reduces enrolment projections by only 13,400 (high population projection) to 24,600 (low population projection). The resulting overall enrolment forecasts of 991,100 and 979,900 are 21.5 per cent and 20.1 per cent over the 1974 levels, respectively-still extremely rapid growth.⁹⁰

The forecasts are based on the implicit assumption that State and Church education policy will evolve in future in a manner similar to that of the past. Any sharp break in practices can obviously affect enrolments. They are also based on the assumption that an adequate number of school places will, in fact, be available. This assumption may be more questionable, but our role is seen as estimating the numbers of places that will be *required*, rather than the number that will be *provided*.

The predictions are in a sense firmer than the widely-publicised forecasts of numbers of jobs needed through 1986 to achieve full employment with minimal or zero net emigration.⁹¹ If the requisite number of jobs is not created, large-scale emigration may resume, if not to Britain then to continental EEC countries, Australia, Canada, and elsewhere. But emigration, while it may in a sense 'solve' the job-creation problem, will not solve the problem of providing for nearly 200,000 new places in first, second and third-level institutions. On the contrary, a resumption of large-scale emigration would make the school finance problem still worse. It is difficult to see

90. See Appendix 4B.

91. e.g., Ŵalsh, (1975) op. cit.

how the nation will be able to pay for the implied growth in school expenditures if the 300,000 or so new jobs are not forthcoming, and national income does not grow enough to pay the cost. On the other hand, if the jobs are created, this enrolment growth will appear far less daunting than the stark statistics suggest.

There is one more point of relationship between the enrolment forecasts and the number of new jobs needed. If school participation rates do not rise as rapidly as forecast here, then still more young people will be looking for jobs. If labour force participation rates turn out in the future to be lower than forecast, then still larger numbers will be looking for school places. Data from the 1971 census of population indicate that while school and labour force participation rates of young boys and girls vary considerably with age in the 14 through 19 year range, the *sum* of the two rates is remarkably constant at around 90 per cent.⁹² There is strong likelihood, then, that anyone in this age range who either does not seek or does not find a school place will seek a job instead, or emigrate. There is really no escape from the problems generated by a rapidly growing youth population. One can hardly hope that young people will neither seek jobs, *nor* go to school, nor emigrate.⁹³

Pupils, Types of School and Teachers

The rest of this chapter consists not so much of forecasts as of illustrations. While it is possible with a small number of assumptions to attempt to forecast population growth and changes in participation rates, exactly how future enrolments might be sorted out among types of schools—e.g., secondary, community, vocational, etc.—is a question so influenced by policy that prediction is virtually impossible. Instead, we have attempted to show one plausible pattern according to which the forecast enrolment might be distributed; and then, on the basis of that pattern, we have attempted to forecast the numbers of teachers that would be needed in each type of school. In the second half of this chapter, where costs and expenditures are considered, we will employ this same conjectural or illustrative pattern in our calculations.

Table 4.7 shows a predicted allocation of enrolments among Private Primary Schools, National Schools, Secondary (into which Comprehensive and Community were merged), and Vocational, according to past patterns and recent trends. There is also an allocation to third level, but no allocation to institutions within that level; and in the remainder of the chapter, our focus is narrowed mainly to first and second levels. On the basis of recent trends,

92. Tussing, (1976b) op. cit., p. 298. The sum of school and labour force participation rates in 1971 was as follows, for the years of age indicated: 14, 93.9 per cent; 15, 89.2 per cent; 16, 89.1 per cent; 17, 88.4 per cent; 18, 87.0 per cent; and 19, 89.0 per cent. 93. The Radical Economists Group argue that the current and impending population growth is a

93. The Radical Economists Group argue that the current and impending population growth is a potential boon as well as a burden, in that it implies a surge in home market demand, which potentially can create jobs in the industries that satisfy such demands. If this demand can be channelled into Irish industry and employment, then the neccessary jobs will be created; if it cannot, then the jobs will not be created and the demand surge will not occur anyway. Radical Economists Group, 'The Irish Economy—A New Strategy for Development', pamphlet No. 1 (Dublin: Radical Economists Group), 1976.

| Age | Private primary | National schools | Total 1st level | Jr cycle second'y ^a | Jr cycle vocat'l | Total jr cycle | Sr cycle second'y ^a | Sr cycle vocat'l | Total sr cycle | Total 2nd level | 3rd level | Total |
|------------------|--------------------|---------------------|--------------------|-----------------------------------|---------------------|-------------------|-----------------------------------|---------------------|-------------------|--------------------|-----------|----------------------|
| 4-5 | 4% | 86% | 90% | 0% | 0% | 0% | 0% | 0% | 0% | 0% | 0% | 90% |
| 6-12 | 4 | 90.5 | 94.5 | 4 | 1 | 5 | 0 | 0 | 0 | 0 | 0 | 99.5 |
| 13 | 0 | 20 | 20 | 57.5 | 22 | 80 | 0 | 0 | 0 | 80 | 0 | 99.5 |
| 14 | 0 | 4 | 4 | 65 | 28 | 93 | 0 | 0 | 0 | 93 | 0 | 97 |
| 15 | 0 | 0 | 0 | 51 | 22 | 73 | 16 | 3 | 19 | 92 | 0 | 92 |
| 16 | 0 | 0 | 0 | 10 | 9 | 19 | 46 | 10 | 56 | 75 | 0 | 75 |
| 17 | 0 | 0 | 0 | 1 | 2 | 3 | 38 | 10 | 48 | 51 | 4 | 55 |
| 18 | 0 | 0 | 0 | 0 | 0 | 0 | 11 | 8 | 19 | 19 | 15 | 34 |
| 19 | 0 | · 0 | 0 | 0 | 0 | 0 | 2 | 2 | 4 | 4 | 18 | 22 |
| 20+ ^b | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 1 | 1 | 1 | 14 | 15 |
| Enrolmen | ts (x 1,000) | | | | | | | | | | | |
| 4-5 | 6.4 | 137.1 | 143.5 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 143.5 |
| 6-12 | 20.0 | 451.6 | 471.6 | 20.0 | 5.0 | 25.0 | 0 | 0 | 0 | 25.0 | 0 | 496.6 |
| 13 | 0 | 13.8 | 13.8 | 39.6 | 15.1 | 54.7 | 0 | 0 | 0 | 54.7 | 0 | 68.5 |
| 14 | 0 | 2.7 | 2.7 | 44.6 | 19.2 | 63.8 | 0 | 0 | 0 | 63.8 | 0 | 66.5 |
| 15 | 0 | 0 | 0 | 35.3 | 15.2 | 50.5 | 11.1 | 2.1 | 13.2 | 63.7 | 0 | 63.7 |
| 16 | 0 | 0 | 0 | 6.6 | 5.9 | 12.5 | 30.4 | 6.6 | 37.0 | 49.5 | 0 | 49.5 |
| 17 | 0 | 0 | 0 | 0.6 | 1.3 | 1.9 | 24.2 | 6.4 | 30.6 | 32.5 | 2.5 | 35.0 |
| 18 | 0 | 0 | 0 | 0 | 0 | 0 | 7.2 | 5.3 | 12.5 | 12.5 | 9.9 | 22.4 |
| 19 | 0 | 0 | 0 | 0 | 0 | 0 | 1.3 | 1.3 | 2.6 | 2.6 | 11.8 | 14.4 |
| 20+ | 0 | 0 | 0 | 0 | 0 | 0 | 0 | 3.0 | 3.0 | 3.0 | 41.7 | 44.7 |
| Total | 26.4 | 605.2 | 631.6 | 146.7 | 61.7 [°] | 208.4 | 74.2 | 24.7 | 98.9 | 307.3 | 65.9 | 1.004.5 ^c |
| 1974 Tota | al 23.3 | 521.1 | 544.4 | 124.3 | 48.9 | 173.2 | 54.4 | 14.1 | 68.5 | 241.7 | 29.7 | 814.8 |
| Growth | | | | | | | | | | | | 0 1 1 0 |
| year | 1.0% | 1.2% | 1.2% | 1.3% | 1.9% | 1.5% | 2.6% | 4.7% | 3.0% | 2.0% | 6.8% | 1.7% |

Table 4.7: Predicted school participation rates and enrolments, 1986, by type of school and age

Notes:

(a) Secondary, Secondary Tops, Comprehensive and Community.

(b) Enrolment 20 and over as per cent of population aged 20-24.(c) Detail does not add to total because of rounding.

IRISH EDUCATIONAL EXPENDITURES -PAST, PRESENT AND FUTURE Vocational Schools are expected to grow somewhat faster than other secondlevel institutions combined. But this prediction is based on present policy, which evidently consists of re-building or expanding Secondary and Vocational Schools as these become obsolete or too small. If the policy reverts to one of amalgamating such schools into Community Schools on the occasion of their requiring building grants, then the future pattern of growth will look rather different.

Table 4.8 shows the required number of teachers associated with the enrolment allocations reported in Table 4.7. The table shows an overall increase in the number of full-time teachers of 20.6 per cent between 1974 and 1986. We expect a very slight reduction in average class size in National Schools;⁹⁴ no change in class size in Private Primary Schools; an increase in average class size in Secondary/Comprehensive/Community; and a slight decline in average class size in Vocational Schools. In second-level schools, average class size in senior cycle has been considerably smaller than that in junior cycle. Senior cycle is expected to grow faster than junior cycle in the coming decade. This structural change will have opposing effects on average class size. Senior cycle classes will undoubtedly tend to grow in average size, which *ceteris paribus*, should increase second-level class size on average. But the increasing weight attached to the senior side, expected still to be smaller than the junior cycle, should reduce second-class size on average.⁹⁵

Of the two tendencies, the former will be stronger in Secondary/Comprehensive/Community, and the latter in Vocational Schools.

Table 4.8 indicates an increase of 28.7 per cent in the numbers of parttime teachers. It is possible and even likely that full-time teachers will increase, especially in Vocational Schools, somewhat more rapidly than Table 4.8 indicates, and that fewer part-time teachers will be used. Budget amounts will not be significantly influenced by this change.

Our prediction that 19,707 full-time National School teachers will be needed stands midway between the NESC low and high forecasts of 18,500 and 20,590 based on an assumed 30:1 pupil teacher ratio.⁹⁶ Our prediction of a need for 19,040 second-level teachers stands well outside the NESC range of 16,403 to 17,774.⁹⁷ This difference represents a cumulation of all the differences already referred to: our estimate for population aged 15-19, slightly higher than the NESC high estimate (see Table 4.2); our assumption that participation rates will continue to rise, as against NESC's assumption that they will cease growing at latest in 1978; and finally, differences in predicted pupil-teacher ratios. Our weighted second-level average pupil-teacher

94. The required number of teachers is calculated from recent incremental pupil-teacher ratios.
See Appendix 4.C.
95. The required number of teachers is calculated on the basis of equations estimated by multiple

95. The required number of teachers is calculated on the basis of equations estimated by multiple regression techniques (ordinary least-squares). See Appendix 4.C.

96. NESC, (1976, pp. 26, 29).

97. Ibid., (1976) p. 29.

| Type of school | Number of full-time teachers 1974 | Number of part-time teachers 1974 | Pupils per full-time teacher 1974 | Number of full-time teachers 1986 | Number of part-time teachers 1986 | Pupils per full-time teacher 1986 | Increase in number of full-time teachers, 1974-86 |
|------------------------------|--|--|--|--|--|--|---|
| National Schools | 16,592 | n.a. | 31.4 | 19,707 | n.a. | 30.7 | 3,115 |
| Private Primaries | 1,163 ^a | n.a. | 20.0^{a} | 1,320 | n.a. | 20.0 | 157 |
| Secondary, Comprehensive and | | | | | | | |
| Community | 10,794 | 2,786 | 16.6 | 12,558 | 3,283 | 17.6 | 1,764 |
| Vocational | 4,675 | 3,010 | 13.5 | 6,482 | 4,134 | 13.3 | 1,847 |
| Total | 33,224 | 5,796 | | 40,067 | 7,457 | · · | 6,843 |

Table 4.8: Required number of teachers, and students per full-time teacher, 1974 and 1986

(a) These values were obtained from our own survey of Private Primary Schools; see Chapter 5.

.....

1.1

¥.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

ratio is 16:1; NESC's is 17:2.⁹⁸ A rise in pupil-teacher ratios may indeed occur over the coming decade in response to the fiscal pressures discussed here (though this is not the reason for the difference between the two sets of forecasts); our forecasts, it will be recalled, are based on an assumption of the persistence of today's policies and trends into the future. Such forecasts are unlikely to 'come true', and are not really meant to, since their purpose is to state the nature of the problem, rather than to anticipate both the problem and its solution.⁹⁹

EXPENDITURES

Real public expenditures on first and second level education are likely approximately to double over the twelve-year period 1974-1986. While we have not carried out any supporting analysis on third level, it appears likely on the basis of evidence available that third level expenditures are more than likely to double. Public education expenditures, then, unless there are shifts and reversals in policy, are likely to comprise a rapidly rising share of public expenditures and of Gross National Product.

In this section, we build upon the enrolment allocations and teacher needs (Tables 4.7 and 4.8) presented in the previous section, by attaching 'price tags' to those enrolments and teacher needs. The resulting expenditure forecast is obviously related only to the one, single set of enrolment allocations and teacher needs set forth, and different though equally plausible patterns would yield different expenditure forecasts. However, changing these assumptions, while they might affect expenditure patterns, would be significantly unlikely to affect the totals, or even the main sub-totals.

A first step in predicting future public education expenditures is obviously to set forth the amounts and types of *present* public education expenditures. It is a remarkable thing that this is not done, in any public or semi-public document dealing either with public expenditures or with education.¹⁰⁰ While information on public education expenditures does exist, it must be drawn together from a variety of sources, and the person doing so is left to his or her own devices in interpreting data sources and their comparability.

Our interpretation, as indicated in Tables 4.9 and 4.10, is that current public expenditure on first and second level education was approximately

^{98.} This is based on our calculations from NESC figures. Their stated assumptions for pupil-teacher ratios are Secondary, 19:1; Vocational, 13:1; and Comprehensive/Community, 17:1. *Ibid.*, (1976) p. 28. 99. Thus our forecasts are comparable to those of the 'Ghost of Christmas Yet to Come' in Dickens' *Christmas Carel*: they are 'visions of what might be', rather than 'visions of what will be'.

^{99, 1} nus our torecasts are comparaote to those or the Gnost or Christmas Yet to Come' in Dickens' Christmas Carol: they are 'visions of what might be', rather than 'visions of what will be'. 100. This has not been done, for example, in the Department of Education's Tuarascail, which while it includes budget data on Primary, Secondary, Comprehensive/Community and Vocational divisions, does so in ways which are not wholly comparable with each other; and they omit administrative overhead, and public third level expenditure. Sheehan (op. cit. NESC, 1976 No. 12) includes a table (p. 9) on public expenditure on first and second level education, with the same four divisions; but he does not provide a total, and does not include third level. We are advised that the Department of Education is moving toward such reporting in future issues of its Tuarascail.

| Type of school | Salaries ^a | Pensions a superann'n | nd other current ^b | Sub total col. 1+2+3 | Alloc overhead ^d | cated Transport ^c | Total current | Current per pupil | Capital | Grand total |
|---|-----------------------|--------------------------|----------------------------------|--------------------------|--------------------------------|---------------------------------|--------------------------|----------------------|---------------------|--------------------------|
| 1st Level: National schools | 37,022 | 5,547 | 1,249 | 43,818 | 1,521 | 2,149 | 47,487 | 91 | 6,697 ^e | 54,184 |
| 2nd Level: Total Secondary schools ^f Comprehensive and | 31,713 18,359 | 325 } 42 | 15,672 | 47,710 | 1,679 | 2,928 | 52,317 } 32,282 | 211 | 8,347 2,116 | 60,664 |
| Community Vocational ^g Reformatory and Industrial schools | 1,384 11,970 | 283 |) 6,375 | 18,628 |) 644 10 | 763 | 20,035 |) 289 | 4,791 1,440 | 21,475 |
| Sub total: 1st and 2nd Levels | (11.a.) | (11.a.) | (11.a.) | 92,079 | 3,219 | ,,077 | 570 100.374 | Ļ | 15,303 | 829 |
| <i>3rd Level: Total</i> Regional technical | | | | 14,742 | 451 | 0 | 15,193 | 512 | 4,128 | 19,321 |
| colleges ¹¹ Training colleges ¹ Universities | | | | 1,489 1,666 10,130 | 54 54 343 | 0 0 0 | 1,543 1,720 10,473 | | 2,048 0 2,080 | 3,501 1,720 12,553 |
| expenditures ^j | | | | 1,457 | | | 1,457 | | | 1,457 |
| Total, All Levels | | | | 106,221 | 3,661 | 5,077 | 115,559 | | 19,431 | 134,998 |

Table 4.9: Public education expenditures, year ending March 31, 1974 (x £1,000), current (1974) prices

Sources and Notes apply to Tables 4.9 and 4.10.

Sources; Except as indicated in notes, below, data are from Appropriation Accounts 1973-74, as follows: 1st level, vote 28; secondary, vote 29, except that secondary tops are added in from vote 28; comprehensive and community, vote 29; vocational, vote 30; reformatory and industrial schools, vote 31; 3rd level, vote 32. Enrolment data used in estimating per-pupil expenditures from Department of Education.

Notes:

(a) For National Schools, includes capitation grants to "capitation convent and monastery schools"; for Secondary Schools, includes only incremental but not school salaries; for Vocational Schools, includes "instructional" expenditure. (b) Excludes outlays for departmental overhead and for transport services, which are shown in columns 5 and 6; for secondary schools, includes school salaries; in general, includes part-time teachers not paid directly by the department. (c) The total is taken from vote 27; it is allocated between 1st and 2nd level according to use of transport services, as reported by Department of Education; and is allocated within 2nd level according to numbers of students. (d) From vote 27 (Office of Minister), but excluding items in note (j) Transport; also excluding National Library, National

| Type of school | Salaries ^a | Pensions an superann'n | d other current ^b | Sub total col. 1+2+3 | Allo overhead ^d | cated Transport ^c | Total current | Current per pupil | Capital | Grand total |
|--|-----------------------|---------------------------|--|-------------------------|-------------------------------|---------------------------------|------------------|---------------------------------------|--------------------|------------------|
| <i>1st Level:</i> National schools | 24,357 | 3,649 | 822 | 28,828 | 1,000 | 1,414 | 31,242 | 60 | 4,406 ^e | 35,648 |
| 2nd Level: Total Secondary schools ^f | 20,864 12,078 | 214 | 10,310 6,116 | 31,382 | 1,105 | 1,926] 1,424 | 34,419 21,238 | 139 | 5,492 1,392 | 39,914 24,782 |
| Comprehensive and Community Vocational ^g Reformatory and | 911 7,875 | 186 | 4,194 | 12,255 | 424 | 502 |] 13,181 | 190 | 3,152 947 | 14,128 |
| industrial schools Sub total: 1st and 2nd | (n.a.) | (n.a.) · | (n.a.) | 362 | 13 | 0 | 375 | | 170 | 545 |
| Levels | | - | - 1 | 60,572 | 2,118 | 3,340 | 66,036 | 0.07 | 10,068 | 76,107 |
| 3rd Level: Total Regional technical colleges ^h | • | | | 9,698 980 | 35 | U | 9,995 | 337 | 1,347 | 2,203 |
| Training colleges ¹ Universities | | • | an a | 1,096 6,664 | 35 226 | | 1,131 6,890 | · · · · · · · · · · · · · · · · · · · | 1,368 | $1,131 \\ 8,258$ |
| expenditures | | | | 959 |) . | | 959 | · · · · | | 959 |
| Total, all levels | | ана — 11 л. К | | 70,270 | 2,415 | 3,340 | 76,031 | | 12,784 | 88,815 |

Table 4.10: Public education expenditures, year ending March 31, 1974 (x £1,000), Constant (1970) prices^k

Museum, organisational Grants-in-Aid; allocated in proportion to expenditures given in column 4 above, except "Other higher education expenditures". (e) From vote 8. (f) Includes secondary tops. (g) Current data from Department of Education; capital data from central Statistics Office, National Income and Expenditure 1974; includes some 3rd level from Technological Colleges. (h) Includes some 2nd level students; data from vote 30. (i) From vote 28. (j) Higher education grants, scientific research grants to students; university scholarships and fellowships; and National Council for Education Awards; from vote 27. (k) Deflated with Gross Domestic Product implicit price deflator, 1974 = 152.0, throughout.

For Sources see Table 4.9; Notes are carried forward from Table 4.9.

£100 million in 1973/74, and capital expenditures added another £16 million. Third level expenditures from public sources were £15 million for current and £4 million for capital items. These amounts include administrative overhead (as indicated in the notes to Tables 4.9 and 4.10), allocated in proportion to expenditures on salaries, pensions and superannuation, and other current expenditure in each type of school or at each level. Total public current and capital expenditures at all levels were approximately £135 million. When one divides these amounts by the numbers of full-time students,¹⁰¹ the resulting per-pupil average expenditures for 1973/74 are: National Schools, £91; Secondary, Comprehensive, and Community, £181; Vocational, £289; second level as a whole, £211; and third level as a whole, £512.

These figures indicate a remarkable disparity among levels. Third-level institutions enrol 3.6 per cent of all full-time students at all levels in the State (see Table 2.1, in Chapter 2), but absorb fully 13 per cent of public current and 21 per cent of all capital expenditures. Second-level institutions enrol 29.6 per cent of all full-time students, but take 45 per cent of all current and 43 per cent of all capital expenditures. National Schools, which enrol 62.8 per cent of all full-time pupils, are provided with only 41 per cent of current and 34 per cent of capital funds. According to 1974 participation rates, approximately 96 per cent of the population attends National School in the appropriate years of age; about 85 per cent pursue a junior cycle course; only about 40 per cent attend senior cycle; and about 12¹/₂ per cent go on to third level. These figures are obtained by dividing relevant enrolments by the appropriate age groups in the population. These are not random percentages: the 40 per cent who go to senior-cycle courses, and the 15 per cent who go on to third-level education are drawn mainly from the upper strata of society. The Higher Education Grants Scheme, however, has something of an equalising influence. The implication is that public education expenditures as currently structured are regressively redistributive in the extreme. When one considers that life-long income is liable significantly to be influenced by access to schooling, i.e., that these inequalities in the distribution of public resources are a source of further social inequalities, it becomes clear that some criticism is justified.

All calculations were carried out in constant, 1970 prices. Table 4.10 repeats the information of Table 4.9, but in 1970 prices.

The following are the principal assumptions underlying the 1986 forecast. It is assumed that teacher salaries will rise at an annual rate of 2.75 per cent (a conservative estimate of the 'Baumol's Disease' effect described in Chapter 3), in addition to the costs of equal pay and maturation of the 'bulge' among second-level teachers (also as described in Chapter 3). It is assumed that budgeted amounts at all levels for public expenditures on pensions and super-

^{101.} This involves a bit of an error in that expenditures are made also on behalf of part-time students. Ideally we would divide public expenditures by full-time-equivalent enrolments, but no statistical series exists on full-time-equivalents.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

annuation will retain the same relationship to salaries as they had in 1973/74. It is also assumed that other current expenditures, departmental overhead, and transport costs will all rise at the identical rate of 2.75 per cent per year per child. These are items that are influenced strongly by labour costs and hence by 'Baumol's Disease'. Transport costs are also influenced by fuel costs, and may rise still more rapidly than 2.75 per cent annually. Similarly, 'other current' includes whatever costs are associated with an increasing technical content of the curriculum, as discussed in Chapter 3. It is clear that the forecast growth rate in these items is extremely conservative, and actual growth could be significantly higher. We have tried to make our predictions conservative at every turn, because even doing so yields alarming trends; and it was thought preferable to have the predictions conservative than to have them dismissed over sensible but arguable assumptions showing faster growth. Capital outlays are somewhat more complex. The number of new places required in new and expanded schools depends on the growth of enrolments, to which an amount must be added for population mobility and for obsolescence of old schools. It has been assumed that the cost per pupil place will rise at an annual rate of 3 per cent per year. In light of recent trends in construction costs, that is again a very conservative estimate. Finally, it is assumed that the present arrangement for sharing new school and expansion costs between the public and private sectors in National and Secondary Schools will continue into the future. Any further public absorption of costs, such as a public contribution to site acquisition, will increase future public expenditures still further.¹⁰²

The results are shown in Table 4.11. The principal findings are as follows. First and second-level expenditures on education, taken as a whole, are predicted to rise by 95 per cent. In light of the fact that our assumptions are conservative throughout, it seems justifiable to 'round upward' this projection and to state that these expenditures are likely to double. This forecast concerns real public expenditure, i.e., public expenditure adjusted for inflation. This represents an annual growth rate of 5.7 per cent. Since this is almost certain to be substantially greater than the annual growth in Gross National Product, the indicated increase in public education expenditures will have to be at the expense of some other types of expenditure in the economy. By this prediction, the growth in total expenditure in the various types of schools will be as follows: National Schools, 73 per cent for the twelve-year period, or 4.9 per cent per year; Secondary, Comprehensive, and Community, 101 per cent overall, and 6.0 per cent per year; Vocational, 234 per cent overall, and 6.9 per cent per year. While the assumptions were conservative, the results are not; this would be explosive growth in any country.

It has not been part of this study to analyse third-level expenditures. However, it is in third-level that the most rapid enrolment growth is expected. If per pupil current expenditures grow at the same rate as that forecast in 102. Detailed assumptions are spelt out in Appendix 4.D.

IRISH EDUCATIONAL EXPENDITURES -

| Type of school | Salaries ^a | Pensions and superann'n co | l other urrent ^b | Sub total col. 1+2+3 | Allo overhead | cated Transport | Total current | Current per pupil | Capital | Grand total |
|------------------------------|-----------------------|-------------------------------|--------------------------------|-------------------------|------------------|--------------------|------------------|----------------------|---------|----------------|
| 1st Level: | | | | | | | | | | C4 04 |
| National schools | 44,774 | 6,716 | 1,330 | 52,820 | 1,618 | 2,287 | 56,725 | 94 | 7,616 | 64,34 |
| 2nd Level: Total | 46,166 | 504 | 18,543 | 65,213 | 1,983 | 3,411 | 70,607 | 230 | 12,349 | 82,95 |
| Secondary schools | 23,190 | | | ٦ |) |) |) | ٦ |) | ٦ |
| Comprehensive and | 4 1 9 0 | 59 | • 10,531 | > 37,900 | > 1,173 | > 2,452 | > 41,525 | > 188 | > 8,388 | 49,91 |
| Community | 4,120 | 445 | 8.012 | 27.313 | 810 | 959 | 29.082 | 337 | 3,961 | 33,04 |
| Reformatory and | 10,000 | 110 | 0,014 | _ ., | | | | | , | |
| industrial schools | | | | 747 | 27 | 0 | 774 | | 343 | 1,11 |
| Total: 1st and 2nd Levels | | | | 118,780 | 3,628 | 5,698 | 128,106 | 5 | 20,308 | 148,41 |
| Percentage growth, 1974-86: | (per cent | per annum) | | | | | | | 1.07 | 4.0 |
| National schools | | | | | | | 5.0% | » 3. 7% | 4.6% | 4.9 |
| All second level | | | | | | | 6.1 | 4.3 | 6.8 | 6.2 |
| Secondary, Comprehensive a | nd Comm | unity | | | | | 5.7 | 3.8 | 5.2 | 6.0 |
| Vocational | | | | | | | 6.8 | 4.8 | 12.6 | 6.9 |
| Reformatory and industrial s | schools | | | | | | 6.2 | | 6.0 | 6.1 |
| Total, all levels | | | | | | | 5.6 | | 6.0 | 5.7 |

| Table 4.11: Predicted public education expendit | tures, calendar year 1986 (x £1,000), constant (1970) prices |
|---|--|
|---|--|

Notes: See Tables 4.9 and 4.10. Sources and methods: See text and footnote 99, Chapter 4.

. .

PAST, PRESENT AND FUTURE

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

second-level (i.e., 4.3 per cent per year), they would rise from £375 in 1974 to £562 in 1986 (both in 1970 prices). With our forecast enrolment of 65,900 in third level, this would imply public current expenditure of £37 million, an increase of 270 per cent over the 1974 figure. If such rapid enrolment growth does take place, however, it is likely to occur largely in the non-University institutions, such as the Regional Technical Colleges, where costs may be lower. Expenditure increases may be held to 200 per cent (i.e., a trebling of present levels). Any enrolment increases would also imply heavy capital outlays on new facilities.

What can be done?

It is a fair guess that the enrolment and expenditure pressures discussed in this chapter will precipitate efforts to economise wherever possible. For example, average class sizes in second-level schools, and especially Vocational Schools, are liable to grow substantially in County Dublin and other expanding areas. Other efficiencies and economies have been proposed by experts on Irish education.¹⁰³ But unless there ensues a wide-ranging review of the entire question of the public role in education, and unless needs are systematically weighed against costs, one can confidently predict that the following will be the principal *de facto* responses to the problem:

(1) Parents and children will learn of the crisis not from government statements and actions, but from overcrowded classrooms, widespread use of temporary classroom buildings, and lack of sufficient places for all those who want to continue in school, especially in growing areas. Sheehan shows that an estimated 27.6 per cent of new pupil places in National Schools 1965-66 through 1973-74 were provided in temporary buildings, and guesses that about 20 per cent of places provided in second-level institutions in more recent years have also been in temporary classrooms. He shows that while the acquisition cost of temporary classrooms is less than permanent structures, the long-run cost is considerably higher. Use of such structures is typically the result of, and can in fact be taken as a sign of, inadequate forward planning (NESC, 1976, op. cit., pp. 62-69).

(2) The system will not be permitted to grow as indicated. It is far less likely that existing programmes will be cut back in absolute physical terms than that limits will be placed on further access to the system. This would

103. A preliminary version of this chapter was presented, as indicated in note 79, above, in a seminar sponsored by the Association of Democracy in Education, in Dublin, June 27, 1976. Economies proposed at the same seminar in response to the problem by Ms Joy Rudd included rationalisation of intermediate and group certificate examinations; issuance of free bicycles in place of costly bus service in some areas; and rationalisation of small, rural schools. Senator (now TD) John Horgan suggested re-examination; consideration of charging 'the going rate' for university education, and using a means-tested grants' system; reconsidering free pre-compulsory education; considering more use of non-formal education; consideration of a voucher system by which the State finances individuals and not institutions; and using more senior citizens, who seek a role in society, in the educational process. This quick summary fails to do justice to the thoughful contributions by these two authorities.
mean that whatever inequalities and inefficiencies presently exist in the system will in effect be locked in. This is all too frequently the response to fiscal stringency: the 'last ones in' are the first to be cut out.

These two would almost certainly be the products of a 'crisis approach' to the problem. They need not occur, however, if there is a full public consideration of the problem, and a review of the social interest in education. Obviously, economies will be necessary, and they will be burdensome to some; but if adequate discussion of the issue takes place, the economies and burdens can be those chosen by society, rather than the chance results of a failure to decide.

In this paper, it is not our purpose to lay out specific policy proposals. What is advocated here is an approach, not a programme. The approach may be illustrated or exemplified by a particular set of proposals, but these should be interpreted as only one possible application of an idea. The theoretical underpinnings of the approach consist of three propositions. They provide the main basis for four principles of funding of education in Ireland. The four principles, in turn, are drawn on in a suggested scheme of finance.

The first and most important of the theoretical propositions concerns the fundamental distinction between public and private benefits from education. Education, like many other goods and services, is a 'mixed' or 'quasi-public' good, with both public (or social) and private (or individual) aspects. When and to the extent that the individual pupil gains – whether in the form of a higher subsequent income, or lifelong satisfaction and intellectual curiosity, or even current pleasure in learning - to that extent, education is a private good. To that extent, there is, in principle, no more reason for State finance than there is, in general, for State finance of entertainment, recreation, food, and/or other private goods. Where society in general benefits - as, for instance, through having a literate and competent electorate, or a population with an appreciation of national history and language, etc. - to that extent, education is a public good. Education does not become a public good simply because individuals are trained in skills needed by society. While it is true that society gains from the presence of qualified physicians, accountants, teachers and mechanics, it is also true that these people are *paid* for whatever benefit society reaps from their presence. This fact makes their education a private rather than a public good. The fundamental issue that distinguishes private from public goods is whether the consumer – the pupil in this case – can capture the benefits arising out of his or her education, or whether these spill over into society as a whole.

The second proposition can be stated more briefly. It distinguishes, again among the benefits from education, between those which occur within Ireland and those which occur elsewhere. This distinction must be made, because as has been noted, many Irish school leavers and graduates emigrate. The question may be asked as to why Irish tax-payers should pay for benefits, public or private, which occur in Britain, the United States, or elsewhere. Even where a school leaver emigrates and never returns, however, some of the benefits from his or her education may occur within Ireland. This is so, for example, to the extent that educational benefits are current or consumption benefits, which by definition take place at about the same time as the education takes place. Only the yield from the *investment* element in education is likely to be lost on account of emigration.

Looked at together, the first two propositions might serve to delimit the range of *public* concerns in Ireland with education, to, first, the public element, and, second, to the Irish element.

The third proposition, which may modify the effects of the first two, is that income and social class should have no effect on any person's access to schooling. It has long been recognised that public finance of education serves legitimate redistributive as well as allocative functions and it is appropriate to reflect that point in the design of schemes for the finance of education.

When these propositions are employed in developing approaches to the financing of schooling in Ireland, they yield the following five principles.

(1) Private resources should be used for the private-good component in education, and public resources should be reserved for the public-good component. It is difficult in practice to separate these with absolute precision. A rule of thumb might be that the years of compulsory attendance (ages 6 through 15) be considered essentially public or social in character. Social compulsion, through the State, implicitly invests these years of schooling with a social character. This rule seems intuitively to be consistent with observable public-private distinctions. Beyond the age of compulsory schooling, i.e., in senior cycle and third level, employment and earning opportunities appear to vary appreciably with amount of schooling. Society makes the investment, in most cases; but the individual appropriates the returns.

(2) Where virtually every member of society takes education at a given level, and where the benefits from that education are mainly current rather than in the form of an investment, the distinction between public and private is probably superfluous. Public support does not appear inappropriate, even where some of the benefits are private, because it does not result in redistribution from the many to the few, nor does it explain lifelong earnings differences. (The same cannot be said, by contrast, of education whose benefits are mainly private, and which goes to but a minor fraction of the relevant population.) This statement applies only to education whose benefits are largely current (as in the infants' divisions of Primary Schools) because of the possibility of emigration.

(3) To the extent that public resources are used in the financing of education, they should be used, in so far as is practical, only to finance aspects of education which yield benefits within Ireland. This principle has relevance only for the later flow of benefits from the investment component in education, as (by definition) current or consumption-type benefits from an education in Ireland occur in Ireland.

(4) A more moderate alternative to the principle contained in item number one, above, is to continue the present practice of public finance for aspects of education which yield private benefits, but to treat such expenditures as personal income of the family of the pupil, and to tax it accordingly.¹⁰⁴ By contrast, the present scheme provides tax-free in-kind income benefits, related to lifelong income and social class. If this approach is to assist in solving the coming fiscal problems in education, the proceeds of such a tax treatment must be applied to education.

(5) Loan or grant funds should be made available to students with more modest means, as is required, in order that lack of funds may not deter anyone from obtaining an education that he or she would otherwise take. While the first three approaches, above, are based on considerations of allocational efficiency, and perhaps on the principle of reciprocal equity, this fifth one is based on that of distributional equity and equality of opportunity.

In sum, then, scarce resources available for education should be reserved, in general, for those aspects of schooling which benefit society at large as opposed to the individual learner, and in particular Irish society; and for the less advantaged. A number of actual financing schemes could be designed which would reflect such a philosophy, though the present scheme is not one of them. The following scheme is offered as an illustration, not as a definitive policy proposal. In our discussion, we follow the child through the years of schooling, beginning with the infants' division of primary school, and concluding with third level.

First, then, while attendance in the infants' divisions is not compulsory (and hence according to the first principle would be viewed as a private benefit), participation is very high (above 90 per cent), and so continued State support appears appropriate, on the basis of the second principle. Second, no major increase in the use of private funds is indicated in the financing of primary education, as it is compulsory and hence regarded as a public benefit. Third, the same can be said of junior cycle, second level, as attendance is compulsory through age 15. Thus far, things remain much as they are.

Fourth, senior cycle is regarded as a private, not a public, good. Hence there is a case for a substantial private contribution. Two schemes might be considered. In the first, State support of senior cycle is withdrawn. This would mean termination of the 'free scheme'; abolition of capitation grants, building grants, and State payment of incremental salaries, for Secondary Schools; and a system of fees, not only in Secondary Schools, but in Vocational and Comphrehensive/Community Schools. Fees would be set at levels high enough to recover the full costs of senior cycle, second-level 104. We are indebted to our colleague, J. G. Hughes, for suggesting this approach.

education. The resulting system, however, might deter a number of lower income young people from going on past the intermediate certificate. Hence, in harmony with the fifth principle, a system of State scholarships, based on income, would be introduced, possibly carrying a small stipend for living expenses (i.e., to cover income losses) for the students from families with the very lowest of incomes. Were the State funds today spent on second-level education reorganised along the lines suggested, school incomes would be higher (more private funds would be used, together with an unchanged amount of State funds), and a larger number of pupils would receive an education, than under the present scheme.

A more modest version of the same approach would be to repeal the 'free scheme', but to continue the other forms of State aid to second-level institutions, thus requiring lower (but hardly nominal) fees, not only in Secondary but in Vocational and Comprehensive/Community Schools as well, in the senior cycle. This alternative would still require a scholarship programme, though a smaller one.

If it is felt that the 'free scheme' represents a firm commitment which cannot be withdrawn, it may be necessary to adopt the more moderate alternative, as is contained in the fourth of our five principles, that of levying income tax on the value of State supported senior cycle education. If it is true, as we have argued, that senior cycle schooling is almost wholly a private good, whose benefits are captured or appropriated by the pupils, then any form of State finance represents a type of State transfer payment to the families of pupils, and should be taxed as income. The amount added to reported family income would be the grant given in lieu of (or, in Protestant schools, in aid of) fees, under the 'free schemes'; the capitation grant; and a pro rata share of State payment of incremental salaries. Tax-free allowances in the personal income tax would relieve the lowest income families from any payment, and hence their children would attend senior cycle without charge; for the rest of the population, the progressive rate structure would make the financing of senior cycle slightly redistributive, downward rather than upward as at present.

Fifth, our illustrative scheme calls for substitution of loans to students for grants to institutions and students in third level. Fees would be raised a very great deal — high enough to cover the current costs of third level institutions. While this is a drastic proposal, it is hard to justify the very high current level of State support for third-level education. Only a small fraction of the population has the opportunity to attend third level institutions. Those who do so are evidently rewarded handsomely, in terms of employment opportunities and higher earnings, as well as non-pecuniary amenities. There is a very strong case that learners should pay the full cost of third-level education. Those whose lifelong education-associated income increments will not be sufficient to pay for third-level education should not go on to third level (unless they wish to buy it as a consumption good). Those whose subsequent incomes will increase sufficiently to justify the education should use some part of their income gains to pay for the education.

Strict application of such a rule, however, faces two kinds of snags. First, while for persons with certain aptitudes a third-level education may make economic sense, i.e., the lifetime flow of added income, properly discounted, may exceed the cost, not everyone is equally in a position to finance such an 'investment'. The costs of third-level education are bunched in a short period of time, when the learner is fairly young, whereas the income gains are spread over a lifetime. How can a learner pay for today's education out of subsequent lifelong earnings gains? Only the wealthiest, in fact, could pay the full costs of a third-level education today, in anticipation of higher incomes tomorrow. And second, it is impossible to know with any certainty whether a particular individual's lifetime earnings gain will be sufficient to justify such an investment.

Both of these points suggest that the real problem in third-level finance is a capital market problem — essentially that there exists no efficient market to provide funds where this is economical. This capital market problem surfaces in practice as a distributional equity problem, because it is those with the least means who also face the greatest difficulty in financing a third level education. What seems implied is a system where fees are set so as to cover full instructional costs, i.e., where grants to third-level institutions and students are abolished or at least scaled down very considerably, but where a loan fund is established, to which anyone accepted for matriculation in a third-level institution would be eligible. Repayment of the loan subsequent to finishing or terminating education would be linked to income, and would be spread over a great many years.

The fact that a number of third-level graduates emigrate strengthens the arguments in favour of a loan scheme as opposed to a grant scheme.¹⁰⁵ Students who will not only capture the economic benefits of their own educations but who will do so abroad would seem to have among the weakest of claims to scarce educational resources in a period of fiscal stringency. For the student who, for example, takes a course in dentistry and then emigrates to practice elsewhere, a loan scheme provides a way for the Irish exchequer to recover some of the costs of providing that education.

Taxing the imputed income value of public aid to third-level education seems less appropriate than did the equivalent approach in senior cycle, second level. The reason is that far more third than second level students are financially emancipated from their parents. When second level is considered, the relevant income figure is family, i.e., parents', income. If the value of State aid to second-level education is to be taxed, it is in the parents' income tax that this will occur. When third level is considered, it may often be the students themselves. This means that in many cases, if

105. For this reason, a similar scheme was presented some years ago in connucction with graduate emigration (Lynn, 1968).

the student is to be in a position to pay tax, his or her grant would have to be increased accordingly, obviously a self-defeating proposition.

In summary, then, we suggest a reconsideration of the whole issue of State finance of education, with attention focused primarily on the vital distinction between private and public gain from education. As a principle, we suggest either that where gain is private, so also be the bearing of costs; or, failing that, where gain is private but costs are borne socially, that the beneficiary at least pay tax on the benefits. As an illustration of an application of such an approach, we have offered a programme of taxing the value to the pupil of State aid to second level, senior cycle; and substitution of a generous programme of loans for the present system of aid to third level education.

These may seem like harsh measures. But the alternatives may be still more harsh. The most likely alternative to come to pass is closing off further access to the educational system to those — predominantly in the lower half of the income distribution — who do not now find it possible to attend. But there may be other alternatives. Those who judge the present proposals excessively harsh might well put forth their own.

Chapter 5

Private Current Expenditures on Education

THE purpose of this chapter is to present estimates of privately-financed current expenditures in first and second level education in the Republic of Ireland.¹⁰⁶

In spite of the fact that education absorbs a large and growing fraction of the annual output of the Irish economy, there are no estimates of the total national outlay on education. Because with few qualifications, the Irish system is a private, aided one, rather than a state or public system, estimates of education expenditures do not arise as a by-product either of the budget process or of administration of the system. And for the same reason, estimates of public expenditures on education are not good surrogates for estimates of total expenditures.

This chapter presents estimates of the following:

(1) Total expenditures on private, non-aided primary schools. As noted in Chapter 2, these schools enrol 2.9 per cent of all students, first through third levels, and 4.3 per cent of all first level pupils. As they receive no State aid whatever, they are under no obligation to furnish the State with any expenditure data of any kind. As Sheehan notes, "Financial data are almost impossible to obtain" (NESC, 1976). No aggregate data have been collected by the State or by the schools themselves.

(2) Total and private National School expenditures. Total National School expenditures consist of the expenditures made by the schools, and the payment of teacher salaries by the State.¹⁰⁷ In order to estimate private National School expenditures, it is necessary to estimate total expenditures, exclusive of teacher salaries, and to subtract estimated grants. Effective in 1975, an earlier grants scheme was replaced with one by which the State provides a grant of $\pounds 6.00$ per pupil, provided that there is also a contribution of *at least* $\pounds 1.50$ per pupil from local sources. These local sources include parish collections, religious orders, other contributions, and so-called "voluntary pupil contributions." These last, paid by parents, usually at the annual rate of $\pounds 1.50$, can be the equivalent of unofficial fees, in that parents can feel morally, though not legally, bound to pay them. (Official fees are evidently barred by the constitution. There are indications that "pupil contributions" are not

^{106.} In this context, 'publicly-financed' means financed by State revenue sources (taxes, borrowing), and voted by the Oireachtas, together with expenditures made by Vocational Education Committees, the only type of local education authority in the Republic of Ireland, financed by rates.

^{107.} Technically, or perhaps we should say legally, it is the school rather than the State which pays the teachers' salaries; the State provides a grant to the school for this purpose, and, acting as the school's 'agent', pays the salary directly to the teacher, i.e., on the school's behalf, not its own. Whatever legal purpose this convoluted exercise may serve, for our purposes it is appropriate to describe things as they are, i.e., that the State pays the teachers' salaries directly.

asked from poor families.) While the Catholic Primary School Managers' Association has conducted surveys, to which reference is made below, neither the State nor any voluntary body collects statistics on National School expenditures.

(3) Total and private Secondary School expenditures. Total Secondary School expenditures consist of expenditures made by the schools, plus incremental salaries as paid by the State,¹⁰⁸ less grants. The various State grants to Secondary Schools, unlike those to National Schools, are not conditioned on an amount financed from local sources. Indeed, there are indications that it is not an uncommon occurrence for a Secondary School to receive, 'in a given year, more money in grants than it spends. For this reason, we must estimate *net* private expenditure on Secondary Schools, i.e., the excess of expenditures (exclusive of incremental salaries) over grants in those schools in which such an excess occurred, less the excess of grants over expenditures in those schools in which such a surplus was reported. The Catholic Secretariat of Secondary Schools has recently installed a system of centralised, audited annual financial reports (which they have graciously made available to us). However, they do not endeavour to aggregate these to obtain a total, either for expenditures or privately financed expenditures (nor in fact could they, as response from schools, while very high, has not been complete). The Protestant Schools have no such reporting practice. In addition, there are the 'lay Catholic' Secondary Schools, which as proprietary schools are under no obligation to report to the Secretariat. The State collects no expenditure data from these schools.

Until very recently, the State could not have collected accurate expenditure data from schools, even had it chosen to attempt to do so, because the schools themselves frequently lacked such data. National Schools have, with very few exceptions, been under the Management of the parish priest or clergyman, who many times did not keep a separate set of accounts for his school. Most Secondary Schools have been operated by religious orders, who similarly did not keep separate accounts for 'school' and 'community' functions. With the new Management and grants scheme introduced for National Schools in 1975, separate National School accounts are required to be kept. It is now a short step to State collection, aggregation, and publication of expenditure data. Similarly, the Catholic Secondary Schools' Secretariat now requires separate accounts for its member schools; most other Secondary Schools already maintain separate accounts. Accordingly, there are now few barriers to the development, either in the Department of Education or in the Central Statistics Office, of needed data series on total National and Secondary School expenditures.

To obtain total social current expenditures on education in first and second levels, the totals for Private Primary, National, and Secondary Schools must be combined with figures for Vocational Schools, Compre-108. See Chapter 2. hensive and Community Schools, and second level Regional Technical College expenditures. Since these are public institutions, it is assumed that their publicly-financed expenditures are equivalent to their total expenditures.

OTHER ESTIMATES

Account must be taken of a number of estimates, either of total or private education expenditures, or of related data. The Investment in Education¹⁰⁹ team made estimates for 1961/62. As noted, the Catholic Primary School Managers' Association has conducted surveys of its members' total and nongrant-financed expenditures. Several components of education expenditures are estimated by the Central Statistics Office and published in National Income and Expenditure.¹¹⁰ And the Household Budget Inquiry, also compiled by the Central Statistics Office, reports household expenditures on 'education and training' (Household Budget Survey, 1973).

National Income Accounts

Many states publish estimates of total expenditures on education as part of their National Income and Expenditure accounts. The Republic of Ireland does publish some education accounts in their National Income and Expenditure series, but total education expenditures cannot be inferred from the data published.

Table 5.1 records education expenditure data as they appear in the various tables of National Income and Expenditure 1974. (CSO, 1976). The approach taken by the Central Statistics Office will be more clear if their underlying assumptions are explained:

(1) As regards education expenditures by public authorities (i.e., by the State or by Vocational Education Committees), some are shown as expenditures on goods and services, whilst others are treated as transfer payments.¹¹¹ Those government education expenditures treated as expenditures on goods and services include those on National Schools, Comprehensive and Community Schools, Vocational Schools, and Regional Technical Colleges. The

109. Investment in Education, Dublin, Stationery Office, 1965, Volume II, Appendix V, Sections B and C. The report is discussed in Chapter 2, above, under 'The Recent Period'. However, an intro-duction to Appendix V, Section B, in which National School estimates appear, states: 'This appendix is a condensation of a report' prepared by T. O'Brien, M.A., who was attached to the Survey Team as an OECD Fellowship holder during the year 1963; the Survey Team does not necessarily accept all the conclusions of his report.

110. Dublin, Stationery Office, annually.

111. A transfer payment is a payment which is not part of an exchange, i.e., for which there is no concurrent product or service traded. Transfer payments, hence, include such unilateral or unrequited payments as grants, subsidies, gifts, unemployment assistance, etc. Expenditures on goods and services do, by contrast, involve equivalent exchange. Expenditures on goods and services are some-times described as 'resource-using' or 'exhaustive' expenditures, because unlike transfer payments they involve using resources which, therefore, cannot be used in the production of some other good or newing the second expension of a services by public authorities, concequently, form part of Grass service. Expenditures on goods and services by public authorities, consequently, form part of Gross National Product, while transfer payments do not.

reason these are so treated is that they are considered to be State or local authority schools, or at least more nearly public than private, independent schools. It will be recalled from Chapter 2 that the status of National Schools, as regards the public-private distinction, is ambiguous at best, but that the Department of Education regards them to be independent, aided schools, and their teachers to be employees not of the State but of the school. Evidently, the CSO, by contrast, regards the National Schools as State schools, and the teachers as State employees. Those government education expenditures treated by the CSO as transfer payments include payments to Secondary Schools and the Universities.¹¹²

(2) The household sector, as defined in the National Income and Product Accounts, includes non-profit institutions as well as households.¹¹³ Since Secondary Schools, Universities, and certain other recipients of government transfer payments are non-profit institutions, these grants are formally treated as transfer payments to the household sector.

With this background, we may briefly look at Table 5.1. The last section of this table, corresponding to Table A.24 in the original source, shows total *public* education expenditures.¹¹⁴ Note these are reported to be very nearly £100,000,000 in 1971/72, and estimated as approximately £174,000,000 in 1974. For the same years, current transfer payments are reported as £34,261,000 and £59,800,000, respectively, and current expenditure on goods and services are reported as £48,542,000 and £85,200,000, respectively.¹¹⁵

These totals are disaggregated somewhat in the other tables in the original source, i.e., A.23, which shows expenditures of local authorities; A.22, which shows central government expenditures; A.21, which shows details of capital spending; and A.20, which shows details of transfer payments.

What is missing, in order to obtain a figure for total education spending, is an estimate of private educational expenditure. National Income and Expenditure does include a table (i.e., A.11) which reports 'details of expenditure of personal income', in which one might hope to find a line reporting estimated personal expenditure on fees, contributions, etc., i.e., private education expenditure. This table, as is indicated in Table 5.1, includes an entry named, 'professional services (including education)'.

112. The interpretations given in this paragraph and in the following paragraph are as provided by J. B. Broderick, Deputy Director, Central Statistics Office, in a communication to the author, dated October 20, 1975, in response to an inquiry directed to him, dated September 5, 1975.

113. It is customary in national income accounting to divide the economy into four large sectors: households; businesses; government; and (on a net basis), the rest of the world.

114. Note, however, that in this section, only the data reported for 1971/72 is as reported by the CSO. The remaining entries, as explained in notes to the Table, are either brought forward from other tables in the original source, or are the present author's estimates, based on other tables in the original source.

115. For 1973/74, Table 5.1 shows total public education expenditure to be estimated at $\pounds 148,500,000$, and current expenditure to be estimated at $\pounds 125,000,000$ (the sum of $\pounds 52,000,000$, current transfer payments, and $\pounds 73,000,000$, current expenditures on goods and services. Table 4.9, in Chapter 4, using different sources and perhaps different definitions, showed total public education expenditure as just under $\pounds 135,000,000$, and current expenditure as just under $\pounds 135,000,000$, and current expenditure as just under $\pounds 135,000,000$, and current expenditure as just under $\pounds 135,000,000$, and current expenditure as just under $\pounds 135,000,000$.

| 1 abic 5.1. Euclation expenditure as reported in rational meetine and expenditure 1571 (w 21,000 | Table 5.1: Education es | cpenditure as re- | ported in National | Income and Ex | penditure 1974 | $(x \ \pounds 1,000)$ |
|--|-------------------------|-------------------|--------------------|---------------|----------------|-----------------------|
|--|-------------------------|-------------------|--------------------|---------------|----------------|-----------------------|

| Table number | r Table Heading | Name of Entry | 1971 | 1971 <u>-</u> 72 | 1972 | 1972- 73 | 1973 | 1973- 74 | 1974 |
|-----------------|--|--|--------|---------------------|--------|---------------------|--------|---------------------|---------------------|
| A.11 | Expenditure of Personal Income at current market prices, 1969-74 | Professional services (including educa- | 46,700 | | 55,400 | | 62,900 | | 73,000 |
| A.20 | Details of transfer payments, national debt interest, and capital | Current transfer payments to higher education | | 8,562 | | 9,820 | | 13,380 | 16,600 |
| | grants to households and private non- profits institutions, 1969/70 to 1974 | Current transfer payments to secondary education | | 17,784 | | 22,100 | | 26,750 | 28,700 |
| | | education Current transfer payments to scholarships | | 2,164 | | 3,270 | | 3,600 | 4,500 |
| | | and prizes | | 1.624 | | 1.910 | | 2,390 | 2,500 |
| | | Current transfer payments: school meals Current transfer payments: transport | | 304 | | 350 | | 430 | 400 |
| | | services for school children | | 3,715 | • | 4,630 | | 5,080 | 6,600 |
| | | Capital grants to high education | | 3,597 | | 4,340 | | 4,440 | 4,900 |
| | | Capital grants to training colleges | Ŷ | 291 | | 90 | | 380 | 500 |
| | , | Capital grants to secondary schools | | 2,032 | | 1,550 | | 1,080 | 1,000 |
| A.21 | Public authorities–details of gross | Central government: comprehensive schools | | 1,034 | | 3,440 | | 5,480 | 4,400 |
| | physical capital formation, 1969/70 to 1974 | Central government: regional technical college: Local authorities: vocational education | s | 917 | | 1,170 | | 2,050 | 2,000 |
| | | committees | | 1,788 | | 1,370 | | 1,440 | 2,600 |
| A.22 | Expenditure of central government (including extra-budgetary funds) | Education (total) <i>Education Detail:</i> | | 94,365 | | 114,897 | | 142,850 | 168,000 |
| | classified by purpose of expenditure | Current transfer payments | | 32,240 | | 39,876 | | 49,050 | 56,800 |
| | and economic category, 1969/70 to | Current expenditure on goods and services | | 35,800 | | 42,174 | | 51,080 | 60,400 |
| | 1974 | Current grants to local authorities | | 12,349 | | 15,144 | | 18,280 | 21,300 |
| | | Capital transfer payments | | 6,202 | | 7,109 | | 7,270 | 9,200 |
| | | Loans | | 1,770 | | 1,781 | | 2,630 | 3,500 |
| | | Gross physical capital formation | | 5,759 | | 8,529 | | 14,240 | 16,300 |
| | | Capital grants to local authorities | | 245 | | 284 | | 300 | 500 |
| A.23 | Expenditure of local authorities classified by purpose of expenditure | Education (total) Education Detail: | | 16,551 | | (a) | | (a) | (a) |
| | and economic category, 1969/70 to | Current transfer payments | | 2,021 | | (a) | | (a) | (a) |
| | 1971/72 | Current expenditure on goods and services | | 12,742 | | (a) _ь | | (a) _b | (a) _b |
| | | Gross physical capital formation | | 1,788 | | 1,370 | | 1,440 | 2,600 |
| A.24 | Expenditure of public authorities classified by purpose of expenditure | Education (total) <i>Education Detail:</i> | | 96,996 | | 119,000 | | 148,500° | 174,000° |
| | and economic category, 1969/70 to | Current transfer payments | | 34,261 | | 42,000 | | 52,000 [°] | 59,800 C |
| | 1971/72 | Current expenditure on goods and services | | 48,542 | | 59,500 [°] | | 73,000 | 85,200 ₄ |
| | | Capital transfer payments | | 6,202 | | 7,109 [°] | | 7,270 [°] | 9,200 ^u |
| | | Loans | | 444 | | 492 | | 550 | 900 |
| | | Gross physical capital formation | | 7,547 | | 9,899° | | 15,680 | 18,900° |

.

Notes: (a) Local authority data were only available through 1971/72.
(b) Not reported in A.23; carried forward from A.21.
(c) Author's estimates; not reported by CSO. See Appendix 5-A.
(d) Not reported in A.24; carried forward from A.22.
(e) Not reported in A. 24; this is the sum of corresponding items in A.22 and A.23; see note (b).

However, this item is not of very much use in our task. It will be recalled that the household sector is defined so as to include non-profit institutions. 'Expenditure of Personal Income' hence includes expenditure by Secondary Schools, Universities, etc.: and 'professional services (including education)' includes these institutions' expenditures on teachers' salaries (as well as households' purchases of education from commercial, for-profit institutions). Other lines in A.11 shows other expenditure by Secondary Schools, Universities, etc., for fuel, materials, food, etc.

Non-profit institutions are so significant in Irish economic and social. life that the obfuscation resulting from their amalgamation in the economic accounts with households is regrettable indeed. One would hope that the Central Statistics Office would consider reporting a five-sector economy, i.e., one which includes separately non-profit institutions and households, as well as businesses, government, and the rest of the world.

| Table | 5.2: | Household | education | expenditure | as | reported | in | Household | Budget | Survey |
|-------|------|-----------|-----------|-------------|----|----------|----|-----------|--------|--------|
| | | | | 1973 | 3 | | | | | |

| Item | Expenditure in £ per week, avg. family estate | Expenditure in £ per week, avg. urban family | Expenditure in £ per week, avg. rural family | Total 1973 expenditure for State x £1,000,000 |
|---------------------------------------|--|---|---|--|
| · · · · · · · · · · · · · · · · · · · | | · · · | | |
| Primary and nursery schools | 0.038 | 0.064 | 0.003 | 1.4 |
| Secondary/vocation-day | 0.018 | 0.031 | 0.002 | 0.7 |
| Secondary-boarding | 0.069 | 0.062 | 0.078 | 2.6 |
| Third level- | | | | |
| Tuition fees | 0.060 | 0.076 | 0.039 | 2.3 |
| Expenses away from home | 0.048 | 0.040 | 0.058 | 1.8 |
| Other education and training | 0.123 | 0.172 | 0.059 | 4.6 |
| Total education and training | 0.378 | 0.445 | 0.239 | 13.4 |

Sources: Columns 1, 2, 3 from Central Statistics Office, Household Budget Survey 1973 Dublin, Stationery Office, 1976. Column 4, based on D. C. Murphy, "1973 Household Budget Survey, Special Features and Results," paper read to Statistical and Social Inquiry Society of Ireland, May 20, 1976. See text.

Household Budget Survey

The most recent household budget survey, that for 1973,¹¹⁶ differs from its predecessors in that it includes, for the first time, responses from rural as well as urban households. It consequently gives expenditure patterns for all households in the State. The Survey reports education expenditures in some detail. Column I in Table 5.2 shows average weekly education expenditure

116. The inquiry is conducted using sample survey research methods, primarily for purpose of providing updated weights for the Consumer Price Index. It obviously has other valid uses as well. (CSO, 1976).

by households to be 37.8p. The largest item is "other education and training", presumably reflecting payments to commercial education and training institutions which receive no State aid. The second and third columns, respectively, show average weekly expenditure by urban and rural families. The figures in the second column can be compared with the results of the 1965-66 survey, which covered only urban families (CSO, 1969). This survey (when adjusted for price level changes)¹¹⁷ showed households to be spending 56.7p per week (in 1973 prices) on education, composed as follows: primary day, 3.6p; secondary day, 17.8p; vocational, 1.8p; boarding school, 9.3p; university, 11.3p; and other educational and training, 12.9p. The most striking change is the fall in day secondary/vocational expenditures of 16.5p per week, which more than accounts for the fall in total educational spending of 12.2p per week. The explanation is obviously the introduction of the "free scheme", as discussed in Chapter 2.

The fourth column of Table 5.2 shows estimated total household expenditures on education for the State. These represent a grossing up of the figures in the first column. D. C. Murphy has grossed up the average weekly expenditures into aggregate expenditure categories, using the total number of private households in the 1971 census, and making some additional adjustments which do not affect our expenditure data (Murphy, 1976). For our purposes, the relevant aggregate entry is that for "Services and Other Expenses," estimated by Murphy at £227,000,000. Column 4 in Table 5.2 is estimated simply by taking these items' proportions of Services and Other Expenses and applying them to the Murphy estimate.

Thus for the first time there is a CSO-based estimate, equal to £13,400,000 in 1973, of household expenditure on education. This estimate is conceptually different from an estimate of "personal" expenditure on education, since the latter includes the expenditures of non-profit institutions. It is also not the same as private education expenditure, for similar reasons. It does not reflect, for example, financial contributions by parishioners to their churches, which survey households surely identified as "Church contributions" (on which they spent an average of 41.1p per week in 1973) rather than education expenditures; and, though the Churches in turn contributed to schools, the Household Budget Survey does not reflect this. There are other differences between the two concepts of household and private education expenditure.

The estimate of £13,400,000 in household education expenditure is approximately 9 per cent of the 1973/74 estimate, as shown in Table 5.1, of £148,500,000 of public education expenditure.

117. We have used the implicit price deflator for Public Administration and Defence, calculated from *National Income and Expenditure 1972* and 1973, which (using as a base the two years 1965-66) showed a value of 218.6 for 1973. The implicit deflators were calculated by Barry Murphy, whose assistance is gratefully acknowledged.

Investment in Education Estimates

Estimates made by the Investment in Education team for 1961/62 of "local contribution" may be summarised briefly. This source reports State expenditure on National Schools, exclusive of teacher salaries and superannuation provision, as £185,169, and the local contribution, also exclusive of teacher salaries,¹¹⁸ as £257,667, for a total of £442,836. The State share of this total is 42 per cent; the local or private share is 58 per cent. These figures change when "unrecognised running costs," i.e., running costs for which there is no grant provision, are added. The author "guesstimates" these costs at £55,467, which brings the total to £498,303, and changes the State/private proportions to 37 per cent and 63 per cent, respectively.

Investment in Education also reports on the income and expenditure of Secondary Schools, providing estimates based on a sizeable sample of schools of all sizes and types. For 1961/62, estimated current expenditure, inclusive of both basic and incremental teacher salaries as well as running costs, are £3,184,138. For the same year, estimated income from all sources is £4,301,584, which indicates apparent operating surplus on the part of Secondary Schools of £1,117,446-a massive amount, equal to more than 25 per cent of income. According to the report, "the major reason why revenue should exceed current expenditure is that capital would have to be met from this surplus". Another way of saying this is that depreciation is not included in current expenditures. The State contribution takes two forms: grants to schools, and incremental salaries paid to teachers. These two amount almost exactly to 50 per cent of all school *income*, but they amount to 68 per cent of reported school expenditures.¹¹⁹

CPSMA Survey

Fr. Leo Quinlan and his small staff at the Catholic Primary School Managers' Association, in order to buttress their case for improved grants (an endeavour in which they were successful), conducted a survey of their own members, to determine the annual current ("running") costs, as well as the excess of costs over governments grants, for the school year 1971-72. A summary of results was published in The Education Times.¹²⁰

The survey was not a random one. Schools were selected, Fr. Ouinlan has

118. T. O'Brien, who was responsible for the relevant section, estimates a value of supernumerary teachers (i.e., teachers in excess of the minimum numbers required) in capitation convent and monastery National Schools, which are, in effect, a locally contributed teacher salary item. In that year, supernumaries were a far larger fraction of the total teaching force than they now are.

119. The expenditure estimate appears in Table C.8, p. 313, Vol. 2 of *Investment in Education*. The total of $\pounds 3,184,138$ is divided into $\pounds 1,276,430$ in running costs, and $\pounds 1,907,708$ in teacher salaries. The latter figure is further divided into $\pounds 456,152$ in basic salaries, and $\pounds 1,451,556$ in incremental salaries. The income estimate appears in the same volume, in Table C.3, p. 309. The total of £4,301,584 is divided as follows: fees collected, £1,918,333; other school revenue, £229,399; State grants, £702,296; and incremental salaries, £1,451,556. It will be noted that the same amount for incremental salaries appears as both an expenditure item and an income item. A discussion of the apparent surplus is found in ibid., p. 317. 120. "The Cost of Running a Primary School," The Education Times, July 25, 1974, p. 14.

advised us, on the basis of the likelihood they would respond. Schools of a variety of types (lay, convent, monastery) and sizes were chosen. No information was collected on numbers of pupils (size is as indicated by number of rooms, and number of teachers). No information was collected on grants received; instead, the CPSMA estimated maximum grants possible under legislation existing at that time. The 65 schools responding reported expenditures (exclusive, of course, of teacher salaries) of £182,774, and maximum grants possible for one year of £35,160, or approximately 19 per cent of expenditures, leaving 81 per cent to be financed from local sources. Total expenditure less furniture repairs, sundries, and equipment came to £96,273, on which basis grants financed 37 per cent of expenditures, and local sources 63 per cent. The grants referred to are those for heating, cleaning, and painting. Furniture repairs, sundries, and equipment were subtracted by the CPSMA in one version of their results because some of these may have qualified for grants, and these grants may not be figured into the "grants" total. On the basis of an assumed 40 pupils per teacher, the CPSMA estimates total (non-salary) expenditures of £6.00 per pupil, and maintenance grants of £1.15 per pupil.

For the year ending March 31, 1971, the Department of Education reports total grants toward the cost of heating, cleaning, and painting of schools of £425,821. If it is assumed that these grants covered 37 per cent of such maintenance costs, then total maintenance costs can be estimated at £1,150,868, and the privately (or locally) financed portion can be estimated at £725,047. If it is assumed that grants covered only 19 per cent of expenditures, then total expenditures can be estimated at £2,241,163, and the privately financed portion at £1,815,342. These last figures yield a total maintenance expenditure of £4.42 per pupil, of which £3.58 is locally financed, and £0.84 is financed by grant. In that same year, 1970/71, total National School expenditure from public funds, inclusive of teacher salaries, is reported as £27,570,029.¹²¹ If an estimated £1,815,342 of privately financed expenditure is added to this, a total of £29,385,371 is obtained. This amounts to £57.91 total public and private expenditure per pupil, as compared with £54.33 per pupil of public expenditure.

Other CPSMA Data

The CPSMA routinely surveys schools on a variety of matters, including finance, on a diocesan basis. All dioceses are not surveyed in a given year. Fr. Quinlan was able to provide us with complete surveys of all schools in three dioceses, two for the school year 1970/71, and one for 1969/70. The results are given in Table 5.3. In the Ossory Diocese, grants covered 44 per cent of recorded school expenditures (exclusive of teacher salaries) in 1969/70. In

121. Total primary division expenditure is reported as £28,296,646, of which £726,617 is Training College expenditure, the latter being excluded from the figure in the text. An Roinn Oideachais, *Tuarascail, 1968/69-1971/72*, Dublin, Stationery Office, 1974.

the following year, grants covered 49 per cent of expenditures in both of two dioceses, those of Elphin and Killaloe.

Table 5.3: Summary of results of CPSMA surveys of National Schools, selected areas,
years

| Diocese | Year | No. of schools | No. of pupils | Expendi- ture per school | Expendi- ture per pupil | Grant per school | Grant per pupil | Ratio of grants to expendi- tures |
|----------|---------|-------------------|------------------|--------------------------------|-------------------------------|------------------------|-----------------------|--|
| Elphin | 1970/71 | 140 | 11,060 | £233 | 2.95 | £114 | 1.44 | .49 |
| Killaloe | 1970/71 | 162 | 10,843 | £203 | 2.75 | £ 99 | 1.34 | .49 |
| Ossory | 1969/70 | 95 | 11,068 | £274 | 2.36 | £119 | 1.05 | .44 |

Source: Data provided by Catholic Primary School Managers' Association, Fr. Leo Quinlan, Secretary.

Convent Primary Schools

The convent primary schools are organised into the Conference of Convent Primary Schools of Ireland. In 1974, Sr M. Columba, President of the CCPS, conducted a survey of member schools to determine, among other things, their maintenance expenditures and associated grants. In this survey, both categories were exhaustive. Maintenance expenditures were listed as heating and lighting; cleaning; cleaning materials; painting; and repairs. Grants were listed in some categories. Sr Columba provided us with the survey returns for use with the present study.

Sr Columba provided us with 246 responses, of which 189 were usable. The 189 returns, which amounted to approximately 40 per cent of the whole number of convent primary schools in the State, were comprised of 61 from Dublin Diocese and 128 from elsewhere. For the State as a whole, grants covered 43 per cent of expenditures, amounting to £2.46 per pupil of £7.65 in annual expenditures per pupil. Numbers of pupils are not reported in the survey returns. Instead, per pupil estimates are based on the average number of pupils per convent primary school for the State as a whole, 340. There was a striking geographical difference, however: in Dublin Diocese, grants covered only 26 per cent of expenditures (an average grant of £1,422 per school against an average £5,545 in maintenance expenditures), whilst grants covered 47 per cent (£555 against £1,194 average) outside of the Diocese of Dublin.

Convent Schools are not representative of National Schools, or even of Catholic National Schools. For one thing, their average size (roughly 340 pupils per school) is much larger than the average size of non-convent (including monastery) National Schools, 106. Second, Convent Schools tend to be located mainly in cities and towns, and their centralised locations may

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

affect their cost structures. Third, there are certain economies (discussed in Chapter 3 and estimated later in this chapter) associated with employment of religious in schools operated by religious orders. In sum, one would not expect Convent Schools' grants in relation to expenditures necessarily to follow patterns similar to those of other schools. (Note that we do not include the fact that Convent Schools are exclusively girls' schools among the reasons for expecting such differences. Our own survey revealed no discernible difference between expenditures in boys' and girls' schools.)

SURVEY RESULTS

Because of the inadequate and inconclusive character of existing data, it was necessary to develop estimates and analyses of our own. The Secretariat of Catholic Secondary Schools, under the leadership of Fr. John Hughes, had for the first time initiated centralised, audited accounts for member schools, effective with the school year 1974/75. These were made available to us, in full.¹²² The nature of these data will be discussed below. In order to be comprehensive with respect to Secondary Schools, these data were supplemented with our own surveys of Protestant Secondary Schools, and of "lay Catholic" Secondary Schools. In addition, small surveys were conducted of Private Primary Schools and of National Schools. In the case of National Schools, there was a separate oversample of Protestant National Schools, in order to compare these schools explicitly with Catholic National Schools.

Private Primary Schools

As is indicated in Table 2.3 of Chapter 2, there were in 1973/74 121 nonaided or Private Primary Schools (108 Catholic and 13 Protestant) enrolling 21,693 pupils. By 1974/75, the total had risen to 23,260. A survey questionnaire was sent to all of these schools, asking details of their income and expenditure for 1974/75.¹²³ Follow-up letters were sent to all those not responding to the initial mailing. Thirty schools returned completed questionnaires, of which 21 were usable. While this provides a small and probably biased sample, the response rate did exceed the predictions of a number of knowledgeable persons, and the results, such as they are, constitute the only such estimates known. The 21 schools comprising the sample consists of 16 with Catholic affiliation and 5 with Protestant affiliation; in the population,

122. In this instance, "in full" does not mean that data were available in usable form from every school. While most schools had reported, quite a number did not; and, since it was the first year in which such accounts were required, many which were filed were not deemed usable. Our deep debt to Fr. Hughes for these data and his co-operation must be acknowledged. In addition, we must acknowledge our debt to John Doorley, of Robert Kidney and Co., accountants to the Secretariat, and to Margaret Larminie and Mary Duggan, of the Secretariat's staff, who were extremely helpful in this and other aspects of the research.

123. The aid of Hilary Field, Secretary of the Association of Private Primary Schools, and of Antonia Healy, of Avoca School, Blackrock, County Dublin, in conducting the survey of Private Primary Schools, is gratefully acknowledged.

there are 121 schools in all, of which 13 have Protestant affiliation. In our sample, 15 are from County Dublin, and the remaining 6 from the rest of the country; in the population as a whole, 42 of 121 schools are outside of County Dublin.

In spite of the small sample, a number of ordinary least squares regressions were estimated, in order to find whether any determinants of total or perpupil expenditures could be identified. Independent variables employed included convent/non-convent school (dummy variable, or d); Protestant affiliation (d); joint primary-secondary school (d); outside County Dublin (d): alternatively, outside Counties Dublin and Louth (d); alternatively, outside Counties Dublin, Kildare, and Wicklow (d); all boys (d); all girls (d); number of pupils (not used for per-pupil expenditures); number of teachers (a measure of school size); and number of religious teachers. Obviously, the small number of observations meant that no more than four or five of these variables could be included in any single equation. Consequently, a great many regressions were run, in order to include a wide range of possible combinations, so that net effect of each variable could be observed, controlling for other variables. In general, the results were negative. Most of the variables consistently had no observable effect on total or per pupil expenditures. The number of pupils, needless to say, consistently influenced total expenditures, with extremely high t-statistics. Only two variables had significant effects on per pupil school expenditures; Protestant affiliation, and joint primarysecondary school. Some Private Primary-Schools share campuses with Secondary Schools, and while separate books are kept for the two divisions, there is some need to allocate certain fixed, overhead costs, as between the two. Our data reflects the reported expenditures on primary divisions of such schools. For example, one regression equation showed the following results:

| Per-pupil expenditures | £65.23 | + 73.80 (| Protestant affiliation) | +49.17 | (Joint primary- secondary) | $\overline{R^2} =$ | .574 |
|---------------------------|------------|-------------------|----------------------------|--------------|----------------------------------|--------------------|------|
| t-statistic | 6.19 | 3.61 (s 99 per | ignif. at cent) | 2.66 95 p | (signif. at per cent) | <i>F</i> = 1 | 4.50 |

Almost exactly the same coefficients were obtained when other variables were included with these: the other variables were never significant. (These other results are not shown here.) This form of the equation could not be grossed up in order to estimate total private primary school expenditures, as the incidence of "jointness" in the population is not known. Equations in which the jointness variable was not included showed systematically different values for the Protestant affiliation variable. In these equations, the value of this

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

| | Catholic | Protestant | All schools |
|---|----------|------------|-------------|
| Overall Per-Pupil Expenditure (a) | £77.51 | £168.50 | £95.33 |
| School average, Per-Pupil Expenditure (b) | £78.10 | £172.79 | £99.17 |
| Standard Deviation of Above | £35.08 | £63.13 | £58.11 |
| Sample size | 16 | 5 | 21 |

Table 5.4: Per-pupil expenditures, results of survey of non-aided primary schools, by Catholic and Protestant affiliation, 1974/75

(a) Total expenditures, all schools, divided by total pupils, all schools. Equivalent to weighted average by school.

(b) Unweighted average of school per pupil expenditures.

coefficient was never lower than $\pounds 93.03$ and never higher than $\pounds 93.17$. This should be interpreted as an estimate of the difference between per-pupil expenditures in Protestant and Catholic Private Primary Schools. As Table 5.4 indicates, this is very close to the difference between *average* values of perpupil expenditures in Catholic and Protestant schools. On an overall basis, per-pupil expenditures in Catholic Schools in our survey were $\pounds 77.51$, and in Protestant schools, $\pounds 168.50$, a difference of $\pounds 90.00$. The unweighted averages of school per pupil expenditures were $\pounds 78.10$ in Catholic schools, and $\pounds 172.79$ in Protestant schools, a difference of $\pounds 94.69$.

The overall per-pupil expenditures, rather than the regression results, were used in estimating the total, although the two are virtually equivalent. Overall per pupil expenditures have been carried forward to Table 5.5, where they are multiplied by the estimated total number of pupils in Catholic- and Protestant-affiliated schools, respectively. This yields estimated total expend-

| | Catholic | Protestant | Total |
|----------------------------------|----------|------------|------------|
| Pupils (a) | 21.825 | 1 435 | 23 260 |
| Per pupil expenditures (b) | £77.51 | £168.50 | £83.19 (c) |
| Total expenditures (x 1,000) (d) | £1,692 | £243 | £1,935 |
| | | | |

Table 5.5: Estimated per-pupil and total expenditures, non-aided primary schools byCatholic and Protestant affiliation, 1974/75

Sources:

(a) Total from An Roinn Oideachais, *Tuarascail 1972/73-1973/74*, Dublin, Stationery Office, Catholic/Protestant estimated from proportions as in Table 2.3, Chapter 2, above.

(b) From survey; see text.

(c) Estimated total (Line 3) by total students (Line 1).

(d) Columns 1 and 2, from Line 1 times Line 2; Column 3, sum of Columns 1 and 2.

itures of £1,692,000 in schools with Catholic affiliation, and £243,000 in schools with Protestant affiliation, for a total of £1,935,000 in estimated Private Primary School expenditures. This total, when divided by the reported number of Private Primary School pupils, yields £83.19 in average expenditures per pupil in 1974/75. As will be seen later in this chapter, this is *less* than the per pupil current expenditure in National Schools. Obviously, there is a good deal of variation among Private Primary School teachers are less well paid than National School teachers, thus accounting in part for lower per pupil expenditure in the latter. But it is also clear that, in general, pupils in Private Primary Schools do not receive either more lavish or more costly schooling than those in National Schools.

Unfortunately, our Private Primary School survey was conducted for the school year 1974/75, just as the Department of Education was changing its fiscal reporting to a calendar year basis. Figures on public expenditure, with which one would like to compare the Private Primary expenditure, are available for the year ending March 31, 1974, for the nine-month period ending December 31, 1974, and the year ending December 31, 1975. In later discussions, we have extrapolated from our £1,935,000 estimate of Private Primary expenditures in 1974/75 to obtain the following "guesstimates": 1973/74, £1,389,135; nine months of 1974, £1,597,300; and calendar 1975, £2,044,715 (see Table 5.7).¹²⁴

Protestant National Schools, 1974/75

Earlier in this chapter, we reported the results of surveys conducted by the Catholic Primary School Managers' Association and the Conference of Convent Primary Schools. The data obtained in these surveys were based on the old, pre-1975 grants scheme. In order to determine how Protestant National Schools data compared, we conducted a small survey in what proved to be the final year of the old scheme.¹²⁵

The sample was not random, but was drawn in order to assure representative distribution in terms of size and region, from among managers deemed likely

124. The extrapolation was carried out in the following manner. It was assumed that over the relevant period, Private Primary School expenditures probably grew at about the same rate as National School salaries. These were reported as follows: $1973/74, \pm 37, 022, 013$; nine months $1974, \pm 32, 381, 374$; and $1975, \pm 59, 015, 071$. It was further assumed that National School salaries for calendar 1974 would equal the nine months of 1974 plus one third of the 1973/74 figure, i.e., $\pm 44, 722, 045$; and that National School salaries for 1974/75 (i.e., the twelve months ending March 31, 1975) would equal the nine months of 1974 plus one-third of the 1975 figure, i.e., $\pm 52, 053, 064$. These assumptions yielded the following, expressed as index numbers: 12 months ending March 31, 1974, 71.1; nine months ending December 31, 1974, 62.2; 12 months ending December 31, 1974, 85.9; 12 months ending March 31, 1974, 85.9; 12 months ending March 31, 1974, 75, 100.0; 12 months ending December 31, 1975, 113.4. These index numbers were applied to our 1974/75 estimate of $\pm 1,935,000$ to provide the estimates for Private Primary School expenditure reported in Table 5.7.

125. The kind assistance of Kenneth Milne, Secretary to the General Synod Board of Education of the Church of Ireland, is acknowledged. He assisted in the drawing of the sample, in the construction of the questionnaire, and in other matters.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

to respond. A summary of the results is provided in Table 5.6. Sixty-six questionnaires were sent out by post; a follow-up letter was sent to those not replying in the first round. In all, response was exactly 50 per cent, and of the 33 replies, 28 were usable. The average replying school received grants equal to 45 per cent of its non-teacher salary expenditure. In the total sample, the proportion was 43 per cent. These figures are quite comparable with those cited earlier, for Catholic National Schools in previous years.

Table 5.6: Summary of results of suvey of Protestant national school income and
expenditure, 1974/75

| Number of questionnaires sent out | 66 |
|--|---|
| Number of completed questionnaires returned | 33 |
| Number usable (i.e., sample size) | 28 |
| Total number of pupils, schools in sample | 1,830 |
| Average number of pupils per school in sample | 65 |
| Total expenditures (exclusive of teacher salaries), all schools in sample | £16,864.00 |
| Average expenditure per school | £602.29 |
| Standard deviation | £95.11 |
| Overall expenditures per pupil | £9.21 |
| School average, expenditures per pupil | £11.05 |
| Standard deviation | £7.24 |
| Total grants received, all schools in sample | £7,259.00 |
| Average grant per school | £259.20 |
| Standard deviation | £236.20 |
| Total grant per pupil | £3.97 |
| School average, grant per pupil | £4.98 |
| Standard deviation | £2.33 |
| Total expenditures less total grants | £9,605.00 |
| Expenditures less grants, average per school | £343.04 |
| Standard deviation | £581.87 |
| Total expenditures less grants, per pupil | £5.25 |
| Ratio of total grants to total expenditures Ratio of grants to expenditures, school average | $\begin{array}{c} 0.43\\ 0.45\end{array}$ |

Private Expenditures, National Schools

The estimates discussed thus far of grants as a proportion of total running costs of National Schools, exclusive of teacher salaries, converge in the range 37 to 49 per cent. In the order in which they appear in this chapter, they are 37 per cent (CPSMA, 1971/72, excluding furniture and repairs from both expenditures and grants); 44 per cent (Ossory Diocese, CPSMA, 1969/70); 49 per cent (Elphin and Killaloe Dioceses, 1970/71, CPSMA); 43 per cent (Convent Primary Schools, 1973/74); and 43 to 45 per cent (Protestant National Schools, 1974/75, old grants scheme). The unweighted mean of these estimates is 43.4 per cent; and the only estimates as recent as 1973/74

(Convent Schools and Protestant Schools) provide an unweighted mean of 43.5 per cent (taking 44 per cent as the Protestant National Schools estimate). Confidence in this figure is enhanced by the fact that Convent Schools (43 per cent) and Protestant National Schools (44 per cent) differ in a number of features, especially average size, about as widely as National Schools can differ. For purposes of further estimation, we take it as a working figure that running costs, exclusive of teacher salaries, were 230 per cent of grants for the State as a whole (which is the equivalent of saying that grants were 43.5 per cent of running costs). We assume that this ratio applied in 1973/74 and in the nine months ending December 31, 1974. This means that private or local funds contributed an amount equal to 130 per cent of grants.

Estimates based on this conclusion of private expenditure in National Schools for 1973/74 and the nine months ending December 31, 1974 are found in the first two columns of Table 5.7. (The third column will be discussed presently.) In 1973/74, private expenditure in National Schools is put at £1,389,135, or 3.0 per cent of total National School expenditure. In the

Table 5.7: Public, private, and total primary school current expenditure, nine months ending December 31, 1974, and twelve months ending December 31, 1975, with public and private National School expenditures, twelve months ending March 31, 1974

| | 1973/74 | 1974 | 1975 |
|--|-------------|-------------|-------------|
| National Schools | £45,069,650 | £41,040,729 | £71,941,263 |
| Public expenditure ^a | | | |
| Salaries of teachers ^b | 37,022,013 | 32,381,374 | 59,015,071 |
| Grants toward operating costs ^C | 1,068,565 | 1,228,692 | 3,286,581 |
| Superannuation of teachers | 5,910,979 | 5,535,771 | 8,851,719 |
| Other ^d | 178,958 | 297,592 | 393,502 |
| Total public expenditure | £44,180,515 | £39,443,429 | £71,546,873 |
| Private expenditure, National Schools | | | |
| Total | 1,389,135 | 1,597,300 | 394,390 |
| Private Primary Schools (Total) ^e | £1,282,004 | £1,121,528 | £2,044,715 |
| Total private expenditure, National and | | | , . |
| Private Primary Schools | £2,671,139 | £2,718,828 | £2,439,105 |
| Total Primary School expenditure | £46,851,654 | £42,162,257 | £73,985,978 |
| | | | |

(a) Excludes teacher training, fees for pupils in Secondary Tops, transport, and cost of administration, inspection, etc.

(b) Includes grants to capitation schools.

(c) Old basis, 1973/74 and 1974; new basis, 1975.

(d) Includes Model Schools (miscellaneous expenses); teachers' centres; aid toward the cost of school books; special educational project; and incidental expenses.

(e) Based on survey for 1974/75, indicating total expenditures in that year of £1,935,000 (see Table 5.5), as explained in text above.

nine months of 1974, private expenditure in National Schools is put at $\pounds 1,597,300$, or 3.9 per cent of the total. Total expenditure in National Schools is put at $\pounds 44,180,515$, or $\pounds 84.78$ per pupil, exclusive of administrative overhead and transport costs, for 1973/74; for the nine months of 1974, the respective figures are $\pounds 41,040,729$, and $\pounds 78.45$.

Table 5.7 shows estimated total private expenditures on primary education, consisting of local sources of finance for National Schools plus Private Primary School expenditures, to be $\pounds 2,671,139$. Assuming this figure to be reasonably accurate, it is interesting to note that it is considerably higher than the $\pounds 1,400,000$ reported in Table 5.3, above, as the Household Budget Survey-based estimate of household expenditure on Education and Training: Primary and Nursery Schools. The main reason (apart from the difference between 1973/74 and 1973 in inflationary economy with a growing primary school sector) is that there is an important difference between "private" and "household" education expenditure, as explained earlier. More light on this difference will be shed in the next section.

Table 5.7 shows total private expenditure to be 5.7 per cent of total Primary School expenditure in 1973/74, and 6.4 per cent in the nine months of 1974.

National Schools, New Grants Scheme, 1975

As has been noted previously, a new National Schools grants scheme was initiated, effective with calendar year 1975, whereby the State would pay up to $\pounds 6.00$, if the Committee of Management would raise at least $\pounds 1.50$ from local (i.e., private) sources and subject to other, non-fiscal conditions. This scheme suggests, on its face, a State contribution of 80 per cent of running costs, in addition to payment of teacher salaries. These terms represent a considerable improvement, from the standpoint of National Schools, both in relative and in absolute State contribution.

A survey was conducted of expenditures, grants, and other income sources of National Schools operating under the new scheme. A sample of 203 schools was chosen, using random sampling techniques. Schools were chosen according to the last two digits of their roll numbers in the Department of Education. These two digits, we were assured by the Department, are quite random. Four two-digit numbers were chosen from a table of random numbers. The questionnaire was drawn up with the advice and assistance of Fr. Leo Quinlan of the Catholic Primary School Managers' Association, whose contribution is gratefully acknowledged. Of these, 85 replied, and 79 of these questionnaires were deemed usable. A summary of the results is found in Table 5.8.

The results proved slightly surprising, in one respect. The vast majority of schools in our sample operated with fairly large budget surpluses. That is, the sum of State grants and funds raised locally tended to exceed expenditures, in most schools, by a fairly large margin (\pounds 329.54 in the average school in the sample). In the average school, total income per pupil was \pounds 7.98, com-

IRISH EDUCATIONAL EXPENDITURES – PAST, PRESENT AND FUTURE

| | <u> </u> |
|---|-----------------|
| Sample | |
| Number of questionnaires sent out | 203 |
| Number of completed questionnaires returned | 85 |
| Number usable (i.e., sample size-number of schools) | 79 |
| Protestant management | 5 |
| Catholic management | 74 |
| Parochial | 66 |
| Religious (convent, monastery) | 8 |
| Pupils | |
| Total number of pupils, all schools in sample | 15 514 |
| Average number of pupils per school | 196 |
| Average number of pupils per teacher, schools in sample | 33.1 |
| | 55.1 |
| Expenditures | |
| Total expenditures (exclusive of teacher salaries), all schools in sample | £89,175 |
| Expenditure, average per school in sample | £1,128.80 |
| Standard deviation | £1,960.68 |
| Total expenditure, all schools, divided by number of pupils | · £5.75 |
| Expenditure per pupil, average of schools in sample | $\pounds 5.96$ |
| Standard deviation | £4.09 |
| Grants | |
| Total government grants received, all schools in sample | £79 239 |
| Grants received, average of schools in sample | £1.003.03 |
| Standard deviation | £1.614.79 |
| Total grants, all schools, divided by number of pupils | £5.11 |
| Grants per pupil, average of schools in sample | £5.23 |
| Standard deviation | £1.99 |
| | |
| Relation of expenditure loss total grants | 80.08.0 |
| Total expenditure less total grants divided by number of numil- | £9,936 |
| Expenditure less grants, per pupil, average of schools in sample | £0.54 |
| Ratio of total grants to total expenditures all schools in sample | ±0.73 |
| Ratio of grants to expenditures, average of schools in sample | 0.89 |
| Ratio of grants to experienteres, average of schools in sample | 0.89 |
| Local contribution | |
| Total local contribution, all schools in sample | £35,970 |
| Local contribution, average of schools in sample | · £455.32 |
| Standard deviation | $\pounds715.45$ |
| Total local contribution, divided by number of pupils | £2.32 |
| Local contribution per pupil, average of schools in sample | £2.75 |
| Income | |
| Total income (grants plus local contribution) all schools in sample | £115 209 |
| Total income, average of schools in sample | £1 458 84 |
| Standard deviation | £9 914 85 |
| Total income divided by number of pupils | £7 48 |
| Total income per pupil, average of schools in sample | £7.98 |

 Table 5.8: Summary of results of survey of National School income and expenditure,

 1975

| Relation of income to expenditure | |
|---|-----------------|
| Ratio of total grants to total income, all schools in sample | 0.69 |
| Ratio of grants to income, average of schools in sample | 0.67 |
| Excess of total income over total expenditure, all schools in sample | £26,034 |
| Excess of income over expenditure, average of schools in sample | £329.54 |
| Standard deviation | $\pounds917.20$ |
| Excess of total income over total expenditure, divided by number of pupils, a | all |
| schools in sample | £1.68 |
| Excess of total income over total expenditure, per pupil, average of schools | |
| in sample | £2.02 |
| | |

prised of £5.23 in State grants, and £2.75 in local contribution; while total expenditure per pupil was only £5.96. A possible explanation is that in this, the first year of the new scheme, schools had not fully adjusted to their enlarged incomes, and had not increased expenditures accordingly, or were even covering past deficits.

Whatever be the explanation, for National Schools in our survey, grants were equal to 66 per cent of income, and 89 per cent of expenditures. It is the latter figure that must be used in estimating total expenditures. If it is assumed that this figure (which was the overall ratio as well as the unweighted school average) applies to National Schools as a whole, then total expenditures are estimated as 112 per cent of grants, and private expenditures as 12 per cent of grants. As grants in 1975 are reported as $£3,286,581,^{126}$ then total operating expenditures are estimated at £3,680,971, and private expenditures only £394,390 (see Table 5.7).

Alternative estimates of private expenditures in National Schools, and related data, can be obtained by multiplying the per-pupil averages from Table 5.8 by the number of National School children enrolled in 1975, 568,000. This method provides an estimated total for grants of £2,902,480, as compared with the actual (as found in Table 5.7, taken from Department of Education sources) of £3,286,581, implying an error of 13 per cent, and suggesting either that our sample is unrepresentatively low in grants (and presumably expenditures), or that not all grants reported by the Department were actually received in the survey period. The same method of using averages from Table 5.8 yields an estimate of privately-financed expenditures of £363,520, as compared with £394,390 reported in Table 5.7, a difference of 8.5 per cent. Total operating expenditures estimated in the same way are £3,266,000, 12.7 per cent less than the estimate in Table 5.7 of £3,680,971 (grants plus privately-financed expenditures).

The same method also yields an estimated total of locally contributed funds of $\pounds 1,317,760$; of total income of $\pounds 4,220,240$ (grants plus locally contributed funds); and the excess of income over expenditures of $\pounds 954,240$.

As 1975 was the first year of the new grants scheme, some interest attaches 126. An Roinn Oideachais. See Table 5.7.

to the source of these local or private funds amounting to just under $\pounds 1,000,000$. Figure V.1 indicates that Parish funds are the source of the major fraction of local contributions. Parents' contributions account for a sixth of locally raised funds; and, if pupils' voluntary contributions are added, the figure rises to just under 20 per cent. It is the sum of these two items that, one assumes, would be reflected in the household budget survey as Education and Training expenditures. (Most if not all of the funds represented in Figure V.1 originate in the household sector, but most of them are unlikely to be reported as "Education and Training" expenditures.).

Regression Analysis

Still more understanding of 1975 National School expenditures is provided by the results of a multiple regression analysis of the data derived from the sample. A number of regressions were run, using a variety of measures of income and expenditure as dependent variables, and available information concerning characteristics of the schools as independent variables. The purpose was not so much the testing of hypotheses (though hypotheses are certainly implied by the selection of variables) as it was the estimation of an equation or set of equations which could be used to interpret and understand 1975 expenditures.

A summary of the results is found in Table 5.9. In Lines 1 and 2, total school expenditures (as always, exclusive of teacher salaries) is the dependent variable. The \overline{R}^2 (adjusted throughout for degrees of freedom) is fairly high, largely reflecting the influence of number of pupils on total expenditures. Number of pupils is positive and significant throughout. In addition, in Line 2, the number of teachers (an index of school size) is significant and negative, when pupils per teacher is included. Pupils per teacher is also negative in its influence and is significant, which is hardly surprising.

In Lines 3 and 4, expenditure per pupil is the dependent variable. Moving number of pupils from the right to the left side of the equation drastically reduces the \mathbb{R}^2 . And only one independent variable shows a statistically significant influence on the dependent variable, namely, Protestant management, which accounts for £6.19 or £6.25 in added expenditure per pupil. A similar result is found in Line 5, where the dependent variable is income per pupil, rather than expenditure per pupil. Protestant management accounts for an added £8.02 per pupil in income. No other variable seems to have been significantly related to income per pupil.

In Line 6, the dependent variable is income less expenditure, or operating budget surplus, where income is defined as the sum of State grants and locally raised funds. This variable is expressed in a total, rather than per-pupil form. Though the \overline{R}^2 is fairly low (.213), there are three independent variables whose estimated coefficients are significant at the 99 per cent level—pupils (negative relationship), teachers (positive relationship), and pupils per teacher (positive relationship).



| \overline{R}^2 | Dependent variable F | Const't term | Protestant Mgt. | Religious Mgt. | Pub. works or other | All boys | All girls | No. of pupils | No. of teachers | Pupils per teacher | Per cent lay teachers |
|------------------|-----------------------------|-----------------------|--------------------|-------------------|---------------------------|-------------------|-------------------|---------------------|-----------------------|-----------------------|--------------------------|
| | Total expenditures | -523.99 | 330.26 | | 11.52 | -333.53 | -254.34 | 9.89*** | -112.742 | | 454.25 |
| .837 | 58.27 Total expenditures | (0.939) 1.046.98* | (0.890) 208.32 | | (0.433) | (0.953) 266.30 | (0.667) 288.94 | (3.524) 14.91*** | (1.074) -280.51*** | -47.92*** | (0.802) 289.82 |
| .853 | 65.82 | (1.368) | (0.588) | | | (0.800) | (0.800) | (4.657) | (2.418) | (2.835) | (0.536) |
| E | xpenditure per pupil | `5.76* [*] * | 6.25*** | | 0.139 | -1.269^{\prime} | -1.34 | , , | 0.055 | χ γ | -0.339 |
| 099 | 2.43 | (2.103) | (3.430) | | (1.063) | (0.737) | (0.718) | | (0.807) | | (0.122) |
| E | xpenditure per pupil | 5.80** | 6.19*** | | • • | -1.33 | -1.38 | | 0.053 | | -0.283 |
| 098 | 2.69 | (2.116) | (3.395) | | | (0.777) | (0.736) | | (0.769) | | (0.102) |
| | Income per pupil | 8.41*** | 8.02*** | | -0.015 | 0.578 | -0.528 | | -0.010 | | -0.918 |
| 251 | 5.35 | (3.811) | (5.553) | | (0.145) | (0.427) | (0.356) | | (0.193) | | (0.418) |
| I | ncome—expenditure | -861.63 . | 206.50 | | | 154.26 | -223.43 | -13.46*** | 522.05*** | 45.54*** | -643.85 |
| 213 | 4.02 | (1.039) | (0.538) | | | (0.428) | (0.570) | (3.880) | (4.154) | (2.487) | (1.100) |
|] | Expenditure—grants | -496.83 | 306.54 | 1.97 | | -555.69* | 30.89 | 4.46* | -115.37 | | 492.69 |
| .107 | 2.33 | (0.614) | (0.749) | (0.003) | | (1.437) | (0.061) | (1.404) | (0.992) | | (0.589) |
| | Expenditure—grants | -50.59 | 370.06 | | | | | 4.60* | -126.37 | | |
| .120 | 4.54 | (0.334) | (0.916) | | | | | (1.53) | (1.12) | | |
| j | Expenditure—grants | -15.46 | | | | | | 4.69* | -131.40 | | |
| 122 | 6.41 | (0.106) | | | | | | (1.567) | (1.170) | | |
| Expe | enditure-grants per pupil | 1.202 | 5.71*** | -1.071 | | -2.37* | -1.69 | | 0.095 | | -0.957 |
| 066 | 1.92 | (0.318) | (2.945) | (0.400) | | (1.298) | (0.713) | | (0.897) | | (0.242) |
| Expe | enditure-grants per pupil | 0.185 | 5.93*** | | | | | | 0.029 | | |
| .090 | 4.85 | (0.300) | (3.115) | | | | | | (0.450) | | |

Table 5.9: Regressions of 1975 National School income and expenditure, selected concepts, on selected independent variables

Significance: 90% *

**

95% 99% ***

137

- PAST, PRESENT AND FUTURE

In Lines 7, 8, 9, the dependent variable is expenditures less grants, or privately- (locally-) funded expenditures. It will be observed that the \overline{R}^2 s are quite low, presumably at least in part because the dependent variable is in fact the difference between two variables, which may be independently determined. No coefficients meet standard significance tests.

Finally, in Lines 10 and 11, the dependent variable is expenditures less grants, per pupil, i.e., private expenditures per pupil. Once again, presumably for the same reasons, \overline{R}^2 is extremely low. But on this occasion, as with other per-pupil variables, the Protestant-management dummy variable shows a very significant relationship. Evidently, National Schools under Protestant management spend from £6.19 to £6.25 more (Lines 3, 4), receive an estimated £8.41 per pupil more in income (Line 5), and spend £5.71 to £5.93 more in privately raised funds (Lines 10 and 11) than National Schools under Catholic Management. Throughout, we have found that Protestant Schools, whether they be National Schools, Private Primary Schools, or Secondary Schools, spend more per pupil than Catholic schools. Whether this consistent relationship is due to greater ability to spend, greater willingness to spend, or a greater necessity (e.g., arising out of the smaller scale of Protestant National Schools), is not obvious from the data. But the fact of the difference seems indisputable.

Secondary Schools, 1974-75

In 1974/75, the Catholic Secretariat of Secondary Schools, which has responsibility for the large majority of Secondary Schools in the State, instituted a system of centralised, audited accounts for its schools. These accounts were made available to us, on the condition that confidentiality be preserved. An analysis of these accounts, as supplemented with additional survey data, permits us to estimate total as well as public expenditure on Secondary Schools for 1974/75.

In 1974/75, there were 555 recognised Secondary Schools. Of these, 33 were under lay-Catholic administration, 26 were under Protestant and one under Jewish management, and the remaining estimated 494 came under the aegis of the Secretariat. When the accounts were made available to us, 302 schools had reported. Of the 302 accounts, 130 were considered usable for our purposes. Those excluded reported with insufficient detail to permit us to determine precisely what accounting concepts were used, especially as regards depreciation and the reporting of the value of services of religious: basic school salaries, salaries of non-recognised teachers, and value of non-teaching services, all of clerics, nuns, and brothers, as members of orders. The excluded 172 included virtually all of the Christian Brothers Schools which had responded, as most of them used a summary report form, which provided much less detail on income and expenditures than provided by other schools. These summary data suggest that Christian Brothers Schools were not dissimilar to the schools included here. Before the 172 schools were ex-

cluded, however, all schools were grouped geographically, in counties or groups of counties. The groups of counties do not correspond to provinces or regions established for administrative or other purposes. County groups were organised so that the numbers of schools in each group were roughly equal, irrespective of size. Only adjoining counties which were seen in some respects as similar to each other, were grouped together. Hence some counties, such as Kerry and Westmeath, had large enough representation in the sample for them to stand alone; while in one case, it was necessary to group as many as five counties, (viz., Donegal, Leitrim, Sligo, Mayo, and Roscommon) in a single group. As noted, the grouping of counties was accomplished before the data were pared down. This was unfortunate, as the pared down data turned out not to be grouped in counties and county groups of uniform size. It was felt, however, that it was inappropriate to alter the area boundaries once analysis had begun, and hypotheses, in effect, had begun to be tested.

The main purpose in grouping the schools geographically was to determine whether there were area differences in school incomes or expenditures. A major influence was expected to be income and wealth differences among counties. For similar reasons, County Dublin was divided into three areas. Dublin City (including contiguous suburbs) and the Borough of Dun Laoghaire were separated into "high income", "low income", and "other", with onefourth of the area's electoral wards or subdivisions into each of the first two categories. The other half of Dublin-Dun Laoghaire, together with the rest of County Dublin, were grouped together as "other".¹²⁷

When it imposed centralised, audited accounts, the Secretariat introduced one additional departure which is of import to us. Theretofore, separate records had often not been kept of "school" and "religious community" accounts, and in fact, for legitimate reasons, no difference might be seen between school and community. A consequence was that no record had been kept of the value of unpaid services contributed by members of religious orders. These consisted of three categories. First, religious teachers' school salaries were not actually paid, but were only "deemed to be paid".¹²⁸ Second, some "unrecognised" religious teachers, i.e., religious teachers in respect of whom no incremental salaries were paid, taught in these schools. And third, religious performed other, non-teaching services, such as landscape and garden work, or duties in connection with boarding pupils. The Secretariat's accountants asked with respect to the 1974/75 accounts, that religious be paid for work under these three headings, at amounts equivalent to local rates. If, and to the extent that, this money is returned to the schools by the

128. See Chapter 2, above, on this point. See also Investment in Education, Annexes and Appendices, p. 319, in which the value of basic salaries forgone in 1961/62 was estimated at £384,000.

^{127.} Strictly speaking, the "high income" area was actually a "low poverty" area. The "lowincome" areas were those scoring the highest, and the "high-income areas" those scoring the lowest, on "socio-economic deprivation" in Agnes Breathnach's study of deprived areas in Dublin (Breathnach, 1976).

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

religious communities, this transaction, too, is recorded, as a source of income. Thus, in effect, the religious, instead of contributing services, were to sell the services to the school and contribute the money. In what follows, we refer to the money values listed in the accounts under these three headings as "religious imputations", although they are not, strictly speaking, imputations, but represent actual (if nominal) transactions.

Data on expenditures (various concepts) by schools in the survey are summarised in Tables 5.10 and 5.11. In Table 5.10, total and per-pupil expenditures are estimated by type of school, e.g., day schools, schools accepting boarders (irrespective of numbers of boarders), fee-charging v. free scheme, and Catholic as opposed to Protestant management. Little significance should be attached to the first line, pertaining to all schools in the sample, as the weighting is fortuitious, and does not correspond to the distribution of schools, by type, in the State. Incremental salaries are not reflected in the data in either table.

It will be useful to explain the data in Table 5.10 in more detail. There were 84 day schools in the sample participating in the Department of Education's "free scheme". The "free scheme" is discussed in Chapter 2 above. These schools had an aggregate enrolment of 29,511.¹²⁹ Their overall perpupil expenditure was £75.90, including depreciation and religious imputations. Net expenditure (i.e., excluding depreciation) per pupil was £65.10. In principle, depreciation should be included, rather than excluded, in arriving at the current or operating costs of the schools. Alternatively, estimates of capital costs should be reported. We have excluded depreciation estimates because we have no reason to think that they were made on a consistent basis; and because other schools in the study, whose data were gathered by our own survey (i.e., lay Catholic, Protestant, and Jewish schools) did not report depreciation data. Net expenditures less religious imputations-which comes close to being an out-of-pocket, cash-payment concept of expenditures-was £59.40 on a per-pupil basis in these schools. Hence the average value of contributed services of religious, on a per-pupil basis, was £5.70 in day schools, free scheme, some of which were not operated by religious orders. These include four lay Catholic schools, and a number of Diocesan Colleges. To determine the privately-financed portion of Secondary School expenditures, we subtracted government grants, as reported by the schools, from their estimates of expenditures. The result, on an annual, per-pupil basis, was £11.70 in day schools, free scheme. Finally, to determine the outof-pocket, privately financed expenditure, we subtracted State grants from net expenditures less religious imputations. It will be noted that in our sample, day schools in the free scheme reported an amount equal to $-\pounds1.90$ per pupil. The sign indicates that these schools received more per pupil (and,

129. Enrolments are taken from Department of Education, List of Recognised Secondary Schools, 1974/75, Dublin (Stationery Office), 1976.

Table 5.10: Secondary School expenditure, various concepts, 1974/75, from sample of audited accounts, and supplementary survey, by type of school, totals for the State (Per pupil data in £s per year; total data in £1,000s per year)

| Type of School | No. of schools | No. of pupils | Total ex- pendi- ture ^a | Expendi- ture per pupil | Total net expendi- ture ^b | Net ex- pendi- ture per pupil | Total net expendi- ture less religious imputa- tions ^c | Net ex- penditure less relig- ious im- putations per pupil | Total own funds net expendi- ture ^d | Own funds net ex- penditure per pupil | Total own funds net expendi- ture less religious imputa- tions ^e | Own funds net ex- penditure less relig- ious im- putations per pupil |
|------------------------|-------------------|------------------|--|-------------------------------|--|--|--|---|---|--|---|--|
| All Secondary Schools | - | | | | | | | | | | | |
| in sample | 143 | 48,232 | 4,859.9 | 100.80 | 4,249.1 | 88.10 | 3,904.6 | 81.00 | 1,790.7 | 37.10 | 1,094.5 | 22.70 |
| Day schools, free | | | | | | | | | | | | |
| scheme | 84 | 29,511 | 2,240.0 | 75.90 | 1,921.9 | 65.10 | 1,753.5 | 59.40 | 346.5 | 11.70 | -57.1 | -1.90 |
| Schools with boarders, | | | | | | | | | | | | |
| free scheme | 47 | 15,738 | 1,658.2 | 105.40 | 1,380.5 | 87.70 | 1,253.1 | 79.60 | 600.5 | 38.20 | 371.1 | 23.60 |
| Fee-charging day | | | | | | | | | | | | |
| schools | 3 | 708 | 97.1 | 137.10 | 87.7 | 123.90 | 71.0 | 100.30 | 70.3 | 99.30 | 43.6 | 61.60 |
| Fee-charging schools | | | | | | | | | | | | |
| with boarders | 9 | 2,275 | 864.6 | 380.00 | 859.0 | 377.60 | 827.0 | 363.50 | 773.4 | 340.00 | 736.9 | 323.90 |
| All Protestant schools | 8 | 2,189 | 825.4 | 377.10 | 825.4 | 377.10 | 825.4 | 377.10 | 750.4 | 342.80 | 750.4 | 342.80 |
| All Catholic schools | 135 | 46,692 | 4,034.5 | 86.40 | 3,423.7 | 73.30 | 3,079.2 | 65.90 | 1,040.3 | 22.30 | 34.41 | 7.40 |
| Lay Catholic schools | 4 | 1,091 | 57.2 | 52.40 | 57.2 | 52.40 | 57.2 | 52.40 | -0.6 | -0.50 | -0.6 | -0.50 |

Note: Refer to footnotes at end of Table 5.11.

of course, in total) in grants from the State than they paid out in cash payments to others.

It would be hasty indeed to conclude that these schools were "operated at a profit", which a careless reading of these data would suggest. First, that would be true only if it were proper to ignore depreciation or other capital charges, and further to ignore the contributions of religious, both of which for a number of reasons would be unwarranted. And second, even were these contributions ignored, the point should be made that (as noted earlier-see Chapter 2), grants have been changed discontinuously, and apparent "profits" in 1974-75, in cash terms, were undoubtedly preceded in many schools by several years of "loss" in the same terms. Schools with boarders, participating in the free scheme (the free scheme as it applies to boarders is discussed in Chapter 2), are shown separately. No information was available on numbers of boarders at these schools: in principle, this figure could vary from 1 pupil to all the pupils in the school. For the most part, boarding costs are not educational expenditures. It is a highly arbitrary matter to distinguish between educational and non-educational expenditures. In principle, educational expenditures should include the costs of building, operating, maintaining, and staffing schools, and providing pupils with textbooks and other materials. Certain imputed expenditures, such as pupils' time used or earnings forgone, are added by some. Meals, transportation, uniforms, and the like are said not to be educational expenditures in this sense. Meals would be eaten, and clothing, if not uniforms, would be worn, even were there no schooling. The same cannot be said of transportation, however; it serves an educational, and exclusively educational, purpose. On some occasions, it might be more economical to board pupils than to transport them to school daily, and in this sense some boarding expenditures might well be regarded as educational in nature. In addition, some boarding expense is for staff, who supervise after hours study and perform other quasi-instructional duties. Our rule, however, is to exclude boarding expense. We present estimates of total educational outlays with estimated boarding cost outlays excluded. In effect, we treat the system of Secondary Schools as though all of them were day schools. However, boarding expense is reported, so that the reader who chooses to include them may do so. It is worth noting that net expenditures per pupil in boarding schools in our sample were £87.70, and net expenditures less religious imputations were £77.60, implying an average per pupil contribution by religious of £10.10. This is very close to the day school estimate of £11.70. (Virtually all of the schools in this category are Catholic Schools, operated by religious orders.)

Some interest might be attracted to a comparison of schools under Catholic and Protestant management. However, no direct comparison can be made. Protestant Schools participating in the "free scheme" do so quite differently from Catholic schools, and the State aid in respect of this scheme shows up as "fees" in the former. Additionally, the Protestant Schools in our sample were all boarding schools, and on *a priori* grounds seem likely to have a larger proportion of boarders than their Catholic counterparts. Thus Protestant Schools, while they spend considerably more per pupil than the Catholic Schools, do so on a different basis.

In Table 5.11, figures on day schools in the free scheme have been broken down by county or group of counties, as described earlier. Considerable regional difference in expenditures will be noted, in spite of the uniform grants scheme. Of fourteen areas, six are within 10 per cent of the statewide average of £65.10 per pupil in net expenditures; seven are lower-as low as £54.70 for Kilkenny-Wexford (one school only); one is higher-the highincome areas of Dublin. It will be noted in particular that high-income areas in Dublin spend $\pounds 25.50$ (43.7 per cent) more than low-income areas. Once again it must be emphasised that these figures are exclusive of incremental salaries, whose inclusion would make the relative (though perhaps not the absolute) difference between expenditures in these areas smaller. It is evident that the difference is mainly in the ability of the two Dublin areas to finance education expenditures with their own funds. Our data indicate that, on the average, expenditures in Counties Cork/Waterford, Tipperary, Clare/Limerick, Louth/Longford/Cavan/Meath/Monaghan, and Laois/Offaly/Carlow, were similar to those in low-income areas of Dublin. One should postpone interpreting these data until we have discussed our regression analyses.

Even more variability existed in different areas' expenditures from their own funds on education. For the State, this averaged $\pounds 11.70$ per pupil. In our sample, on an area basis, it ranged from $\pounds 0.70$ in Kilkenny/Wexford and $\pounds 0.80$ in Tipperary to $\pounds 27.20$ in high-income areas of Dublin.

Regression Analysis

Equations regressing per-school and per-pupil expenditures (using the same "expenditure" concepts just discussed) on a variety of independent variables were estimated by the method of ordinary least squares. There were two purposes behind these regressions. First, the underlying data represent a wealth of information not heretofore available for analysis. It was thought appropriate and useful to sort out these data, by multiple regression analysis, to see what patterns emerge. No hypotheses, strictly speaking, were tested; and virtually all available information was used in defining the independent variables. Instead of hypothesis testing, the method, which in substance examines the net effect on the dependent variable of each independent variable, while controlling for ("holding constant") the influence of the other variables included, simply provides an improved way of reading the data. Second, simplified versions of these equations were used to estimate State totals for each of the main expenditure concepts, thus advancing the principalpurpose of this chapter.

Tables 5.12, 5.13, 5.14, and 5.15 show regression results, for, respectively, all Secondary Schools; Catholic Secondary Schools; Catholic Day Secondary

| Counties | No. of schools | No. of pupils | Total ex- pendi- ture ^a | Expendi- ture per pupil | Total net expendi- ture ^b | Net ex- pendi- ture per pupil | Total net expendi- ture less religious imputa- tions ^c | Net ex- penditure less relig- ious im- putations per pupil | Total own funds net expendi- ture ^d | Own funds net ex- penditure per pupil | Total net funds net expendi- ture less religious imputa- tions ^e | Own funds net ex- penditure less relig- ious im- putations per pupil |
|---------------------------------------|-------------------|------------------|--|-------------------------------|--|--|--|---|---|--|---|--|
| Dublin, low-income | | | | | | | • | | | | | |
| areast | 6 | 3,004 | 195.3 | 65.00 | 175.0 | 58.30 | 145.9 | 48.60 | 7.3 | 2.40 | -21.7 | -7.20 |
| Dublin, High-income | | | | | | | | | * | | | |
| areast | 5 | 2,255 | 217.7 | 96.60 | 188.9 | 83.80 | 167.2 | 74.10 | 61.3 | 27.20 | 39.6 | 17.60 |
| Dublin, all other areas ^f | 14 | 6,616 | 533.5 | 80.60 | 469.2 | 70.90 | 402.4 | 60.80 | 115.3 | 17.40 | 48.5 | 7.30 |
| Kildare and Wicklow | 6 | 1,659 | 153.1 | 92.30 | 112.9 | 68.10 | 96.6 | 58.20 | 25.4 | 15.30 | 9.1 | 5.50 |
| Kilkenny and Wexford | ·· 1 | 278 | 16.3 | 58.60 | 15.2 | 54.70 | 115.0 | 41.40 | -0.2 | -0.70 | 3.9 | -14.00 |
| Cork and Waterford | 12 | 3,146 | 215.8 | 68.60 | 182.4 | 58.00 | . 171.1 | 54.40 | 39.4 | 12.50 | 4.9 | 1.60 |
| Tipperary | 4 | 1,719 | 114.5 | 66.60 | 94.0 | 54.70 | 77.5 | 45.10 | -1.5 | 3 -0.80 | -17.8 | -10.40 |
| Kerry | 5 | 1,449 | 99.8 | 68.90 | 91.1 | 62.90 | 76.3 | 52.70 | 11.3 | 7.80 | -3.5 | -2.40 |
| Clare and Limerick | 9 | 2,544 | 168.0 | 66.00 | 141.1 | 55.50 | 117.7 | 46.30 | 10.7 | 4.20 | 9.1 | -3.60 |
| Louth, Longford, Cavan Meath and | | | | | | | | | | | | |
| Monaghan | Ŕ | 1 1 7 8 | 79 4 | 64 70 | 68.6 | 58 50 | 53 4 | 45 50 | 94 | L 910 | _12.8 | |
| Laois Offaly and | 5 | 1,175 | 75.1 | 01.70 | 00.0 | 50.50 | 55.1 | 45.50 | · 2.7 | . 2.10 | -12.0 | -10.50 |
| Carlow | 9 | 802 | 61.6 | 76.80 | 47 1 | 58 70 | 36.6 | 45 60 | 4.8 | 6.00 | _5 7 | -7 10 |
| Galway | . 7 | 9 1 9 9 | 179.4 | 81.60 | 154.7 | 70.40 | 188.9 | 60.60 | 36.4 | 16.60 | 14.9 | 6.80 |
| Westmeath | , 9 | 409 | 31.0 | 75.80 | 97.4 | 67.00 | 94.3 | 59.00 | 3.5 | 8.60 | 0.4 | . 1.00 |
| Donegal, Leitrim, Sligo, Mayo, and | 4 | 105 | 51.0 | 75.00 | 21.1 | 07.00 | 41.5 | | | 0.00 | 0.1 | |
| Roscommon | 8 | 2,258 | 174.6 | 5 77.30 | 154.3 | 68.30 | 136.3 | 60.40 | 32.2 | 2 14.30 | 14.2 | . 6.30 |
| Total for State | 84 | 29,511 | 2,240.0 | 75.90 | 1,921.9 | 65.10 | 1,753.5 | 59.40 | 346.5 | 11.70 | -57.1 | -1.90 |

 Table 5.11: Secondary School expenditure, various concepts, 1974/75, from sample of audited accounts, day schools, free scheme only, by area

 (Per pupil data is £s per year; total data in £1,000s per year)

(a) Excludes incremental salaries; includes depreciation where charged; includes estimated value of services of religious (see text).

(b) Same as total expenditure (see note (a)), except depreciation excluded.

(c) Same as net expenditure (see note (b)), except estimated value of services of religious excluded (see text).

(d) Same as net expenditure (see note (b)), less grants from State.

(e) Same as net expenditure less religious imputations (see note (c)), less grants from State.

(f) See text, note 127.
Table 5.12: Regression results: Secondary School expenditures, various concepts, regressed on selected independent variables, 1974/75, all Secondary Schools (t-statistic in parentheses under estimated coefficient)

| (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) |
|-----------------------------------|------------------------------|------------------------------|--|------------------------------------|---|------------------------------------|---|---|--|
| •Independent variable | Total expendi- tures | Net expendi- tures | Net expend- itures less religious imputations | Own funds net expend- itures | Own funds net expend- itures less religious imputations | Per pupil net expend- itures | Per pupil net expend- itures less religious imputations | Per pupil own funds net expend- itures | Per pupil own funds net expendi- tures less religious imputations |
| Not convent school | -3.898 | -2.942 | 0.704 | -7.155 | -7.699 | 0.008 | 0.021 | -0.032* | -0.035* |
| Protestant management | (0.263) 16.001 (0.777) | (0.205) 21.634 (1.086) | (0.049) 25.863* (1.297) | (0.578) 13.120 (0.817) | (0.627) 20.461 (1.283) | (0.329) 0.133*** (3.834) | (0.871) 0.145^{***} (4.407) | (1.467) 0.235*** (8,388) | (1.630) 0.274*** (9.891) |
| Lay Catholic (proprietary) | -11,008 (0.764) | -6.760 (0.612) | -6.655 (0.602) | (0.011) -9.090 (0.845) | -8.399 (0.787) | (0.031) -0.040** (2.020) | (1.107) -0.043** (2.295) | (0.366) -0.026* (1.385) | (9.891) -0.025* (1.308) |
| Louth/Longford/Cavan/Meath/ | -5.768 | 4.207 | -3.634 | -0.550 | -0.237 | -0.010 | -0.010 | -0.021* | -0.021 [*] |
| Monaghan | (0.838 <u>)</u> | (0.632) | (0.545) | (0.074) | (0.032) | (0.850) | (0.859) | (1.586) | (1.637) |
| Kerry | 5.296 | 6.229 | 7.637 | 6.470 | 8.077 | 0.005 | 0.009 | -0.009 | -0.005 |
| W 111 /14/f1 | (0.605) | (0.735) | (0.900) | (0.719) | (0.905) | (0.315) | (0.596) | (0.558) | (0.105) |
| Kilkenny/wextord | 4.790 | 6.058 | 5.075 | 8.963 | 8.388 | 0.001 | -0.002 | -0.004 | -0.007 |
| Kildare/Wicklow | 11 850** | (0.702) | (0.588) | (0.967) | (0.912) | (0.088) | (0.168) | (0.228) | (0.414) |
| Kildale/ Wicklow | (1 716) | (1.964) | (1 409) | (1.480) | (1 665) | (1.906) | (1.675) | -0.008 | (0.013) |
| Dublin: High income areas | 24.216*** | 24.508*** | 24.935*** | 22.194*** | 99.755*** | 0.034*** | 0.034*** | 0.018** | 0.985 |
| 0 | (3.144) | (3.287) | (3.342) | (2.755) | (2.845) | (2.623) | (2.717) | (1.237) | (1.283) |
| Dublin: Low income areas | -9.357 | -7.127 | -6.378 | $-5.355^{'}$ | -4.452 | -0.012 | -0.010 | -0.022* | -0.020* |
| | (1.207) | (0.949) | (0.850) | (0.683) | (0.572) | (0.890) | (0.774) | (1.566) | (1.437) |
| Clare/Limerick | 2.483 | 2.874 | 4.036 | 5.480 | 6.901 | -0.006 | -0.005 | -0.015 | -0.010 |
| | (0.412) | (0.493) | (0.692) | (0.826) | (1.047) | (0.536) | (0.050) | (1.252) | (0.833) |
| Cork/Waterford | 2.274 | 4.212 | 4.144 | 5.486 | 5.411 | 0.009 | 0.009 | -0.008 | -0.010 |
| Calmer | (0.424) | (0.812) | (0.798) | (0.906) | (0.900) | (0.952) | (1.038) | (0.788) | (0.958) |
| Galway | 0.492 | 0.215 | 1.789 | 1.722 | 3.594 | -0.006 | 0.001 | 0.015 | 0.010 |
| Leitrim/Roscommon/Mayo/ | (0.078) | (0.035) | (0.294) | (0.248) | (0.522) | (0.526) | (0.020) | (1.277) | (0.800) |
| Sligo/Donegal | (0 708) | -0.778 | (0.341) | 4.332 | 5.540 | (1, 196) | -0.003 | -0.012 | -0.009 |
| Westmeath | 1.133 | 1.043 | 0.644 | 0.601 | 0.485 | (1.120) | (0.978) | (1.076) | (0.824) |
| | (0.148) | (0.141) | (0.087) | (0.073) | (0.059) | (0.149) | (0.212) | (1.194) | (1.947) |
| Laois/Offaly/Carlow | 12.011 | 5.803 | 3.649 | 6.710 | 4.825 | 0.023 | 0.013 | 0.009 | -0.001 |
| | (1.110) | (0.554) | (0.348) | (0.616) | (0.446) | (1.232) | (0.725) | (0.481) | (0.033) |
| Tipperary | -1.552 | -2.060 | 1.103 | -0.712 | 0.438 | -0.007 | -0.034 | -0.018 | -0.015 |
| Number of pupils | (0.161) 0.108*** | (0.221) 0.100*** | (0.118) 0 894*** | (0.073) 0.043*** | (0.046) 0.033*** | (0.405) | (0.215) | (1.071) | (0.896) |
| | (9.043) | (8.619) | (7.731) | (3.564) | (2.776) | | | | |
| All boys | 8.374 | 6.219 | 4.522 | -3.504 | -3.535 | 0.002 | -0.091 | 0.039** | 0.047** |
| | (0.590) | (0.452) | (0.329) | (0.298) | (0.303) | (0.067) | (0.399) | (1.909) | (2.323) |
| Any boarders | 16.326*** | 13.414*** | 12.604*** | 10.035*** | 8.923*** | 0.035*** | 0.032*** | 0.027*** | 0.022*** |
| | (4.458) | (3.785) | (3.554) | (2.748) | (2.462) | (5.618) | (5.360) | (4.128) | (3.468) |
| All boarders | 7.743 | 8.231 | 1.629 | -13.717 | -17.994* | 0.253 * * * | 0.198*** | 0.290*** | 0.257*** |
| | (0.444) | (0.488) | (0.097) | (1.002) | (1.324) | (8.477) | (6.978) | (11.977) | (10.739) |
| Not in free scheme (fee-charging) | 38.164*** | 37.572*** | 32.160*** | 53.062*** | 47.162*** | 0.058*** | 0.039** | 0.057*** | 0.031** |
| Par cont low too shing staff | (3.407) | (3.527) | (3.017) | (5.198) | (4.654) | (3.053) | (2.137) | (3.150) | (1.735) |
| Ter cent lay teaching stari | (1 800) | 10,110* | 24.160** | 12.838 | 20.821** | 0.055*** | 0.082*** | 0.044^{**} | 0.071*** |
| Pupils per teacher | (1.609) | (1.500) | (2.047) | (1.102) | (1.800) | (2.604) | (4.131) | (2.116) | (3.503) |
| Firs her consister | (1.001) | (1.126) | (1.151) | (1.599) | -0.037* (1.596) | -0.001 | -0.001 (0.790) | 0.001 | -0.001 |
| Intercept (includes rest of | -21.453* | -17.117* | -23.476** | -0.804 | -17.307 | 0.032* | 0.001 | 0.001** | (0.047) |
| Co. Dublin, etc.)+ | (1.604) | (1.322) | (1.813) | (1.522) | (1.247) | (1.402) | (0.048) | (1.986) | (2.364) |
| \overline{R}^2 adj. for d.f. | .623 | .625 | .614 | .570 | .781 | .841 | .845 | .569 | .784 |
| d.f. | 119 | 119 | 119 | 119 | 119 | 120 | 120 | 120 | 120 |
| | | | | | | | | - | |

All independent variables except number of pupils, per cent lay teaching staff, and pupils per teacher are, "dummy" variables, set equal to 1.0 Note: when the quality indicated is present. +Intercept includes convent school; Catholic management (other than lay, proprietary); rest of County Dublin; day only; not all boys (i.e., all girls

or mixed); and in free scheme. *Significant at 90%. **Significant at 95%.

***Significant at 99%.

Schools; and Boarding Schools (Catholic and Protestant). Inadequate data made it impossible to make separate analyses of Protestant Secondary Schools.

It may be useful to spend a moment explaining these tables to the nonstatistical reader. In doing so, we use as an example Column 7 (per pupil net expenditures) in Table 5.12. The first twenty variables listed (starting with "Not convent schools" and ending with "Not in free scheme", and skipping "Number of pupils", which is not used in this regression) are "dummy variables," whose use will be explained in a moment. The next three lines of this column show, in reverse order, an estimated intercept value of 0.032, an estimated coefficient of -0.007 for pupils per teacher, and an estimated coefficient of 0.055 for per cent lay teaching staff. For reasons to be explained presently, these three by themselves can be taken to be the per pupil school expenditure equation for schools which do not have the properties indicated in the twenty dummy variables. This means that the equation applies to convent schools under Catholic management, in the "other" (neither high- nor low-income) areas of County Dublin, which accept day pupils only, and participate in the "free scheme". For schools meeting this description, a school having 20 pupils per teacher and 60 per cent lay teachers would spend, on average, $\pounds 66.00$ per pupil $(0.032 - [0.001 \times 0.20] + [0.055 \times 0.60] =$ 0.066, or £66, as the data are in thousands of pounds.)

The estimated coefficients of the dummy variables show how much must be added (or subtracted) for schools not of this description. For example, add £8 if the school is not a convent school; add £133 if it is a Protestant School; subtract £40 if it is a lay Catholic School; etc. More than one such addition can be made—e.g., to determine the per-pupil cost of schools in County Kerry, not convent schools, accepting boarders. Only one regional dummy may be used, and other combinations are precluded where they are logically ruled out (e.g., a school cannot be both Protestant and lay Catholic, cannot be both "any boarders", which means accepting both day and boarding pupils, and "all boarders").

The values of the coefficients are, in effect, averages for each class of school. The asterisks and t-statistics in parentheses under the coefficients reflect statistical significance. In general, the higher the t-statistic, the more confidence may be placed in the coefficient. Where three asterisks (***) appear, the estimated coefficient would appear in our study, even were the true coefficient nil or zero (reflecting no relationship), fewer than one time in 100. This is expressed as a significance level of 99 per cent. Two asterisks and one asterisk reflect 95 per cent and 90 per cent significance levels, respectively.

One other statistic in Table 5.12 may be of interest, \overline{R}^2 (on the penultimate line), which, in Column 7, shows a value of .841. This statistic indicates what fraction of the variation in the dependent variable (per pupil net expenditure) is accounted for by the independent variables. In this case, our equation, and its variables, explain 84.1 per cent of the variation from school-toschool of per-pupil net expenditures. As such studies go, this is an extremely high value for \overline{R}^2 .

With this background, we may summarise the results of Tables 5.12 through 5.15.

1. Table 5.12. All Secondary Schools, Catholic and Protestant, day and boarding.

(a) Expenditures per school. Statistically significant determinants of expenditures per school include number of pupils; per cent lay teaching staff; not in free scheme; and any boarders. In addition, high income areas in Dublin and possibly Counties Kildare and Wicklow spend more per school than other areas. The difference between Columns 2 and 3 represents estimated depreciation. The difference between Columns 3 and 4 is religious imputations. The difference between 3 and 5 is State grants. And the difference between 3 and 6 is State grants and religious imputations. These differences apply in all four tables.

(b) Expenditures per pupil. Statistically significant determinants include per cent lay teaching staff; not in free scheme; any boarders; all boarders; Protestant management; and lay Catholic management. The last of these has a negative influence. In addition, high income areas of Dublin spend significantly more per pupil. The difference between Columns 7 and 8 is perpupil religious imputations; between 7 and 9, per-pupil State grants; and between 7 and 10, both religious imputations and State grants. These differences apply in all four tables.

2. Table 5.13. This table reports on regression equations for Catholic Secondary Schools only. That is, the sample consists of the same schools as Table 5.12, except that Protestant Schools are omitted. Protestant Schools seem to differ in a number of respects from Catholic Schools, not only in management but in acceptance of boarders; relation to the "free scheme"; absence of religious teachers; and per pupil average expenditure. This means that Table 5.13 reports on a more homogeneous group of schools than does Table 5.12.

(a) Expenditures per school. The importance of some additional variables in explaining differences in per-school spending *among* Catholic schools comes to light in Columns 2 through 6. Notably, lay Catholic schools spend significantly less than those operated by orders and by Dioceses; significantly more is spent, per school, in Kildare/Wicklow, and significantly less in Louth/ Longford / Cavan / Meath / Monaghan, in Leitrim / Roscommon / Mayo / Sligo / Donegal, and in the low-income areas of Dublin.

(b) Expenditures per pupil. In addition to the variables found significant in Table 5.12, some additional regional variables come to light. More is spent per pupil in Kildare/Wicklow, the high income areas in Dublin, and possibly in Laois/Offaly/Carlow; less is spent per pupil in Louth/Longford/Cavan/ Meath/Monaghan, in the low-income areas in Dublin; and in Leitrim/Roscommon/Mayo/Sligo/Donegal.

| (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) |
|--|---|--|---|--|---|---|---|---|--|
| Independent variable | Total expendi- tures | Net expendi- tures | Net expend- itures less religious imputations | Own funds net expend- itures | Own funds net expend- itures less religious imputations | Per pupil net expend- itures | Per pupil net expend- itures less religious imputations | Per pupil own funds net expend- itures | Per pupil own funds net expendi- tures less religious imputations |
| Not convent school | -0.725 | 0.400 (0.050) | 2.905 (0.394) | -0.601 | 1.898 (0.257) | 0.005 (0.286) | 0.002 | -0.001 | 0.012 (0.690) |
| Protestant management Lay Catholic (proprietary) | | -10.805** | | -10.020* | -10.046** | -0.034*** | -0.042*** | -0.354*** | -0.383** |
| Louth/Longford/Cavan/Meath/ Monaghan Kerry | (2.024) -6.715* (1.527) 0.256 | (1.729) -5.227* (1.388) 0.890 | (1.905) -4.917* (1.432) 1.785 | (1.605) -4.802 (1.115) 0.876 (1.75) | (1.757) -4.358 (1.105) 1.904 (2.102) | (2.211) -0.012* (1.329) -0.001 | (3.278) -0.120* (1.535) 0.002 | (2.445) -0.013 (1.266) -0.003 | (2.919) -0.011 (1.269) 0.002 |
| Kilkenny/Wexford | (0.046) 3.614 (0.635) | (0.185) 4.783 (0.981) | (0.406) 3.540 (0.797) | (0.170) 5.635 (1.062) | (0.402) 4.523 (0.931) | (0.065) 0.001 (0.070) | (0.233) 0.003 (0.352) | (0.231) 0.002 (0.229) | $(0.144) \\ -0.001 \\ (0.384)$ |
| Kildare/Wicklow | `7.860 [*] * (1.773) | 4.224 (1.112) | 4.764 [*] (1.376) | 5.672*(1.310) | 6.354^{*} | 0.011* (1.287) | `0.015** (1.961) | 0.014* (1.360) | `0.019*** (2.095) |
| Dublin: High income areas | 5.965 (1.080) | 5.121 (1.082) | 5.482' (1.271) | `4.303´ (0.866) | 4.769' | 0.014* (1.302) | -0.015^{*} (1.554) | 0.010 | 0.012' |
| Dublin: Low income areas | -9.847** | -7.689** | -7.009* | -7.800** | -7.031** | -0.015* | -0.014* | -0.018** | -0.015* (1.647) |
| Clare/Limerick | -0.488 | -0.265 | 0.544 | (0.125) | 1.422 | -0.009 | 0.004 | -0.008 | -0.003 |
| Cork/Waterford | -1.191 | 0.648 | -0.039 | 1.486 | 0.934 | 0.002 | -0.001 | (0.347) 0.002 | (0.321) (0.003) |
| Galway | -2.624 | -3.101 | (0.014) -1.946 | -2.498 | (0.270) -1.204 | -0.009 | -0.004 | 0.010 | -0.003 |
| Leitrim/Roscommon/Mayo/ Sligo/Donegal Westmeath | (0.648) -1.660* (1.402) 3.674 | (0.894) -1.155 (1.138) 3.273 (0.792) | (0.815) -1.052 (0.138) 2.373 (0.522) | (0.622) -0.610 (0.164) 2.421 (0.422) | (0.327) -0.240 (0.071) · 1.669 (0.021) | (1.135) -0.004* (1.582) 0.003 (2.222) | (0.808) 0.003* (1.531) 0.001 | (1.021) -0.006 (0.706) -0.003 (0.003) | (0.398) -0.003 (0.377) -0.003 |
| Laois/Offaly/Carlow | (0.704) 7.052* (1.309) | (0.732) -2.657 (0.449) | (0.582) 0.107 (0.020) | (0.486) 2.539 (0.408) | (0.364) 0.122 (0.021) | (0.332) 0.019* (1.406) | (0.155) 0.008 (0.687) | (0.230) 0.015 (1.030) | (0.301) 0.005 (0.407) |
| Tipperary | -4.235 (0.689) | -4.902 (0.931) | $-4.173 \\ (0.870)$ | 5.446 (0.987) | -4.608 (0.912) | -0.011 (0.870) | 0.008 (0.715) | $0.014 \\ (1.095)$ | 0.010 (0.857) |
| Number of pupils | 0.090*** (11.448) | 0.081*** (11.989) | 0.069*** (11.228) | • 0.027*** (3.812) | • 0.015*** (2.372) | - | _ | _ | _ |
| All boys | 5.496 (0.607) | $3.182 \\ (0.411)$ | 1.295 (0.183) | 4.641 (0.600) | 2.755 (0.389) | 0.001 (0.076) | -0.006 (0.368) | $0.009 \\ (0.497)$ | 0.001 (0.089) |
| Any boarders | 12.811*** (5.379) | 9.691*** (4.750) | * 8.716*** (4.687) | * 8.482*** (4.063) | * 7.487*** (3.917) | 0.030*** (6.255) | 0.026^{***} (6.210) | 0.027*** (5.484) | 0.023*** (5.200) |
| All boarders | 25.918** (2.160) | 27.046*** (2.631) | 24.182*** (2.587) | * 17.401** (1.685) | 14.454* (1.528) | 0.264*** (11.078) | 0.219*** (10.304) | 0.238*** (9.916) | 0.191*** (8.812) |
| Not in free scheme (fee-charging) | 19.239** (2.301) | 17.883*** (2.497) | 8.706* (1.334) | 27.961** [;] (3.781) | * 18.923*** (2.794) | 0.054*** (3.245) | -0.025** (1.722) | 0.083*** (4.815) | 0.055*** (3.573) |
| Per cent lay teaching staff | 21.827*** (2.804) | 15.935*** | 23.628 ^{***} (3.887) | * 13.673** (2.062) | 21.397*** | 0.051*** | 0.078*** | 0.045*** (2.909) | 0.072*** (5.134) |
| Pupils per teacher | -0.472^{*} | -0.526* | -0.550** | -0.408* | -0.434* | -0.001* | -0.001 ** | -0.001 | -0.001 |
| Intercept (includes rest of County Dublin, etc.)+ \vec{R}^2 adj. for d.f. d.f. | (1.336) -12.157* (1.369) .623 111 | (0.944) (0.944) .635 111 | $(1.55)^{+}$ $(1.164*)^{-12.164*}$ $(1.155)^{-1.155}$ $.609^{-1.11}$ | (1.314) -8.169 (1.211) .426 111 | -13.323** (1.908) .356 111 | 0.046*** (2.679) .797 112 | 0.018 (1.161) .756 112 | -0.011 (1.161) .791 111 | -0.041** (1.937) .743 111 |
| | | | | | | ·· • | | | |

.

~

 Table 5.13: Regressions results: Secondary School expenditures, various concepts, regressed on selected independent variables, 1974/75, Catholic schools only.

 (t-statistic in parentheses under estimated coefficient)

3. Table 5.14. This table reports on regression equations for Catholic Day Schools only. That is, the sample consists of the same schools as Table 5.12, except that schools accepting boarders are omitted. Since all the Protestant Schools in our sample accepted boarders, this means that all Protestant Schools are eliminated, as are Catholic Schools accepting boarders. These schools constitute a still more homogeneous group than those in Table 5.13.

(a) Expenditures per school. In general, the results are similar to those in the previous tables. For the first time, County Tipperary shows a significant coefficient: the Catholic Day Schools in this County spend significantly less than other counties. So perhaps do those in Clare/Limerick.

(b) Expenditures per pupil. As above, Tipperary and possibly Clare/ Limerick are added to the list of schools spending significantly less, per pupil.

4. Table 5.15. This table reports on regression equations for schools accepting boarders. This category includes both day-cum-boarding schools and boarding-only schools; and both Protestant and Catholic schools (but not lay Catholic schools).

(a) Expenditures per school. Schools accepting boarders spend in proportion to the number of pupils. Those not in the free scheme spend thousands more per school than those in the free scheme. Regionally, schools in the high income areas of Dublin spend more, and those in Louth/Longford/ Cavan/Meath/Monaghan spend less, than elsewhere.

(b) Expenditures per pupil. Schools accepting boarders spend more per pupil as the per cent lay in their teaching staffs increases; if they are not in the free scheme they spend in excess of $\pounds 200$ more per pupil; and the schools in the high income areas in Dublin spend substantially more than others.

A number of comments on the findings of Tables 5.12 through 5.15 are in order. First, we have concentrated, in the last several paragraphs, on those independent variables whose coefficients were statistically significant. In the case of other variables, their coefficients, in spite of the lack of statistical significance, represent the best estimate of the average net relationship between them and the respective dependent variables. These other variables should not be ignored.

Second, some interest attaches to the results shown in all four tables for the variable, 'Per cent lay teaching staff.' The results show a considerable, and usually very significant, effect of this variable, even when religious imputations are included. The results can be taken as an estimate of the cost savings which result from the use of religious teachers. Our study strongly suggests that the valuations the religious have made of their own services, which we have called 'religious imputations', substantially understates their economic worth.

And finally, the regressions show (as do Tables 5.10 and 5.11) large disparities in Secondary School expenditures in the State. More is spent in respect of pupils in Protestant than Catholic Schools, as has been noted. Considerably more is spent in respect of pupils in fee-charging schools than

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

in schools participating in the free scheme. And there are substantial regional disparities within the State. Perhaps the most striking are the differences within metropolitan Dublin, between high-income and low-income areas. While our study does not permit conclusions about differences in quality of education, or in lifelong income-earning potential, differences in educational opportunity according to religion, social class, or even region (or neighbourhood) of residence appear to exist which seem contrary to egalitarian standards.

Estimated Total Expenditures on Secondary Schools

Table 5.16 represents a 'boiled down' version of Table 5.12, in which a number of smaller equations are estimated, for later use in calculating total Secondary School expenditures, according to the concepts of expenditures used above. Some variables found in Table 5.12 were excluded in these new regressions because they lacked statistical significance in Table 5.12. In addition, two variables — pupils per teacher and per cent lay teaching staff — were also excluded, because the distribution of these characteristics in the population is not known, and hence these could not be used in 'grossing up' estimates of total expenditures. The equations reported in Columns 6, 7, 8, and 9 in Table 5.16 are used incalculating the data in Table 5.17, using the student numbers found in the column headed 'No. of pupils' in Table 5.17.

The results, as shown in Table 5.17, constitute our estimate of Secondary School expenditures for 1974/75. We estimate total expenditure to be £12,772,594, exclusive of incremental salaries paid directly by the State, and exclusive of additional costs (shown separately) associated with boarding. Our estimate is that an additional £2,015,772 is associated with the boarding function of schools accepting day and boarding pupils; and £623,822 is associated with the boarding function of schools which accept only boarders. To put the matter slightly differently, we estimate that if all the Secondary Schools had only day pupils, total net expenditure would have been £12,772,594. Whether the £2,639,594 additional estimated boarding expense should be considered an 'educational' expense is arguable; in the event, the data are reported separately, so that the reader may choose.

In the next column, total net expenditure, again excluding boarding expense, and less religious imputations, is estimated at $\pounds 10,821,370$; and, when boarding expense is included, at $\pounds 13,026,970$. The differences between these two pairs of numbers constitute estimates of the total value of religious imputations, in effect the contributed services of religious. When boarding expense is ignored this value is estimated at $\pounds 1,951,224$; where boarding expense is included, the value rises to $\pounds 2,385,218$. We have already noted, however, that the strength and significance of the 'per cent lay teaching staff', variable strongly suggests that these figures are underestimates of the economic contribution of religious to Secondary Schools in Ireland.

In the next column, we report estimates of total net expenditures, own

| Table 5.14: Regression results: Sec | ondary School | expenditure | s, various con | cepts, regress | ed on selected | l independent | variables, 19 | 74/75, Cathol | ic day schools |
|--|----------------------------|--------------------------|--|------------------------------------|---|------------------------------------|---|--|--|
| | | (t-statistic | in parenthese | s under estim | ated coefficien | ıt) | | <u> </u> | |
| | | | | | | | | | |
| (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) |
| Independent variable | Total expendi- tures | Net expendi- tures | Net expend- itures less religious imputations | Own funds net expend- itures | Own funds net expend- itures less religious imputations | Per pupil net expend- itures | Per pupil net expend- itures less religious imputations | Per pupil own funds net expend- itures | Per pupil own funds net expendi- tures less religious imputations |
| Not convent school | -6.528 (0.788) | -1.878 (0.321) | 0.582 (0.094) | -3.556 (0.615) | -1.106 | 0.011 (0.659) | 0.012 (0.755) | 0.007 (0.451) | 0.006 (0.156) |
| Protestant management | _ | | | _ | | (| | (| |
| Lay Catholic (proprietary) | -5.829 | -4.448 | -3.631 | -4.283 | -3.469 | -0.025*** | • -0.021** | -0.020** | -0.024*** |
| | (1.121) | (1.212) | (0.932) | (1.179) | (0.888) | (2.568) | (2.077) | (1.953) | (2.463) |
| Louth/Longford/Cavan/Meath/ | -1.929 | -1.321 | -2.826 | -2.618 | -4.126 | -0.0002 | -0.007 | -0.013* | -0.006 |
| Monaghan | (0.375) | (0.363) | (0.732) | (0.727) | (1.066) | (0.022) | (0.766) | (1.364) | (0.640) |
| Kerry | -4.339 | -3.022 | -2.015 | -3.668* | -2.662 | -0.007 | -0.043 | -0.007 | -0.009* |
| | (1.181) | (1.165) | (0.732) | (1.428) | (0.963) | (1.072) | (0.638) | (0.985) | (1.430) |
| Kilkenny/Wexford | -0.125 | -4.023 | -5.333 | -5.364 | -0.005 | -0.013 | -0.155 | -0.020^{*} | -0.017^{*} |
| Vildone Winklow | (0.892) | (0.850) | (1.057) | (1.118) | (1.292) | (1.015) | (1.215) | (1.510) | (1.331) |
| Kildare/wicklow | (0 500) | (0 500) | -0.942 | -1.408 | -0.909 | (0.381) | (0.133) | (0.148) | -0.002 |
| Dublin: High income areas | 6 6 7 5 * * | 5 5 90** | 6 1 7 4 * * * | (0.007) 3 895* | 4 4 8 4 * * | 0146** | 0.017*** | * 0.013** | 0.011** |
| Dubinit High meome areas | (1.991) | (2.333) | (2.459) | (1.633) | (1.779) | (2,292) | (2.593) | (1.983) | (1.700) |
| Dublin: Low income areas | 7.940*** | -5.957** | ·6.025*** | * -7.453** | * -7.522*** | ·0.011** | -0.011** | -0.015** | -0.015*** |
| | (2.470) | (2.625) | (2.502) | (3,318) | (3.112) | (1.871) | (1.812) | (2.371) | (2.460) |
| Clare/Limerick | -3.674 | -3.826* | -3.348* | -3.907** | -3.426* | -0.010** | 0.006 | -0.007 | -0.011** |
| | (1.122) | (1.655) | (1.364) | (1.707) | (1.392) | (1.757) | $(1.069)^{\circ}$ | (1.174) | (1.875) |
| Cork/Waterford | -3.936* | -2.108 | -1.816 | -2.085 | -1.798 | -0.001 | -0.0004 | -0.0002 | -0.001 |
| | (1.319) | (1.000) | (0.812) | (1.000) | (0.801) | (0.244) | (0.079) | (0.040) | (0.200) |
| Galway | -0.970 | 0.952 | -0.228 | -1.146 | -0.418 | -0.002 | 0.001 | -0.0004 | 0.003 |
| | (0.301) | (0.419) | (0.094) | (0.510) | (0.173) | (0.418) | (0.219) | (0.072) | (0.565) |
| Leitrim/Roscommon/Mayo/ | -1.957 | -1.259 | -0.857 | -1.971 | -1.575 | -0.002 | 0.0003 | -0.002 | -0.004 |
| Sligo/Donegal | $(0.612)_{,}$ | 0.558 | (0.358) | (0.883) | (0.656) | (0.357) | (0.058) | (0.434) | (0.735) |
| Westmeath | -2.677 | -2.249 | -1.196 | -3.754 | -2.702 | 0.005 | -0.0002 | -0.006 | -0.010 |
| | (0.517) | (0.615) | (0.308) | (1.037) | (0.694) | (0.503) | (0.024) | (0.605) | (1.094) |
| Laois/Offaly/Carlow | -2.705 | -6.021** | -5.135° | $-5.4/0^{+}$ | -4.586 | -0.012 | -0.008 | -0.007 | -0.011 |
| Tinnerse | (0.555) 5 994* | (1.080) | (1.350) | (1.044) | (1.202) * 7.966*** | (1.220) | (0.040) | (0.703) | (1.000) |
| Tippetary | (1.382) | (2 252) | (1 990) | (2 860) | (2 5 2 8) | (1696) | (1.308) | (1 000) | (9.990) |
| Number of pupils | 0.077*** | 0.068*** | * 0.057*** | * 0.014** | * 0.003 | (1.080) | (1.550) | (1.505) | (2.220) |
| ramber of papils | (11.518) | (14.545) | 11.351 | (3.065) | (0.502) | | | | |
| All boys | 5.741 | 2.357 | -1.203 (0.2065) | 4.846 | (0.202) (0.222) | -0.012 (0.820) | -0:014 (0.918) | 0.007 (0.485) | -0.006 |
| Any boarders | | _ | _ | _ | _ | _ | _ | | _ |
| All boarders | _ | | , | _ | — | | _ | _ | _ |
| Not in free scheme (fee charging) | 17.587*** | 15.335** | * 6.839** | 24.422** | * 15.924*** | * 0.051*** | * 0.025*** | * 0.053** | * 0.079*** |
| Per cent lay teaching staff | (3.429) 4.941 | (4.236) 1.237 | (1.780) 1.620*** | (6.816) * 0.605 | (4.131) 8.972** | $(5.408) \\ -0.001$ | (2.600) 0.345*** | (5.428) * 0.034** | (25.867) * 0.011 |
| | (0.791) | (0.280) | (2.055) | (0.139) | (1.910) | (0.923) | (2.890) | (2.831) | (0.903) |
| Pupils per teacher | 0.512** | -0.418** | -0.320* | -0.205 | 0.107 | 0.515** | -0.001** | 0.001 | -0.001 |
| The second distance of the | (1.702) | (1.969) | (1.419) | (0.977) | (0.473) | (2.295) | (2.045) | (0.964) | (1.217) |
| Intercept (includes rest of | /./94 (1.09r) | 1.856 | 0.189 | 4.956 | -2.702 | 0.085** | • U.U53*** | [™] 0.010 [™] [™] 0.010 [™] | 0.022 |
| County Dublin, etc.)+ $\overline{\mathbf{p}}^2$ add for d f | (1.035) 775 | (1.4//) | (0.034) | (0.109) | (0.994) | (0.007) | (3./38) | (0.189) | (1.150) |
| A f | 64 | .077 64 | 64 | .557 64 | 6A. | .440 65 | .272 | .004 | .505 |
| u.i. | UT | UT | JT | 04 | 04 | 00 | 05 | 00 | 05 |

Note:

+Intercept includes convent school; Catholic management (other than lay, proprietary); rest of County Dublin; day only; not all boys (i.e., all girls or mixed); and in free scheme.

*Significant at 90%.

**Significant at 95%.

***Significant at 99%.

All independent Variables except number of pupils, per cent lay teaching staff, and pupils per teacher are "dummy" variables, set equal to 1.0 when the quality indicated is present.

| (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) |
|---|------------------------------|-----------------------------|--|------------------------------------|---|------------------------------------|---|---|--|
| Independent variable | Total expendi- tures | Net expendi- tures | Net expend- itures less religious imputations | Own funds net expend- itures | Own funds net expend- itures less religious imputations | Per pupil net expend- itures | Per pupil net expend- itures less religious imputations | Per pupil own funds net expend- itures | Per pupil own funds net expendi- tures less religious imputations |
| Not convent school | -19.908 | -21.723 | -18.797 | -22.578 | -19.654 | -0.006 | 0.017 | -0.008 | 0.015 |
| Protestant management | (0.037) -4.248 (0.119) | (0.788) 1.988 (0.043) | (0.088) 8.823 (0.259) | (0.788) 8.745 (0.244) | (0.690) 16.085 (0.452) | (0.105) -0.056 (0.789) | (0.344) -0.018 (0.262) | (0.136) -0.016 (0.221) | (0.291) 0.022 (0.841) |
| Lay Catholic (proprietary) | (01110) | (0.010) | (0.255) | (0.211) | (0.454) | (0.789) | (0.202) | (0.221) | (0.841) |
| Lougford/Cavan/Meath/ Monaghan | ` —20.449* (1.541) | $-17.937 \ (1.281)$ | -17.208*(1.358) | -17.987*(1.314) | $\stackrel{-}{-16.759}(1.266)$ | | -0.019(0.834) | | |
| Kerry | - | _ | | | | | — | - | - |
| Kilkenny/wextord | -2.134 (0.129) | -0.508 (0.032) | 2.499 (0.757) | 0.278 (0.017) | -1.713 (0.104) | -0.004 (0.111) | —0.140 (0.460) | -0.001 (0.028) | -0.011 (0.350) |
| Kildare/Wicklow | 10.093 (0.591) | 7.219 (0.438) | 7.900 (0.483) | 9.518 (0.555) | 10.199 (0.599) | 0.031 (0.878) | 0.034 (1.075) | 0.037 (0.996) | 0.040 (1.208) |
| Dublin: High income areas | 58.209*** (2.530) | 61.214*** (2.757) | 57.472*** (2.609) | 62.701*** (2.716) | 58.959*** (2.569) | 0.125^{***} (2.643) | 0.098** (2.347) | 0.121** (2.445) | 0.093** (2.142) |
| Dublin: Low income areas | -33.423 (1.202) | -31.082 (1.158) | -27.982 (1.051) | -29.201 (1.046) | -26.101 (0.940) | -0.034 (0.604) | -0.029 (0.594) | -0.028 (0.488) | -0.024 (0.464) |
| Clare/Limerick | -13.074 (0.835) | -10.597 (0.701) | -9.929 (0.662) | -10.413 (0.663) | -9.746 (0.624) | -0.016 (0.499) | -0.015 (0.549) | -0.015 (0.444) | -0.014 (0.485) |
| Cork/Waterford | -5.794 (0.439) | -3.24 (0.254) | -6.480 (0.513) | -3.225 (0.245) | -6.462 (0.491) | 0.025 | 0.012 | 0.025 | 0.013 |
| Galway | -11.264 (0.776) | -12,083 (0.862) | -10.059 (0.724) | -11.948 (0.820) | -9.924 | -0.024 | -0.018 | 0.023 | -0.018 |
| Leitrim/Roscommon/Mayo/ | -2.911 | 2.352 | -2.300 | -2.237 | 2 184 | -0.004 | _0.004 | 0.008 | (0.045) |
| Sligo/Donegal | (1.010) | (0.846) | (0.834) | (0.774) | (0.760) | (0.616) | (0.718) | -0.003 | -0.003 |
| Westmeath | 2.380 | -3.266 | -5.045 | -4 379 | -6 1 5 1 | 0.010 | 0.009 | (0.551) | (0.013) |
| | (0.161) | (0.229) | (0.357) | (0.205) | (0.4.1.9) | -0.001 | -0.008 | 0.004 | 0.011 |
| Laois/Offaly/Carlow | 26.786 (1.021) | (0.6567) (0.654) | 8.226 (0.328) | (0.235) 14.007 (0.532) | 5.671 | 0.065 | (0.231) 0.230 (0.479) | (0.120) 0.057 (0.924) | (0.375) 0.010 (0.205) |
| Tipperary | <u> </u> | _ | (0.0-40) | (0.001) | (0.21.) | | (0.175) | (0.524) | (0.205) |
| Number of pupils | ° °0.147*** (5.311) | 0.136*** (5.101) | 0.126*** (4.767) | 0.082*** (2.957) | 0.092*** (2.615) | _ | - | _ | _ |
| All boys | 27.528 (1.005) | 25.042 (0.947) | 24.334 (0.927) | 25.681 (0.934) | 24.976 | 0.024 | 0.011 | 0.027 | 0.013 |
| Any boarders All boarders | | | | | - | (0.117) | (0.220) | (0.170) | (0.270) |
| Not in free scheme (fee-charging) | 68.510*** (4.086) | 69.682*** (4.305) | 63.312*** (3.943) | 71.308*** (4.238) | 64.937*** (3.882) | 0.243*** | 0.194*** | 0.239*** | 0.190*** |
| Per cent lay teaching staff | 30.776 (1.137) | 23.752 | 32.740 (1.263) | 21.074 | 30.061 | 0.077* | 0.117** | 0.069 | (0.110** |
| Pupils per teacher | -0.629 | -0.809 | -0.855 | -0.695 | -0.742 | -0.002 | (2.302) 0.002 | (1.100) -0.002 (0.607) | (2.152) -0.002 |
| Intercent (includes rest of | -18 189 | _19.890 | 10 010 | 12 017 | (U.U47) 91.056 | (-0.041) | (1.027) | (0.685) | (0.846) |
| County Dublin, etc.)+ | (0.698) | -12.020 | (0 800) | -13.31/ | -21.050 | 0.077* | 0.038 | 0.020 | -0.020 |
| $\overline{\mathbf{R}}^2$ adi, for d.f. | (0.090) | (0.511) | (0.000) | (0.555) | (0.812) | (1.490) | (0.803) | (0.359) | (0.346) |
| d.f. | 37 | 37 | 37 | .025 | .030 37 | .778 38 | .820 .38 | .787 38 | .829 38 |

Note: All independent variables except number of pupils, per cent lay teaching staff, and pupils per teacher are 'dummy' variables, set equal to 1.0 when the quality indicated is present.

+Intercept includes convent school; Catholic management (other than lay, proprietary); rest of Co. Dublin; day only; not all boys (i.e., all girls or mixed); and in free scheme.

,

*Significant at 90%.

**Significant at 95%.

***Significant at 99%.

funds, which are, in effect, private Secondary School expenditures. Where boarding expense is excluded, this total is estimated at £3,254,924; where boarding expense is included, the figure rises to £6,086,472. It should be noted that these figures included religious imputations. Total estimated grants to Secondary Schools can be derived from the Table; these are estimated either at £9,517,670, the difference between the sub-totals in the first and third columns of estimates, or £9,336,716, the difference between the totals. Both are two different estimates of the same figure. Our estimates, it will be recalled, are for school year 1974/75. As noted earlier the State was in 1975 in the process of shifting from a fiscal year beginning April 1 to a calender year. The Department of Education reports grants of £7,149,587 (capitation grant of £2,193,953 and supplement grant in lieu of tuition of £4,955,634) for the nine-month period ending December 31, 1974. Adding one-third, to annualise, gives an estimated one-year equivalent of £9,532,783, which is very close indeed to our survey-based estimates.

An alternative way of estimating a 1974/75 figure for total grants is to add one-fourth of the 1975 reported figure to the nine-month 1974 figure. The 1975 reported figure is £11,128,825; one-fourth of this amount is £2,782,206, which, when added to the nine-month 1974 figure of £7,149,587, yields a total for 1974/75 of £9,931,793. Whichever figure is used, it is apparent that our survey-based estimate is surprisingly close to the official reported figure.

Our best estimate of the private expenditure on Secondary School education is the sub-total figure for net expenditure, own funds, exclusive of boarding expense, of $\pounds 3,254,924$, reported in Table 5.17. Others, we know, will want to exclude religious imputations, to obtain an estimate of *cash* private expenditures. Such an estimate is found in the last column of Table 5.17. Total net expenditure, less religious imputations, from own funds is estimated of $\pounds 1,122,902$. Note should be taken of how this total is arrived upon. Positive estimates for Protestant Schools, fee-charging schools, and schools in the high-income areas of Dublin, are substantially offset by negative figures for the remaining schools. Indeed, the sub-total is almost exactly the same as the figure for Protestant Schools alone.

The Department of Education reports payment of £15,819,161 in incremental salaries to Secondary School teachers in the nine months ending December 31, 1974, and £31,743,892 in the year ending December 31, 1975. Using the method described above, this suggests an estimated £21,092,214 for 1974/75. If this amount is added to our estimate, from Table 5.17, of £12,772,594 for total net expenditure by the schools, a total Secondary School expenditure item (exclusive of depreciation, capital costs, boarding expense, administration, examinations, and transport) of £33,864,808 is obtained. Our estimate of £3,254,924 in private expenditure is 9.6 per cent of the total. Using the alternative method described above, adding one-fourth of the 1975 figure to the nine months of 1974, we obtain

157

| Independent variable | Net expendi- tures | Net expend- itures less religious imputations | Own funds net expendi- tures | Own funds net expendi- tures less religious imputations | Per pupil net expendi- tures | Per pupil net expend- itures less religious imputations | Per pupil own funds net expend- iture | Per pupil own funds net expend- itures less religious imputations |
|---|---|--|---|---|---|---|--|--|
| Protestant management Louth/Longford/Cavan/Meath/Monaghan | | | | | 0.164*** (7.875) 0.010 (1.019) | 0.199*** (9.706) 0.013 (1.257) | 0.174*** (6.513) 0.014 (0.910) | 0.206*** (7.884) -0.016 (1.185) |
| Kildare/Wicklow Dublin: High income areas Number of pupils Any boarders All boarders | 7.590* (1.292) 31.872*** (5.050) 0.091*** (8.962) 14.57*** (4.937) | 7.482 (1.243) 33.804*** (5.225) 0.079*** (7.742) 13.474*** (4.419) | $\begin{array}{c} 7.563 \\ (1.284) \\ 27.656^{***} \\ (4.553) \\ 0.034^{***} \\ (3.445) \\ 12.518^{***} \\ (4.154) \end{array}$ | 8.209 (1.343) 30.638*** (4.861) 0.025 (2.421) (1.535 (3.689) | 0.036** (2.066) 0.033*** (5.900) 0.262*** | 0.034** (2.919) 0.028*** (5.051) 0.208*** (7.867) | 0.035** (2.284) 0.037*** (5.173) 0.240*** (6.516) | 0.027** (1.791) 0.032*** (4.331) 0.328*** (8.378) |
| Not in free scheme (fee-charging) Intercept (includes rest of County Dublin, etc.)+ R ² adj. for d.f. d.f. | 52.636*** (9.284) -12.864*** (3.092) .637 137 | 51.829*** (9.068) -12.177*** (2.977) .601 137 | 62.507*** (10.866) -10.895 (3.054) .620 137 | 60.277*** (10.098) -11.531 (1.327) .586 137 | (3.122) 0.056*** (3.050) 0.063*** (18.521) .840 135 | (1.307) 0.034** (1.879) 0.053*** (15.848) .843, 135 | 0.077*** (3.255) 0.006* (1.382) .789 135 | 0.056*** (2.396) -0.005* (1.143) .774 135 |

Table 5.16: Regression results: Secondary School expenditures, various concepts, regressed on selected independent variables, 1974/75, all Secondary Schools (t-statistic in parentheses under estimated coefficient)

All independent variables except number of pupils, per cent lay teaching staff and pupils per teacher are "dummy" variables, set equal to 1.0 Note: when the quality indicated is present.

+Intercept includes convent school; Catholic management (other than lay, proprietary); rest of County Dublin; day only; not all boys (i.e., all girls or mixed); and in free scheme.

*Significant at 90%.

Significant at 95%. *Significant at 99%.

ECONOMIC AND SOCIAL RESEARCH INSTITUTE

58

THE

| Table 5.17: | Total Secondary | School | expenditures, | various | concepts, | 1974/75, | estimated | from | estimating | equations | in Tab | le 5.16, | together | with num | bers of |
|-------------|-----------------|--------|---------------|---------|-----------|-----------|-------------|--------|------------|-----------|--------|----------|----------|----------|---------|
| | | | | | students, | total and | selected ca | tegori | ies | | | | | | |

| Pupils, category | No. of Pupils ^a | Times net expendi- tures per pupil | Times net expendi- ture less religious imputations per pupil | Times net expendi- tures, own funds, per pupil | Times net expendi- ture less réligious imputations, own funds, - per pupil |
|---|----------------------------|---------------------------------------|---|--|--|
| All schools | 173,188 | £10,910,844 | £9,178,964 | £1,039,128 | £865,940 |
| Protestant schools | 5,647 | 926,108 | 1,123,753 | 982,578 | 1,163,282 |
| Schools not in "free scheme" | 16,844 | 943,264 | 572,696 | 1,296,988 | 943,264 |
| Schools in Counties Louth, Longford, Cavan, Meath | | | | | |
| and Monaghan | 13,175 | -131,750 | -171,275 | -184,450 | -210,800 |
| Schools in high-income areas, County Dublin | 3,448 | 124,128 | 117,232 | 120,680 | 93,096 |
| Expenditures, Sub-totals | | £12,772,594 | £10,821,370 | £3,254,924 | £1,122,902 |
| pupils | 61,084 | 2,015,772 | 1,710,352 | 2,260,108 | 1,954,688 |
| Boarding expense, schools with only boarding pupils | 2,381 | 623,822 | 495,248 | 571,440 | 566,678 |
| Expenditure totals | | £15,412,188 | £13,026,970 | £6,086,472 | £3,644,268 |

(a) Source: Department of Education List of Recognised Secondary Schools, 1974/75.

Note: The first line is obtained by multiplying the total number of Secondary School pupils by the constant terms in the appropriate equations in Table 5.16. The expenditure estimates on Line 1 are hence those which would obtain if *all* pupils were Catholic day pupils, in "free scheme" schools outside of Counties Louth, Longford, Cavan, Meath and Monaghan, and the high-income areas of County Dublin. The other lines indicate the *incremental* amounts associated with the types of schools indicated.

an estimate of $\pounds 23,772,594$ for incremental salaries, and total expenditures of $\pounds 36,527,728$, of which private expenditures are 9.0 per cent. The correct figure being somewhere between these two, we estimate private expenditure as 9.3 per cent of total Secondary School expenditure.

This assumption is employed in Table 5.18. Public Secondary School expenditures are entered, using Department of Education data. Private expenditure is estimated at 9.3 per cent of total expenditure, or 10.25 per cent of public expenditure. Private expenditure is allocated among cash and religious imputations in the same proportions as are reflected in Table 5.17.

Table 5.18: Public, private, and total Secondary School current expenditure, nine monthsending December 31, 1974, and twelve months ending December 31, 1975

| | 1974 | 1975 |
|---|-------------|-------------|
| Public expenditure (schools), total | £24,048,029 | £44,770,237 |
| Incremental salary paid to Secondary teachers | £15,819,161 | £31,743,892 |
| Capitation grant | 2,193,953 | 4,551,994 |
| Supplemental grants to Secondary Schools in lieu of | | |
| tuition fees | 4,955,634 | 6,576,831 |
| Miscellaneous other ^a school expenditures | 1,079,281 | 1,897,520 |
| Private expenditure, total | £4,056,215 | £7,551,460 |
| Own funds net expenditure, exclusive of religious | | |
| imputations | £1,615,088 | £3,006,812 |
| Religious imputations | 850,695 | 1,583,741 |
| Sub-total, private expenditure | 2,465,784 | 4,590,553 |
| Sub-total, public (schools) and private expenditure | £26,513,813 | £49,360,790 |
| Boarding expense | 1,590,431 | 2,960,907 |
| Total, public (school) and private expenditure | £28,104,243 | £52,321,697 |
| Public expenditure (Department of Education) ^b | - - - | |
| Examinations | £614.637 | £759,221 |
| Costs of administration, inspection, etc. | 510.017 | 877.253 |
| Transport | 2.183.420 | 3.791.041 |
| Other | 453,807 | 526,578 |
| Total secondary school expenditure | £31,412,317 | £58,275,790 |

- (a) Includes Science and Equipment Grants; Grants for Irish and Bilingual Schools; Bonus for Choirs and Orchestras; Educational Television Services; and Annual Repayment of Building Loans. Some of these may include small amounts for Comprehensive and Community Schools.
- (b) The Department of Education reports total amounts for these items for "Secondary Education," i.e., Secondary Schools, Comprehensive Schools, and Community Schools. The reported amounts have been adjusted slightly, according to the proportion of Secondary School pupils in "Secondary Education".

Source: Public expenditure: Department of Education. Private expenditure, author's estimates-see text.

Similarly, boarding expense is entered, using the same proportions as are found in Table 5.17. Finally, non-school expenditure (i.e., Departmental expenditure, such as examinations, administration, transport) is added, to obtain total Secondary School expenditure.

Conclusions

Total and per pupil expenditures are among the most basic kinds of educational data. Without them, it is difficult if not impossible to make intelligent and rational judgements regarding educational policy, particularly where resources are involved. Without them, it is impossible to know whether society is getting its money's-worth from education. It is impossible to know whether resources are being allocated fairly and efficiently.

In this chapter, we have reported an estimated $\pounds7,029,658$ in private expenditures in first and second level schooling in Ireland in 1975, among Private Primary Schools ($\pounds2,044,715$), National Schools ($\pounds394,390$), and Secondary Schools ($\pounds4,590,553$). If an estimated $\pounds2,960,907$ in boarding expense is added to Secondary School private expenditures, the overall total rises to $\pounds9,990,565$, or roughly ten million pounds. The typical amount will probably be proportionately higher; the 1975 figure for National School private expenditure is liable to be lower than other years, for reasons mentioned.

These are, we admit, fairly crude estimates, using incomplete and often shaky data, and extrapolating in order to be consistent with a change in the State's accounting year. But far more precise and timely data are now available, if only the State will collect and process them. With the exception of Private Primary Schools, the schools discussed in this chapter have quite recently begun systematically to keep records which would make it possible for the State to gather expenditure data. It now seems time for the State, through the Department of Education, to gather and publish school expenditure data, and to make available (if not to publish) background data on each school which would permit analysts to make more sophisticated analyses than we have been able to perform here.

Chapter 6

Summary and Concluding Remarks

The Irish educational system, in all its interesting complexity, is a product of the unique character and history of Ireland. It is in the process of important change, in size, purpose, and nature. This study has discussed the structure of Irish education, especially as regards finance; the history of the system, especially as it has influenced structure and finance; the probable future course of the system, especially with respect to enrolments and expenditures; and the role of private funds in financing Irish education.

In this chapter, we summarise the study, and offer a few concluding remarks.

First Level

Two-thirds of all Irish students, in first, second, and third level institutions, are at the base of the pyramid, in first level or primary schools. Most of these -94 per cent of first level pupils, and 63 per cent of all pupils – are in National Schools. In recent years, much more public attention seems to have been focused on second and third levels, so the disproportionate importance of the National Schools may sometimes be forgotten.

Compulsory schooling begins at age six (though 90 per cent enrol in noncompulsory infants' classes at age four) and continues through approximately the eleventh year of age. There is no examination at the end of primary education.

National Schools are almost always owned and managed by Church bodies or religious orders. Each National School has a Patron (usually the Bishop), and the Patron in turn appoints the majority of members of the School's Committee of Management. The Chairman of the Committee of Management is almost always the Parish Priest or, in the case of the Church of Ireland, Rector. Thus National Schools are not national in administration; they are parochial or, at highest, diocesan. National Schools are distributed around the country more or less in proportion to the distribution of children of the various faiths. Consequently, in 1975/76, 92.6 per cent of the National Schools were under Catholic Patronage, and these schools enrolled 97.4 per cent of all pupils. It cannot be concluded that all of these pupils had the same religious affiliation as their school, though of course the vast majority did.

The National School system is an 'aided' one: almost all the funds come from the State. The locality (i.e., the Management Committee) must provide the site on which the school stands; the State then pays from 66 to 99 per cent of construction costs. Teacher salaries are paid in their entirety by the State (an exception to this statement is the capitation convent and monastery National Schools, which are paid a per-pupil grant in lieu of payment of teacher salaries), as are principals' allowances, etc. In practical terms, the teachers are public employees, though the official theory is that the teachers are employees of the Management Committee, who receive an implicit grant from the State equal to teachers' salaries. In addition, the State pays a grant toward school operating costs of £6 per pupil (this amount is subject to a per-school minimum, which increases somewhat the grant to the smallest of schools), on the condition that an additional £1.50 per pupil be raised locally. In future, as the £6.00 grant rises, it is assumed that the local contribution will rise, *pari passu*.

The £6.00 per pupil grant scheme is a relatively new one. Prior to its introduction in 1975, the State contributed to National School operating budgets through a more complex, and less generous, system of grants for heating, cleaning, and painting. In 1973/74, under the old scheme, we estimate total locally (i.e., privately) raised funds toward current National School expenditures to have been £1,389,135, or 3.0 per cent of total current National School expenditures. Our estimate for calender 1975 (in 1974, accounting was shifted from a year ending March 30 to a year ending December 31) is only £394,390, approximately 0.5 per cent of the total. This latter figure we believe to be temporarily depressed, because of the introduction of the new scheme. In brief, National School expenditures did not increase as quickly as their funds did. Privately raised revenues were a much larger percentage of total revenues than privately financed expenditures were of total expenditures. The schools did not, on average, spend $\pounds7.50$ per pupil. In 1975, the average school appears to have taken in income (grants plus locally raised funds) of £7.98 per pupil, but to have spent only £5.96 per pupil.

While the National School system reflects uniquely the Irish situation and experience, it was established, in its main details, in the early to midnineteenth century, when all of Ireland was still a part of the United Kingdom. In 1831, the British announced the creation, in Ireland, of a 'system of National Education', which was to be effectively non- or multidenominational, though under a 'Board of National Education' of whose members a majority were Protestants. There then ensued a struggle between the British and the Roman Catholic hierarchy in Ireland (and the Presbyterian Church as well) over the terms and structure of the system. The outcome, in the early 1860s, was a National School system very much like the present one: an aided, denominational, managerial system. The establishment of Saorstat Eireann altered the curriculum and choice of textbooks, but it had little influence on the structure of the primary system. The only important change in this regard was the abandonment of the 'results system,' by which teacher salaries were determined, in part, by their pupils' results on examinations conducted by inspectors representing the Board.

One characteristic of the National School system from the very start was

that it was operated with spartan simplicity and frugality. It was possible to provide a primary education to all children in the country in part because teachers were poorly paid, facilities were often rudimentary, and there was little expenditure on equipment. These characteristics persisted until fairly recently, and indeed, with the exception of teacher salaries, they persist today in some areas.

An examination of the history of the National School system suggests that there are, in fact, four principal reasons why Ireland, an underdeveloped, peasant economy at the outset, was able to establish such a highly developed school system, especially in the sense of participation rates, often in advance of other European countries. One was, as just noted, that the education of Irish children has been provided at a low, sometimes even primitive, level of amenity. A second was the involvement and concern of the Catholic Church, whose interest in education probably exceeds that of any other large organised religious body in the world. The Church itself has made a major contribution to the financing of education in Ireland; and its interest and efforts have inspired the contributions of others, including those of the State.

Third is the contributions made by members of religious orders, both in terms of actual cash contributions, especially to construction of schools, and and in terms of the contribution of their services. Approximately two-thirds of religious in Ireland have taken up teaching vocations. Traditionally, they return their salaries to their communities, which in effect usually means to their schools. Moreover, they have often worked more hours per day, and more days per week, than lay teachers.

Fourth, the Irish system of education has economised by traditionally emphasising the subjects in the curriculum which have low technical content and, more to the point, which have required little in the way of costly equipment. There are a number of reasons for this. Only fairly recently has preparation for employment become a priority objective of the school system. For the greater part of its history, the Irish school system has been devoted to moral, intellectual, and religious objectives. Moreover, even had there been a major training objective, the structure of the economy was such that mainly verbal skills were required. And, of course, the subjects taught were cheap to teach.

All four of these reasons for the cheapness of the system are in the process of disappearing, some of them rapidly. Standards for school construction, maintenance, heating and equipment, while lagging behind other areas of Irish life, have been pushed forward by general development, and by the insistence of primary teachers on better working conditions. The relative or proportionate role of the Church appears to be declining, and additional resources for education will come in greater degree than in the past from public sources.

The most dramatic, as well as the most easily quantified, changes concern the decline in numbers of religious, which affects, primarily, Secondary Schools, but has an impact on National (and Private Primary) Schools as well. Religious as a percentage of National School teachers appears to be stable at around 23 per cent from 1956 (and perhaps before that) through 1966. It then suddenly falls to 15 per cent by 1974, and our prediction is that it will fall to about 9 per cent by 1986. The era is virtually, if not completely, over in which the primary system could economise through use of large numbers of religious.

Finally, an increasing technical content, mainly in second-level but also in first-level schools, will certainly raise the cost of instruction a good deal in years to come. At the same time, it will almost certainly increase the optimal size (in a cost sense) of schools: the larger is the number of pupils who share the use of a piece of equipment, the lower is the per pupil cost. For this reason, we must expect tomorrow's schools to be a good deal larger, in terms of numbers of pupils than today's. Girls' and boys' schools may share equipment; perhaps economic forces may in the future reduce the extent of sex-segregation in the schools.

On top of the forces we have discussed, the coming of equal pay - an EEC requirement - will raise average wages, and hence costs, more in the schools than in any other area of the Irish economy. And the process known as 'Baumol's Disease,' after economist William J. Baumol, who first set forth its properties, by which prices increase more rapidly in the services sector of the economy (including education generally) because of the greater difficulty in that sector of getting productivity increases, virtually guarantees an increase in per pupil costs, *ceteris paribus*, of around three per cent per year.

These forces suggest explosive growth in per pupil school expenditures in the coming years. Unfortunately, this growth occurs in the midst of explosive growth in school enrolments, partly a result of a rapidly rising youth population, and partly as a result of rising school participation rates. In first level, where participation rates are already approximately 100 per cent, only population growth is an influence. In 1975 and 1976, births were fewer than had been predicted, probably because of the economic depression; but the shortfall in births was more than offset by a greater-than-expected child immigration, probably associated with the return to Ireland of past *émigrés*, now parents. In the event we expect child immigration to taper off at about the same time as births increase, so that once again the one will offset the other. In 1974, there were 559,900 children aged 12 and under in Irish schools. Our prediction is that by 1986, this number will have increased to 640,000, an increase of 80,200, or 14.3 per cent.

We expect little absolute growth in the Private Primary system so most of this growth will occur in the National Schools. In 1974, total National School enrolment was 521,100; in 1986, we expect it to total 605,200, or an increase of 84,100, or 16.1 per cent. Plausible patterns of enrolment and of class sizes suggests an increase in full-time National School teachers of 3,115 between 1974 and 1986. The same plausible patterns suggest an increase in National School public expenditure of $\pounds 28,693,000$ (in constant 1970 prices), or by 80.5 per cent, including transport and administrative overhead costs.

Discussion of first level education is not complete without a mention of Private Primary Schools. While this sector seems small by comparison with National Schools (in 1974 it enrolled only 23,260, just 4.3 per cent of first level pupils and only 2.9 per cent of all pupils), it is larger than the Universities, a somewhat striking statistic. Moreover, the vast majority of Private Primary Schools are concentrated in and around Dublin, and in that area they form a larger percentage of the system than statewide figures would suggest.

These schools receive no State funds of any kind (they are financed through fees); and as a result, the State collects very little data on them, and exercises no control over them. We estimate that in 1974/75, an average of £95.33 per pupil (£77.51 in Catholic and £168.50 in Protestant Schools) was spent in Private Primary Schools. In the same year, public and private per pupil expenditure, including transport and administrative overhead, was approximately £94.00 in the National Schools. These figures are so close to one another as to be indistinguishable.

As a rule, then, pupils in Private Primary Schools do not appear to receive a more lavish education than those in the National Schools; they only pay more for them. What, then, is the attraction of Private Primary Schools? One answer, of course, is that while they are not superior, they are at least somewhat exclusive, which many parents see as an advantage. More important is the fact that the vast majority of Private Primary Schools are operated jointly and share grounds with Secondary Schools. In many cases, parents are said to enrol children in Private Primary Schools in order to assure a place in a Secondary School when the children are older. Private Primary Schools are concentrated in the Dublin area, as noted, and secondarily in Cork City. These areas contain more wealthy persons who might be able to afford fees more easily, and certain middle-class families, especially in Dublin City, might not wish their children to mix with economically disadvantaged and educationally deprived children, who would be found in a relatively greater concentration in National Schools in those areas. Probably more important is the fact that there is more demand for Secondary School places, relative to supply, in these areas, and hence more competition for entry; so that the motivation for parents to seek secure Secondary School places by enrolling children in affiliated Private Primary Schools is presumably greater than elsewhere.

The same enrolment projections and plausible patterns mentioned earlier suggest Private Primary School enrolment rising from 23,300 in 1974 to 26,400 in 1986, an increase of 3,100, or 13.3 per cent. Assuming no change

in the pupil-teacher ratio (estimated at 20.0 to 1 in 1974), this should provide approximately 157 new teaching posts in this sector.

Second Level

Second level schools in Ireland consist, in descending order of enrolment, of Secondary Schools, Vocational Schools, and Comprehensive and Community Schools (the last two usually being considered together). Instruction is divided into junior and senior cycles. The junior cycle normally requires three years of study, and thus terminates roughly at the end of compulsory schooling, at age 15, with an Intermediate Certificate examination. Some pupils also take a Group Certificate Examination, after two years of junior cycle. The senior cycle, which consumes an additional two years, culminates in the Leaving Certificate.

In 1974, Secondary Schools enrolled 68.9 per cent of all second-level pupils. These are voluntary or private schools, emphasising academic or 'arts' education, similar in many respects to grammer schools in other countries, except that they enrol a larger percentage of second level students. Almost all Secondary Schools are denominational. In 1973/74, 433 of 534 Secondary Schools were operated by Roman Catholic religious orders; another 31 were Catholic diocesan colleges; and 33 were 'lay Catholic' schools, operated, in effect, as businesses, by individuals. In addition, there was one Jewish and 26 Protestant Schools. As 'recognised' Secondary Schools, all of those institutions receive a very considerable amount of State aid.

The largest part of State aid to Secondary Schools is direct State payment of 'incremental salaries' to teachers. The school itself also pays a 'school salary' (set in recent years at £400 per year), but the State 'incremental' component forms the bulk of teacher incomes. The State also pays a socalled 'capitation grant,' in respect of all pupils. These payments go to all recognised Secondary Schools.

In addition, in 1967/68, the State introduced the so-called 'scheme for free post-primary education,' whereby an additional, supplementary grant (originally £25 per pupil, and increased to £50 in 1975/76) would be paid to all schools willing to forgo fees. Grants are also paid on behalf of boarding pupils at 'low fee' boarding schools, defined in 1975/76 as those with boarding fees of no more than £265 per year. Catholic Secondary Schools enrolling approximately 92 per cent of all pupils in such schools joined the new scheme. For Protestant Schools, whose fees had been higher, a slightly different scheme was created, by which the State provides a block grant proportional to that given to Catholic schools to a Secondary Education Committee, which distributes scholarships to pupils in accordance with need.

In effect, the free scheme extended 'free' or publicly supported education through second level. But it can be criticised in that it used a very large amount of scarce public funds in such a way as mainly to substitute for existing private funds. Only a small fraction of the public money spent on

the free scheme actually increased resources available to education. Our estimate is that in the first eight years of the scheme, approximately £31,740,000 went as windfall gains to families enrolling children in Secondary Schools (in the sense that these families would have paid fees), rather than to education.

In calendar 1975, £58,275,790 was spent on Secondary Schools, according to our estimates and Department of Education data. Of this, £44,770,237 was public expenditure made in or by the schools, including incremental salaries, and grants of all kinds; and £5,954,093 in departmental overhead and transport expense; for a total of £50,724,330 in public expenditure. Private expenditure is estimated at £4,590,790, thus giving an overall total of £55,314,883, of which private expenditure is 8.3 per cent. This total for private expenditure is estimated by statistically excluding boarding-related expense, on the theory that this expense is not educational in nature. That judgement is arbitrary and arguable, however. Estimated boarding expense (all private) was £2,960,907, thus bringing the total to £58,275,790, and the private component to £7,551,460, or 13.0 per cent of the total. Where administrative overhead and transport costs are excluded, the private proportion rises further.

Cross-section, multiple regression analysis of Secondary School expenditures reveals some interesting regional differences in per pupil expenditures by schools (excluding incremenatl salaries, which are paid directly by the Department of Education, and excluding administrative overhead and transport costs). More is spent per pupil than elsewhere in the high income parts of Dublin City and Dun Laoghaire (including suburbs), in Kildare/ Wicklow, and probably in Laois/Offaly/Carlow. Less is spent in the low income parts of Dublin, in Louth/Longford/Cavan/Meath/Monaghan, in Leitrim/Roscommon/Mayo/Sligo/Donegal, and (with respect to Catholic Day Schools only) in Tipperary and probably Clare/Limerick. These differences in expenditures are statistically significant and persist even though other variables are controlled for. The differences are attributable in large part to the differential ability of schools in various areas to raise funds for the non-State-financed component of education expenditure. While the study does not permit conclusions about differences in quality of schooling, or in lifelong earning potential, but only about differences in per pupil expenditure, it does appear that regional (and by implication income or social class) differences account for differences in educational opportunity which seem contrary to egalitarian standards. The same might be said of religion: our study shows that significantly more is spent in respect of education in Protestant than Catholic Secondary Schools, even where other variables — including use of religious teachers — are controlled for.

Our study also shows the influence on per pupil and per school educationexpenditures of a number of other variables. Of particular interest is the fact that the per cent of the lay teaching staff is strongly and positively related to both per pupil and per school expenditures, even where a charge is included for the services of religious. We interpret this as a clear indication of the economic contribution of religious to education in Ireland. In addition, pupils per teacher, an indicator of class size, is positively related to both per school and per pupil expenditures, as regards Catholic day schools.

We also find that schools not in the free scheme spend considerably more than schools in the free scheme, on both per school and per pupil bases, even where other influences are controlled for. Again, this finding seems to reflect against equality of educational opportunity within the Irish Secondary School system. Of course, such inequalities are unlikely ever to be fully eradicated, so long as it is possible for élite, private schools to operate. But it should be pointed out in the Irish case that these schools receive a considerable amount of State aid, including the payment of incremental salaries.

Secondary Schools are also eligible for State capital aid, on terms similar to those affecting National Schools. The religious order or other party operating the school must provide the site, and pay for 20 per cent of building and equipment costs; the remaining 80 per cent is paid by the State. Boarding facilities are not covered by State grants. Fee charging Secondary Schools are also eligible, but are not approved for capital grants as frequently as are free scheme schools.

The second largest component of the second level system in Ireland is the Vocational Schools, which in 1974 enrolled 24.4 per cent of all second level pupils. These are wholly public institutions, owned, operated, and maintained by Vocational Education Committees (VECs), the only form of local education authority in the Republic. The VECs are also responsible for Regional Technical Colleges and for the Colleges of Technology in Dublin. In 1973/74, over 85 per cent of current expenditures, and all of capital expenditures, were from State funds, with most of the remainder financed through the rates.

The Vocational Schools have recently been through a major change. Effective in the 1967/68 school year, with the introduction of the 'free scheme,' tuition fees for second-level students were abolished, and State grants to the VECs correspondingly increased. At the same time the curriculum was radically changed. Vocational Schools had theretofore provided 'continuation' and 'technical' education. 'Continuation' refers to essentially continuing and expanding on the subjects offered in National Schools; it culminated, after two years, in the 'group certificate' examination. 'Technical' education referred to specialised, advanced, work-oriented courses, including apprenticeships. Thus there was little if any overlap between subjects studied in Vocational and in Secondary Schools. At the time of the introduction of the 'free scheme,' Intermediate and Leaving Certificate courses, which theretofore had been the exclusive domain of the Secondary Schools, were introduced in Vocational Schools. Hence a very substantial overlap came to exist between the curricula of the two types of schools. Some have characterised the resulting Vocational Schools as, in effect, 'second-rate Secondary Schools'. Whether the characterisation is proper or not, it is clear that in the future, the Vocational Schools will grow relative to the Secondary Schools, by absorbing some of the demand — mainly from the lower part of the income distribution — for second level places which otherwise would have been met by Secondary Schools.

The fastest growing portion of the second level has been, and in the foreseeable future will be, Community Schools. Comprehensive and Community Schools are State schools which offer both arts and vocational courses, and have features of both Secondary and Vocational Schools. The first Comprehensive School was opened in 1966; the last (no more evidently will be opened) was opened in 1974. In 1973, the first Community School was opened. Irish Comprehensive Schools are State schools, very much on the pattern of British Comprehensives, except that they are specifically denominational, either Catholic or Protestant. Community Schools are somewhat different. They are joint ventures of one or two religious orders (depending on whether the school is co-educational), the local VEC, and the Department of Education. Indeed, quite a number of Community Schools have been formed through merger of existing Secondary and Vocational Schools. Almost all the second level schools newly opened in the future are expected to be Community Schools, though existing Secondary and Vocational Schools may be rebuilt and expanded as needed.

The second level system, like the National School system, is to a very considerable extent a product of nineteenth century struggles between the British authorities and the Irish Catholic Hierarchy, and large parts of it were developed in substantially their present form while all of Ireland was still part of the United Kingdom. In 1835, the British announced plans for a nationwide system of what would have in effect been second level schools in Ireland, but the plan was finally dropped, largely because of the opposition of the hierarchy. The present system of Secondary Schools dates from 1878, when an 'aided system' was introduced, in which most of the schools were operated by religious orders. The level of funding was so low that the system required either an extremely economical operation, the imposition of fees, or great personal sacrifice on the part of teachers and others, or some combination of these. The Vocational system also dates to the British, with the establishment of local committees for vocational education in 1898/99. The Comprehensive/Community system, as noted, has more recent origins.

Like the National School system, the second level system, and particularly the Secondary Schools, could develop extensively in spite of economic underdevelopment and limited resources in part because of four circumstances: the spartan and frugal character of facilities and equipment; the interest of the Church; the contributions of the religious; and the low technical content of the curriculum. The first of these is somewhat less important in second than first level; the last two are decidedly more important. No statistics exist on the involvement of religious in Secondary education in the nineteenth century, but it seems likely that the vast majority of teachers were nuns and brothers, especially in the Sisters of Mercy, the Presentation Sisters, and the Christian Brothers. By 1961, religious as a per cent of all Secondary teachers stood at approximately 50 per cent. It is likely that the proportion had been roughly that for some time. In 1966, it was still 50 per cent. Then the numerator and denominator started moving rapidly, in opposite directions: the effects of declining vocations of the previous decade began significantly to affect the numbers of religious teachers, at the same time as the total number of teachers rose as a consequence of enrolment gains attributed to the introduction of the 'free scheme'. By 1974, religious teachers had fallen to 31 per cent of the total (and the inclusion of Comprehensive and Community School teachers would have reduced the figure further). Our prediction is that by 1986, the proportion will have fallen to 20 per cent. There is no question that this trend will increase school expenditures. Equally important, but outside the domain of this study, it will also change the character of Irish Secondary education.

Beginning in the late 1960s, there began to be an increase in the technical content in the Secondary School curriculum; and this trend is continuing. It has increased and will continue to increase school expenditures. Thus, for a variety of reasons, it appears that the day of low-cost second level education in Ireland is at an end.

There are still further reasons to expect costs in second-level schooling to rise in the coming decade. One, as noted in the case of National Schools, is the cost of the coming of equal pay. In addition, there is a 'bulge' in the age distribution of second level teachers, accounted for by the very large increase in numbers of these teachers hired in the early years of the 'free scheme'. As the 'bulge' moves through the increments structure, there will be disproportionate increases in salary costs. And finally, 'Baumol's Disease' will probably account for an annual growth in per pupil costs of approximately three per cent per year, on top of all other forces.

These rapid increases in per pupil costs come at a time of rapid enrolment increase, the product of rising youth population and participation rates. We predict enrolment growth of 36,800 (a 23 per cent gain) between 1974 and 1986 in the group aged 13 through 15, and 35,400 (a 50 per cent gain) in the group aged 16 through 18. Most of these gains will be in second-level institutions. A plausible allocation of these gains is as follows (with Comprehensive and Community Schools included with Secondary Schools): junior cycle, Secondary Schools, 22,400 enrolment gain, or 18.0 per cent; senior cycle, Vocational Schools, 12,800 enrolment gain, or 26.2 per cent; senior cycle, Vocational Schools, 10,600 enrolment gain, or 75.2 per cent. In second level combined, this would amount to an enrolment gain of 65,600,

. ..

or an increase during the 12-year period of 27.1 per cent. Our estimate is that such an enrolment increase, if realised, would require an additional 1,764 full-time Secondary (with Comprehensive and Community) School teachers, and an added 1,847 full-time Vocational School teachers.

The budget implications of these increases are somewhat alarming. In constant, 1970 prices, they imply an increase in public second level expenditures from £39,914,000 in 1974 to £82,956,000 in 1986, an increase of 6.2 per cent per year, and 107.8 per cent for the whole period. This breaks into gains of £25,131,000 (or 101.4 per cent) in Secondary (with Comprehensive and Community) Schools, and of £18,915,000 (or 133.9 per cent) in the Vocational Schools. If these gains are added to the somewhat more modest increases expected in National Schools (see above), a total first and second-level public expenditure increase might be expected of some £72,307,000 between 1974 and 1986, which amounts to a growth rate of 5.7 per cent per year, or 95.0 per cent over the whole period. Since conservative assumptions have been used in arriving at these figures, it might be fair to round upward, and state that real public expenditures on first and second level education are likely to double over the period 1974-1986.

It should be stated that our enrolment forecasts and the expenditure figures accompanying them are based on the assumption that places in school will actually be provided for all those who might seek them. That is, we have tried to deal with the demand for places — in effect, with the problem, not its solution — in our forecasts.

Third Level

Third level is not really a subject of this study, but a few details can be provided to complete the picture. The structure of third level education has been in flux in recent years, and it may be a few more before the dust settles.

The principal institutions are the Universities, which in 1974 enrolled 68.7 per cent of all third level students (though only 2.5 per cent of all students, levels one, two, and three combined). The State provides funds to a Higher Education Authority, which in turn allocates funds to the Universities. The Universities also receive a substantial amount of funds from the Department of Agriculture. Approximately 85 per cent of their income for current expenditures is derived from these two kinds of State grants. Most of the remainder is received as fees. However, a fraction of fee income, too, arises indirectly from the State. Roughly one-quarter of University students are recipients of Higher Education Grants, which also come in part from local authorities.

The other third-level institutions are the Regional Technical Colleges (a rapidly growing part of the system, initiated in the 1960s with World Bank assistance) and the Dublin technological colleges, under the VECs (though with very considerable State funding); the Colleges of Education, organised along denominational lines, with State funds, to provide teacher training; and a small miscellany of other aided and non-aided institutions.

Since third level has not been an important part of this study, enrolment forecasts are somewhat more speculative here than elsewhere. None the less, we anticipate very substantial growth at third level, if places are provided. The expected increase in enrolment of those aged 19 and over, between 1974 and 1986, is 36,300, a gain of 159 per cent. Not all of this is third level, however. We expect third level enrolments to rise by 36,200, or 121.9 per cent. If this increase should occur, expenditures would very much more than double over the period.

Policy Recommendations

If the foregoing analysis is in substance correct, there will be very great enrolment and budgetary pressures in the coming years. There is a danger that little will be said or done about these problems until they have reached crisis proportions, and that the public will learn about the pressures not from Ministerial statements but by overcrowded classrooms, widespread use of temporary classroom buildings, and lack of sufficient places for all those who want to go to school.

If that should occur, the burden of the problem will undoubtedly be borne in a capricious and, on balance, a regressive manner. It will be capricious in that the availability of school places and the quality of schooling and equipment will depend on the happenstance of residence, and other factors over which individuals have no control. It will be regressive in that, as in most unanticipated crises, it is the last ones having access to the system who are cut out. In any unplanned response to the crisis, it will be the poor and disadvantaged, who do not now attend second level schooling past compulsory years, rather than the wealthy and the advantaged, whose opportunities will be reduced.

It is for this reason that we urge a major national debate on educational priorities. Unless there are significant reductions elsewhere in the economy (and such do not seem likely), there will not be adequate public resources available in education to permit continuation of the trends of past years. Decisions will have to be taken, either openly and explicitly, or implicitly and by default. We make our own recommendations as to priorities below; but these recommendations, appropriate as they seem to us, are pressed less forcefully than the single recommendation of a national debate.

What is advocated here is an approach, not a programme. The approach may be illustrated or exemplified by a particular set of proposals, but they should be interpreted as only one possible application of an idea. The theoretical underpinnings of the approach consist of three propositions. They provide the main basis for five principles of funding of education in Ireland. The five principles, in turn, are drawn on in a suggested scheme of finance.

The first and most important of the three theoretical propositions con-

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

cerns the fundamental distinction between public and private benefits from education. Education, like many other goods and services, is a 'quasi-public good,' with both public and private characteristics. It is a private good in that its principal beneficiaries are the pupils, who receive benefits in both current or consumption form (the enjoyment of learning, of socialising with peers, of recreation, etc.) and in investment form (yielding lifelong enhanced potential earnings, as well as enhanced enjoyment of literature, music, etc.). Even were there no social provision for the financing of education, many individuals would pay for it; the implication is that it has private benefits worth paying for. But education of an individual often also benefits society at large. The social system, the political system, and the economic system all demand of their participants a certain minimal level of education. The fact that there is a large social interest in education is indicated not only by public schemes for its provision, but also by the almost universal tendency of societies to make a certain amount of education compulsory.

The second proposition can be stated more briefly. It distinguishes, again among the benefits from education, between those which occur with Ireland and those which occur elsewhere. This distinction must be made, because many Irish school leavers and graduates emigrate. Even where a school leaver emigrates and never returns, some of the educational benefits may occur within Ireland. This is so, for example, to the extent that educational benefits are current or consumption benefits, which take place at about the same time as the education takes place. Only the yield from the investment element in education is likely to be lost on account of emigration.

Looked at together, these first two propositions might serve to delimit the range of public concerns in Ireland with education, to, first, the public element, and, second, to the Irish element.

The third proposition, which may modify the effects of the first two, is that income and social class should have no effect on any person's access to schooling.

When these propositions are employed in developing approaches to the financing of schooling in Ireland, they yield the following five principles.

(1) Private resources should be used for the private-good component in education, and public resources should be reserved for the public-good component. A rule of thumb might be that the years of compulsory attendance (ages 6 through 15) be considered as conterminous with the years of education that is essentially public or social in character. This rule seems intuitively to be consistent with observable public-private distinctions. Beyond the age of compulsory schooling, i.e., in senior cycle and third level, employment and earning opportunities vary with schooling. Society makes the investment, in most cases; but the individual appropriates the returns.

(2) Where virtually every member of society takes education at a given level, and where the benefits from that education are mainly current rather than in the form of an investment, the distinction between public and private is probably superfluous. Public support does not appear inappropriate, even where some of the benefits are private, because it does not result in redistribution from the many to the few. (The same cannot be said, by contrast, of education whose benefits are mainly private, and which goes to but a minor fraction of the relevant population.) This statement applies only to education whose benefits are largely current (as in the infants' divisions of primary schools) because of the possibility of emigration.

(3) To the extent that public resources are used in the financing of education, they should be used, in so far as is practical, only to finance aspects of education which yield benefits within Ireland. This principle applies only to the later flow of benefits from the investment component in education, as (by definition) current or consumption-type benefits from an education in Ireland occur in Ireland.

(4) A more moderate alternative to the principle contained in item number one, above, is to continue the present practice of public finance for aspects of education which yield private benefits, but to treat such expenditures as personal income of the family of the pupil, and to tax it accordingly. If this approach is to assist in solving the coming fiscal problems in education, the proceeds of such a tax treatment must be applied to education.

(5) Loan or grant funds should be made available to students with more modest means, as is required, in order that lack of funds not deter anyone from obtaining an education that he or she would otherwise take. While the first three approaches, above, are based on considerations of allocational efficiency, and perhaps on the principle of reciprocal equity, this fifth one is based on that of distributional equity and equality of opportunity.

In sum, then, scarce resources available for education should be reserved, in general, for those aspects of schooling which benefit society as opposed to the individual, and in particular Irish society; and for the less advantaged. A number of actual financing schemes could be designed which would reflect such a philosophy, though the present scheme is not one of them. The following scheme is offered as an illustration, not as a definitive policy proposal. In our discussion, we follow the child through the years of schooling, beginning with the infants' division of primary school, and concluding with third level.

First, then, while attendance in the infants' divisions is not compulsory (and hence according to the first principle would be viewed as a private benefit), participation is very high (above 90 per cent), and hence continued State support appears appropriate, on the basis of the second principle. Second, no change is indicated in the financing of primary education, as it is compulsory and hence regarded as a public benefit. Third, the same can be said of junior cycle, second level, as attendance is compulsory through age 15.

Fourth, senior cycle is regarded as a private, not a public, good. Hence there is a case for a substantial private contribution. Two schemes might be considered. In the first, State support of senior cycle is withdrawn. This

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

would mean termination of the "free scheme"; abolition of capitation grants, building grants, and State payment of incremental salaries, for Secondary Schools; and a system of fees, not only in Secondary Schools, but in Vocational and Comprehensive/Community Schools. Fees would be set high enough to recover the full costs of second-level education. The resulting system, however, might deter a number of lower income young people from going on past the Intermediate Certificate. Hence, in line with the fifth principle, a system of State scholarships, based on income, would be introduced, possibly carrying a small stipend for living expenses (i.e., to cover income losses) for the students with the very lowest of incomes. Were the State funds now spent on second-level education reorganised along the lines suggested, school incomes would be higher (more private funds would be used, together with an unchanged amount of State funds), and more pupils would receive an education, than under the present scheme.

A more modest version of the same scheme would be repeal of the "free scheme," but continuation of the other forms of State aid to second-level institutions, thus requiring lower (but hardly nominal) fees, not only in Secondary but in Vocational and Comprehensive/Community Schools as well, in the senior cycle. This alternative would still require a scholarship programme, though a smaller one.

If it is felt that the "free Scheme" represents a commitment which cannot be withdrawn, it may be necessary to adopt the more moderate alternative, as is contained in the fourth of our five principles, that of levying income tax on the value of State supported senior cycle education. If it is true, as we have argued, that senior cycle schooling is almost wholly a private good, whose benefits are captured or appropriated by the pupils, then any form of State finance represents a form of State transfer payment to the families of pupils, and should be taxed as income. The amount added to reported family income would be the grant given in lieu of (or, in Protestant schools, in aid of) fees, under the "free scheme"; the capitation grant; and a *pro rata* share of State payment of incremental salaries. Tax-free allowances in the personal income tax would relieve the lowest income families from any payment, and hence their children would attend senior cycle without charge; for the rest of the population, the progressive rate structure would make the financing of senior cycle slightly redistributive, downward rather than upward as at present.

Fifth, our illustrative scheme calls for substitution of loans to students for grants to institutions and grants to students in third level. Fees would be raised a very great deal—high enough to cover the current costs of third level institutions. While this is a drastic proposal, it is hard to justify the very high current level of State support for third level education. Only a small fraction of the population has the opportunity to attend third level institutions. Those who do so are evidently rewarded handsomely, in terms of employment opportunities and higher earnings, as well as non-pecuniary amenities. There is a very strong case that learners should pay the full cost of third level education. Those whose lifelong education-associated income increments will not be sufficient to pay for third level education should not go on to third level (unless they wish to buy it as a consumption good). Those whose subsequent incomes will increase sufficiently to justify the education should use some part of their income gains to pay for the education.

In practice, there are two things wrong with the proposition contained in the last two sentences. One is that the costs of third-level education are bunched in a short period of time, when the learner is fairly young, whereas the income gains are spread over a lifetime. How can a learner pay, for today's education out of subsequent lifelong earnings gains? The other is that no one can be certain that his or her education will actually pay for itself, in the sense suggested, i.e., that the present or discounted value of lifetime earnings gains will equal or exceed the cost of education. These two problems suggest that public support of third level education takes place primarily because of inefficient capital markets, properly so called, rather than because third level education is in any common sense of the term a public good. What is called for is a very liberal loan programme, available to all who qualify for third level education, providing funds not only for fees and education-related expenses but for living expenses as well. Repayment would come from-and only from-deemed education-associated earnings gains. For the student who, e.g., takes a course in dentistry and then emigrates to practice elsewhere, a loan scheme provides a way for the Irish exchequer to recover some of the costs of providing that education.

These recommendations, especially as regards third level, may seem drastic to some. But it appears that some drastic measures *will* be taken, in one way or another. Those who find one or another feature of this programme objectionable are urged to suggest schemes of their own.

Concluding Remarks

Much of the information estimated and published in this study is more or less routinely gathered and published by the public authorities in many other countries. So should it be in Ireland.

It would be useful and desirable, and entirely appropriate, for the Department of Education to publish, and periodically to revise, a document explaining and describing the organisational and fiscal structure of the system not for the benefit necessarily of economists or other observers and analysts, but for the benefit of principals, teachers, parents, and even civil servants, all of whom seem ill-informed, as a rule, about the system.

In most countries, the education authorities publish, and periodically update, enrolment forecasts, for the information of all who are concerned with education. In Ireland, where the school system is so decentralised and where expansion and related decisions are taken by so many diverse individuals and groups, it is more than ordinarily important that these projections be published.

Finally, how can rational decisions about the amounts and allocation of public support of education be made where no information is collected, never mind published, on total education expenditures, i.e., public and private expenditures combined, and on its distribution? It is now true, as it was not as few as two years ago, that all of the major first and second level educational institutions (and perhaps all of them) keep records appropriate to the State collection and publication of complete statistics on education expenditures. As our final recommendation, we strongly urge that the State do so.

We have been told, again-and-again, by people within the education system, and even by people within the civil service, that the State will resist publication of the kinds of information called for here, not because of any administrative costs or difficulties, but because they fear that public knowledge will limit their own freedom of action. For example, we are told that the State does not want to publish information on private expenditures on education, because they fear that such publication would lead to pressure for increased public support. The State does not want to publish information on regional or other disparities in per pupil education expenditures, because they fear, or so we are told, that such publication would lead to pressures for equalisation, which would mean "levelling up" rather than "levelling down," and hence greater expenditure. The State does not want to publish enrolment projections, we are told, because resulting public debate over responses to rising enrolments might limit politicians' or civil servants' discretion to make policy decisions themselves.

We hope that these assertions are false. Public knowledge, debate, and even pressure may be inconvenient to politicians and civil servants; but they are fundamental to a democratic system. Unilateral decisions, suddenly announced, have often been taken in the past, and have occasionally led to regrettable errors. Future decisions are too crucial to permit continuation of this method. It is time to include the public as a full partner in education decisions.

Appendix 4.A

Adjustments to Published Population Estimates and Predictions

In developing the population estimates and predictions in the text, three kinds of adjustments were made to published population estimates and predictions. Estimates of the numbers of children in the State in recent years were increased, to take account of the fact that the Department of Education reported more children in school in given ages than the CSO had estimated were in the State, as discussed in the text. The CSO's predictions of numbers of births in the years 1975-79 were reduced, to take account of the effects of the current depression. And a particular pattern of child net migration is assumed. These adjustments are discussed in detail below.

Underestimate of Numbers of Schoolchildren:

The adjustments made to the CSO population estimates for 1974 and 1975 are as follows: 1974, in the 5-9 age group, the CSO estimates 318,700 children, while our corrected figure is 323,100, a difference of 4,400; and in 1974, in the 10-14 age group the CSO estimates 309,700, and our corrected figure is 313,200, a difference of 3,500; for 1975, the 5-9 figures are CSO, 319,300, corrected figure 324,200, difference 4,900; and 10-14, CSO, 313,500, corrected figure 320,000, difference 6,500. These figures are allocated by year of age by interpolation, as described later. We have accepted as correct the CSO estimates of 1974 and 1975 population in other relevant groups, i.e., 0-4, 15-19, and 20-24, lacking any contrary evidence. While there is some school attendance in these ages, participation rates do not tend toward 100 per cent and hence it is impossible to infer population from school enrolment.

Decline in Number of Births

In the present paper, the estimated number of births is taken from CSO-5000. The CSO estimates of numbers of births are, however, for five-year periods, and we have had to interpolate, assuming that during the relevant period births rise at an increasing rate. No adjustments are made prior to 1975 or subsequent to 1979. The figures below show, for 1975 through 1979, (1) the year; (2) the age reached in 1986 by the cohort born in the year; (3) the actual number of births, for 1975, the preliminary estimate of the number born in 1976, and the revised prediction of numbers born in 1977 through 1979; (4) the CSO predicted number of births, taken from CSO-5000; and (5) the difference between the last two, which then is also the amount by which we have adjusted downward the predicted 1986 population of each age group shown. Population is in thousands.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

| (1) Year born | (2) Age in 1986 | (3) Actual/estimated number born | (4) CSO births prediction | (5) Difference: (3) – (4) |
|---------------------|-----------------------|--|---------------------------------|---------------------------------|
| 1975 | . 11 | 67.5 | 69.9 | -2.4 |
| 1976 | 10 | 68.0 | 71.0 | -3.0 |
| 1977 | 9 | 69.0 | 72.8 | |
| 1978 | 8 | 72.1 | 74.1 | -2.0 |
| 1979 | 7 | 75.2 | 75.7 | -0.5 |

Net Migration Assumptions

The exact assumptions are as follows. Numbers born each year as indicated in CSO-5000, except adjusted as indicated above. No net change in number is predicted in the first two years of life. Evidently, infant mortality has been roughly offset by infant immigration in recent years. The assumed pattern of child migration between 1975 and 1981 is as is shown in the table below. For example, between ages 2 and 3, net immigration of 200 is assumed between 1974 and 1975, and the same amount in the next two years, followed by 100, 100, nil and nil. Between 1981 and 1986 (not shown in the table), no further migration is assumed, except as follows: Over the whole five-year period, it is assumed that net immigration of 10-year olds (becoming 15 year olds) will be -200; for 11 (becoming 16) year olds, -300; 12 (to 17) year olds. -400; for 13 (to 18) year olds, -500; and for 14 (to 19) year olds, -500. Considerable emigration is assumed throughout between the 19th and the 24th year of age, as can be inferred from the various tables in the text. Note that the assumptions have been stated in terms of net immigration a negative number implies net emigration.

| Between ages: | 1975 | 1976 | 1977 | 1978 | 1979 | 1980 | 1981 |
|------------------|------|------|-------|--------|------|------|---------------|
| 2 and 3 | 200 | 200 | 200 | 100 | 100 | 0 | 0 |
| 3'and 4 | 200 | 200 | , 200 | 100 | 100 | 0 | 0 |
| 4 and 5 | 200 | 200 | 100 | 100 | 100 | 0 | 0 |
| 5 and 6 | 300 | 300 | 200 | 100 | 100 | 100 | 100 |
| 6 and 7 | 400 | 400 | 300 | 200 | 100 | 100 | 0 |
| 7 and 8 | 500 | 500 | 500 | 400 | 300 | 200 | 100 |
| 8 and 9 | 500 | 500 | 500 | 400 | 300 | 200 | 100 |
| 9 and 10 | 400 | 400 | 400 | 300 | 200 | 200 | 100 |
| 10 and 11 | 300 | 300 | 300 | 200 | 200 | 100 | 0 |
| 11 and 12 | 200 | 200 | 200 | 100 | 100 | 100 | 0 |
| 12 and 13 | -300 | -300 | 300 | -200 | -200 | -100 | 0 |
| 13 and 14 | | -300 | -300 | -200 | | -100 | 0 |
| 14 and 15 | -300 | -300 | -300 | -200 | -100 | -100 | 0 |
| 15 and 16 | -500 | -500 | -500 | -400 · | 300 | -200 | -100 |
| 16 and 17 | -400 | -400 | -400 | 300 | -200 | -100 | -100 |
| 17 and 18 | -300 | -300 | -200 | -100 | 0 | 0 | 0 |
| 18 and 19 | -600 | -600 | -500 | -400 | -300 | -200 | —100 · |

Appendix 4.B

Alternative Enrolment Forecasts for 1986 from New Population Projections

The following enrolment forecasts are made by applying the predicted participation rates shown for 1986 in text Table 4.6 to the population predictions made by W. Keating (W. Keating, "An Analysis of Recent Demographic Trends with Population Projections for the Years 1981 and 1986", Irish Statistical and Social Inquiry Society, March 3, 1977), and reported above in text Table 4.2. Data are in thousands of persons.

| Years of Age | Keating Low | Keating High |
|-----------------|----------------|-----------------|
| 4-5 | 128.0 | 130.0 |
| 6-12 | 485.0 | 488.0 |
| 13 | 70.2 | 71.3 |
| 14 | 68.2 | 69.2 |
| 15 | 63.3 | 64.3 |
| . 16 | 49.5 | 50.2 |
| 17 | 35.5 | 36.0 |
| 18 | 22.4 | 22.7 |
| 19 | 14.4 | 14.6 |
| 20+ | 43.4 | 44.8 |
| Total | 979.9 | 991.1 |

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

Appendix 4.C

Teacher Requirements

Teacher requirements are calculated as follows:

National Schools:

Over the five years 1969-74, the incremental pupil-teacher ratio for the system of National Schools as a whole (i.e., the ratio of the change in numbers of pupils to the change in numbers of full-time teachers) has ranged from 5.9:1 to 27:1, and has averaged 18.17:1. In order that our estimates be conservative, we have used not the average but the highest value, i.e., 27:1. Taking the increment in National School students as 84,100, this gives an increment in teachers of 3,115. The average number of pupils per full-time teacher of 30.7 is arrived at not by assumption but by dividing the predicted number of pupils in by the resulting number of teachers.

Second-Level Schools:

The relationship between numbers of teachers and junior and senior cycle students was estimated by least squares regression, for 1963-74, separately for Secondary/Comprehensive/Community and for Vocational teachers. The results were as follows:

Full-time Secondary/Comprehensive/Community teachers=

-788.96+.02123 (Jun. Cycle Enrolment)+.14894 (Sen. Cycle Enrolment) $(t = .62159) (t = 2.46888) (<math>\overline{R}^2 = .961$)

Full-time + Part-time Secondary/Comprehensive/Community teachers=

 $\begin{array}{c} -453.29 \pm .04206 \text{ (Jun. Cycle Enrolment)} \pm .144911 \text{ (Sen. Cycle Enrolment)} \\ (t=1.17906) & (t=2.3003) & (\overline{R}^2=.969) \end{array}$

Full-time Vocational Teachers =

-442.805+.07943 (Jun. Cycle Enrolment)+.11845 (Sen. Cycle Enrolment)
(
$$t = 5.117$$
) ($t = 3.919$) ($\overline{R}^2 = .985$)

Full-time + Part-time Vocational teachers=

186.469+.12905 (Jun. Cycle Enrolment)+.12329 (Sen. Cycle Enrolment) (t = 4.7470) (t = 2.3292) ($\overline{R}^2 = .976$)

The coefficients represent, in effect, incremental teacher-pupil ratios; to

find incremental pupil-teacher ratios it is necessary to take the reciprocals. For Secondary/Comprehensive/Community, the implied incremental pupil teacher ratios are 47.1 and 6.7, for junior and senior cycles respectively, for full-time teachers; and 23.8 and 6.9 respectively for full-time and part-time combined. The second equation was used in forecasting, because the coefficient estimate for junior cycle was intuitively more plausible, because the t value of this coefficient was somewhat healthier, and marginally, because the equation as a whole had a slightly better \overline{R}^2 adjusted for degrees of freedom. It was assumed that the incremental ratio for junior cycle would remain at or near its present value of 23.8, but that for senior cycle would rise to 15.0, which is approximately midway between the estimated value of 6.9 and the junior cycle value. With predicted junior cycle enrolment of 22,400 and senior cycle enrolment of 19,800, this yields an estimated increase in number of teachers of 2,261. If 78 per cent of these (as in 1974) are assumed to be full-time teachers, the predicted value of added Secondary/ Comprehensive/Community teachers is taken to be 1,764. For Vocational Teachers, the implied incremental pupil-teacher ratios respectively for junior and senior cycles are, in the first (full-time) equation, 12.6 and 8.4, and in the second (all teachers combined) 7.7 and 8.1. The first equation gives intuitively more sensible results; has higher t values; and has a marginally higher $\overline{R^2}$. If it is assumed that the junior cycle ratio will remain at 12.6, and the senior cycle ratio will rise to the same value, 12.6, then the predicted increase in enrolment of 23,400 implies a need for 1.857 additional full-time vocational teachers. In the second equation, evidently the constant term has absorbed much of the teaching requirements of junior cycle. If one uses the implied ratios in this equation one finds a need for 2,971 added teachers. If of these, 62 per cent (as in 1974) are full-time teachers, we obtain a need for 1.847 additional full-time teachers, and 1,129 part-time teachers. This estimate of 1,847 is very close to the 1,857 obtained on the basis of the first equation. We have adopted the smaller value, even though it is based on what is evidently an inferior equation, in order to be on the conservative side in our estimates. As in the National Schools, our resulting average pupil-teacher ratios for the systems in aggregate are results, not assumptions, of our calculations, (though they are, of course, the product of other assumptions). Average class size in Secondary/Comprehensive/Community rises by one pupil; in Vocational Schools it falls fractionally. These forecasts are based on a continuation of existing policies and trends, and would be belied, e.g., by a decision to increase average class size in Vocational Schools.

Appendix 4.D

Assumptions and Methods used in Illustrative Expenditure Forecast

The detailed assumptions and methods are as follows. National School total salaries, divided by the number of full-time National School teachers, and deflated to 1970 prices, were £1,500 in 1974. Annual growth of 2.75 per cent adds £589. An adjustment is added of £183 for equal pay (this is an average for all teachers; the gain to the teachers affected is greater). For Secondary Schools, the base is total incremental salaries, divided by the number receiving incremental salaries in 1974, adjusted to 1970 prices, vielding £1,307. The 2.75 per cent annual increment adds £514. The "bulge" adjustment, on the assumption that one-half the number of Secondary/Comprehensive/Community teachers in 1974 will get 4 per cent per year more as they rise through the ranks, yields an average figure of $\pounds 399$. The equal pay adjustment, including an additional "bulge" effect, is assumed at £150. This provides an average incremental salary of £2,370, in 1970 prices. (As a memo item, if the school salary of £260-the 1974 amount in 1970 prices-is added, the total salary is $\pounds 2,630$.) In Vocational Schools, the base is determined by dividing the Vocational "instructional" budget item by full-time-equivalent teachers, on the basis of three part-time teachers equal one full-time teacher. This yields a base of $\pounds 1,387$. The annual 2.75 per cent increment adds $\pounds 545$ in 12 years. The "bulge" adjustment is done as in the Secondary schools, except that .375 instead of half of the teachers employed in 1974 are assumed to be affected; the resulting amount is £317, averaged over all teachers. Again, £150 is assumed to be the average cost, in 1970 prices, of equal pay. This yields a mean figure per full-time-equivalent teacher of £2,399. The Comprehensive/Community average salary expense in 1986 is assumed to be the midpoint between those of Vocational and Secondary (where the latter includes the school salary), i.e., £2,515. All Reformatory and Industrial School expenditures are assumed to grow proportionately to the rest of first and second level combined. Capital expenditures are calculated as follows. In the National Schools, over the twelve years 1963-74, new and expanded National Schools have created 21,000 places per year average, in addition to the added number of pupils, at an average price (1970 prices) of $\pounds 185$ (standard deviation of $\pounds 26.38$). If it is assumed that 21,000 places will need to be provided in each of the twelve years 1974-86, in addition to the 7,000 places implied by enrolment growth of 84,100, then 28,000 places will be provided. If £190 per place is taken to be the 1974 base, 3 per cent annual growth in costs yields a 1986 cost per place of £272. Similar statistics indicate a need for 6,500 Secondary/Comprehensive/Community places each year for obsolesence and mobility, and 3,500 more for growth, giving a total of 10,000 places per year. The Secretariat of Catholic Secondary Schools estimates a 1974 cost per pupil place of £1,000, which is £650 in 1970
prices. Growth in cost per pupil place at 3 per cent per year yields £932. In this sector, which includes both Secondary and Community schools, it is assumed that the State share will be 90 per cent of the per pupil cost, or £839. In Vocational Schools, it is assumed that 2,300 replacement places, for obsolescence and mobility and 1,950 new places, for growth, or a total of 4,250, will be required each year, and that the cost will be as in Secondary/Comprehensive/Community, i.e., £932, all paid for from public (though not necessarily State) sources.

REFERENCES

- AKENSON, DONALD, H., 1967. The Irish Education Experiment. London: Routledge & Kegan Paul.
- ATKINSON, NORMAN, 1969. Irish Education. Dublin: Allen Figgis.
- BAUMOL, WILLIAM J., 1967. "Macroeconomics of Unbalanced Growth." The American Economic Review.
- BREATHNACH, AGNES, 1976. "Towards the Identification of Educational Priority Areas in Dublin." The Economic and Social Review, Vol. 7, No. 4.
- BRUCK, DAVID, 1970. "The Schools of Lowell, 1824-1861: A Case Study in the Origins of Modern Public Education in America." Cambridge Mass: Harvard University, (unpublished Thesis, cited in Samuel Bowles and Herbert Gintis, 1970). Schooling in Capitalist America New York: Basic Books.
- CLARKIN, JAMES, 1969. Irish Education: Church State Relationships. unpublished Doctoral of Educational dissertation, Rutgers: The State University of New Jersey.
- CORCORAN, JOHN, J., 1973. The Regional Technical Colleges, dissertation presented to the National University of Ireland for the Degree of Master of Arts (pp 38, 39).
- COUNCIL OF EDUCATION, 1954. Report of the Council of Education. Dublin: Stationery Office.
- COUNCIL OF EDUCATION, 1960. Report of the Council of Education. Dublin: Stationery Office.
- CSO, 1969. Household Budget Inquiry, 1965-66. Dublin: Stationery Office.
- CSO, 1976. Household Budget Survey 1973. Dublin: Vol. 1 Summary Results, Stationery Office.
- CSO, 1976. National Income and Expenditure 1974. Dublin: Stationery Office.
- GEAREY, VINCENT, 1973. "A Comparison of Secondary School Entrants, Vocational School Entrants and Terminal Leavers." Irish Journal of Education. Vol. 7, No. 2. (Winter).

HMSO, 1976. Population Trends 4. London: Summer.

- HORGAN, JOHN, 1973. "Education in the Republic of Ireland." in Robert Bell, et al. (eds.) Education in Great Britain and Ireland. London: An Open University Source Book. Routledge & Kegan Paul.
- HORGAN, JOHN, 1976. "Religious Discrimination in Secondary Schools." *Hibernia* 7th May.
- HUGHES, J. G. and B. M. WALSH, 1976. "Migration Flows between Ireland, the United Kingdom and the Rest of the World, 1966-71." European Demographic Information Bulletin, 7 (Winter).
- KEATING, W., 1977 "An Analysis of Recent Demographic Trends with Population Projections for the Years 1981 and 1986" Paper read to the Statistical and Social Inquiry Society of Ireland. March 3.
- LYNN, RICHARD, 1968. The Irish Brain Drain. Dublin: The Economic and Social Research Institute, Paper No. 43.

LYONS, F.S.L., 1971. Ireland Since the Famine. London: Weidenfeld and Nicolson.

186

Mac ARDLE, DOROTHY, 1965. The Irish Republic. New York: Farar, Straus and Girous.

- McELLIGOTT, T. J. 1966. Education in Ireland. Dublin: Institute of Public Administration.
- MURPHY, D.C. 1976. "1973 Household Budget Survey, Special Features and Results" Paper read to The Statistical and Social Inquiry Society of Ireland, May 20.
- NESC 1975. Population and Employment Projections 1971-1986. Dublin: Report No. 5, Stationery Office.
- NESC, 1975. Jobs and Living Standards: Projections and Implications, Dublin: Report No. 7. Stationery Office.
- NESC, 1976. Educational Expenditures in Ireland. Dublin: Report No. 12, Stationery Office.
- NESC, 1976. Population Projections 1971-86: The Implication for Education. Dublin: Report No. 18. Stationery Office.
- Ó BUACHALLA, S. 1977. "Education as an Issue in the First and Second Dáil." Administration, Vol. 25, Spring.
- PEARSE, PADRAIG, 1916. "The Murder Machine" Reprinted in Education, Economy and Politics: Case Studies, Parts 3 and 4, Milton Keynes.
- TUSSING, A. Dale, 1976(a). "Investment and its Control" in Noel Harris et al. The Irish Economy: What Has to be Done. Association of Scientific, Technical and Managerial Staffs.
- TUSSING, A. DALE, 1976(b). "Labour Force Effects of 1967/68 Changes in Education Policy in the Irish Republic." *The Economic and Social Review*, Vol. 7, No. 3.
- TUSSING, A. DALE, and JOHN A. HENNING, 19,4(a). 'Baumol's Disease': "A Closer Look at Unbalanced Growth." Paper presented to the New York State Economics Association. Buffalo: New York, 29 March.
- TUSSING, A. DALE and JOHN A. HENNING, 1974(b). "Long-run Growth of non-Defense Government Expenditure in the United States" Public Finance Quarterly, April.

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

THE ECONOMIC AND SOCIAL RESEARCH INSTITUTE

| | 1 |
|---|---------------------|
| Broadsneet Series: | |
| 1. Dental Services in Ireland | P. R. Kaim-Caudle |
| 2. We Can Stop Rising Prices | M. P. Fogarty |
| 3. Pharmaceutical Services in Ireland | P. R. Kaim-Caudle |
| assisted by Annette O'Toole and Ka | thleen O'Donoghue |
| 4. Ophthalmic Services in Ireland | P. R. Kaim-Caudle |
| assisted by Kathleen O'Donoghue an | d Annette O'Toole |
| 5. Irish Pensions Schemes, 1969 P. R. Kaim-Cau | dle and J. G. Byrne |
| assisted b | y Annette O'Toole |
| 6. The Social Science Percentage Nuisance | R. C. Geary |
| 7. Poverty in Ireland: Research Priorities | Brendan M. Walsh |
| 8. Irish Entrepreneurs Speak for Themselves | M. P. Fogarty |
| 9. Marital Desertion in Dublin: an exploratory study | Kathleen O'Higgins |
| 10. Equalization of Opportunity in Ireland: Statistical Aspects | |
| R. C. Geary and F. S | 6. Ó Muircheartaigh |
| 11. Public Social Expenditure in Ireland | Finola Kennedy |
| 12. Problems in Economic Planning and Policy Formation in Ireland, . | 1958-1974 |
| | Desmond Norton |
| 13. Crisis in the Cattle Industry R. O'Co | onnor and P. Keogh |
| 14. A Study of Schemes for the Relief of Unemployment in Ireland | |
| R. C. Gea | ry and M. Dempsey |
| with Appendix | E. Costa |
| | * |
| | |
| Canaral Research Series | |
| 1 The Ownership of Personal Property in Ireland | Edward Nevin |
| 2. Short Term Economic Forecasting and its Application in Ireland | Alfred Kuehn |
| 2. The Irich Teriff and The F.F.C. A Factual Survey | Edward Nevin |
| A Demand Relationships for Ireland | C E. V. Leser |
| 5. Local Convernment Finance in Ireland: A Preliminary Survey | David Walker |
| 6. Prochasts of the Irish Economy in 1962 | Alfred Kuehn |
| 7. The Irich Woollen and Worsted Industry 1946-59: A Study in Sta | tistical Method |
| 7. The Thish Woolien and Worstea Thaastry, 10 10 00. 12 Datay in Dia | R. C. Gearv |
| P. The Allocation of Public Funds for Social Development | David Walker |
| 0. The Inich Price Level: A Comparative Study | Edward Nevin |
| 9. The INDRINCE Level. A Gomparative Study | D. I. Revnolds |
| 10. Intuka Transport in Treana. In Fucture Survey | Edward Nevin |
| 11. Public Debt and Economic Development | Edward Nevin |
| 12. Wages in Ireland, 1940-02 | D L Reynolds |
| 13. Road Iransport. The Problems and Prospects in Preland | C. E. V. Leser |
| 14. Imports and Economic Growth in Ireana, 1347-01 | C. E. V. Leser |
| 15. Ine inits a local transformed in 1902 und 1903 16. L'h Oriente Income in 1060 F A Atture | od and R. C. Gearv |
| 10. ITISA Country Incomes in 1900 E. A. Altwo | Edward Nevin |
| 17. Ine Capital Stock of Irish Industry | David Walker |
| 18. Local Government rinance and Gounty Incomes | David O'Mahony |
| 19. Inaustrial Relations in Ireland: The Background | D R Kaim Caudle |
| YU NACIAI NECUTIV IN ITELUNU UNU WESLETI EUTOVE | I TALIXAMIN GAUGE |

20. Social Security in Ireland and Western Europe 21. The Irish Economy in 1963 and 1964

22. The Cost Structure of Irish Industry, 1950-60

23. A Further Analysis of Irish Household Budget Data, 1951-1952 C. E. V. Leser 24. Economic Aspects of Industrial Relations David O'Mahony

25. Psychological Barriers to Economic Achievement

C. E. V. Leser

Edward Nevin

P. Pentony

General Research Series—continued

26. Seasonality in Irish Economic Statistics C. E. V. Leser C. E. V. Leser 27. The Irish Economy in 1964 and 1965 28. Housing in Ireland : Some Economic Aspects P. R. Kaim-Caudle 29. A Statistical Study of Wages, Prices and Employment in the Irish Manufacturing. Sector C. St. J. O'Herlihy 30. Fuel and Power in Ireland: Part I. Energy Consumption in 1970 J. L. Booth 31. Determinants of Wage Inflation in Ireland Keith Cowling 32. Regional Employment Patterns in the Republic of Ireland T. J. Baker 33. The Irish Economy in 1966 The Staff of The Economic and Social Research Institute 34. Fuel and Power in Ireland: Part II. Electricity and Turf J. L. Booth 35. Fuel and Power in Ireland: Part III. International and Temporal Aspects of Energy Consumption J. L. Booth 36. Institutional Aspects of Commercial and Central Banking in Ireland John Hein 37. Fuel and Power in Ireland: Part IV. Sources and Uses of Energy J. L. Booth C. Ĕ. V. Leser 38. A Study of Imports 39. The Irish Economy in 1967 The Staff of The Economic and Social Research Institute 40. Some Aspects of Price Inflation in Ireland R. C. Geary and J. L. Pratschke 41. A Medium Term Planning Model for Ireland David Simpson 42. Some Irish Population Problems Reconsidered Brendan M. Walsh 43. The Irish Brain Drain Richard Lynn 44. A Method of Estimating the Stock of Capital in Northern Ireland Manufacturing Industry: Limitations and Applications C. W. Jefferson 45. An Input-Output Analysis of the Agricultural Sector of the Irish Economy in 1964 R. O'Connor with M. Breslin 46. The Implications for Cattle Producers of Seasonal Price Fluctuations R. O'Connor 47. Transport in the Developing Economy of Ireland John Blackwell 48. Social Status and Inter-Generational Social Mobility in Dublin Bertram Hutchinson 49. Personal Incomes by County, 1965 Miceal Ross 50. Income-Expenditure Relations in Ireland, 1965-1966 John L. Pratschke 51. Costs and Prices in Transportable Goods Industries W. Black, J. V. Simpson, D. G. Slattery 52. Certain Aspects of Non-Agricultural Unemployment in Ireland R. C. Geary and J. G. Hughes 53. A Study of Demand Elasticities for Irish Imports Dermot McAleese 54. Internal Migration in Ireland R. C. Geary and J. G. Hughes C. J. Gillman with Appendix 55. Religion and Demographic Behaviour in Ireland B. M. Walsh R. C. Geary and J. G. Hughes with Appendix 56. Views on Pay Increases, Fringe Benefits and Low Pay H. Behrend, A. Knowles and J. Davies 57. Views on Income Differentials and the Economic Situation H. Behrend, A. Knowles and J. Davies 58. Computers in Ireland F. G. Foster Richard Lynn 59. National Differences in Anxiety 60. Capital Statistics for Irish Manufacturing Industry C. W. Jefferson 61. Rural Household Budget-Feasibility Study Sile Sheehy and R. O'Connor 62. Effective Tariffs and the Structure of Industrial Protection in Ireland Dermot McAleese

General Research Series - continued

63. Methodology of Personal Income Estimation by County

64. Further Data on County Incomes in the Sixties

Micael Ross Micael Ross J. G. Hughes E. W. Henry

65. The Functional Distribution of Income in Ireland, 1938-70

- 66. Irish Input-Output Structures, 1964 and 1968
- 67. Social Status in Dublin: Marriage, Mobility and First Employment
- Bertram Hutchinson 68, An Economic Evaluation of Irish Salmon Fishing, 1: The Visiting Anglers R. O'Connor and B. J. Whelan
- 69. Women and Employment in Ireland: Results of a National Survey Brendan M, Walsh assisted by Annette O'Toole
- 70. Irish Manufactured Imports from the UK in the Sixties: The Effects of AIFTA Dermot McAleese and John Martin
- 71. Alphabetical Voting: A Study of the 1973 General Election in the Republic of Ireland Christopher Robson and Brendan M. Walsh
- 72. A Study of the Irish Cattle and Beef Industries Terence J. Baker, Robert O'Connor and Rory Dunne
- 73. Regional Employment Patterns in Northern Ireland William Black and Clifford W. Jefferson
- 74. Irish Full Employment Structures, 1968 and 1975 E. W. Henry
- 75. An Economic Evaluation of Irish Salmon Fishing, II: The Irish Anglers
 - R. O'Connor, B. J. Whelan, and A. McCashin
- 76. Factors Relating to Reconviction among Young Dublin Probationers Ian Hart
- 77. The Structure of Unemployment in Ireland, 1954-1972 Brendan M. Walsh
- 78. An Economic Evaluation of Irish Salmon Fishing, III: The Commerical Fishermen B. J. Whelan, R. O'Connor, and A. McCashin
- 79. Wage Inflation and Wage Leadership W. E. J. McCarthy, J. F. O'Brien and V. G. Dowd
- 80. An Econometric Study of the Irish Postal Services Peter Neary
- 81. Employment Relationships in Irish Counties _____ Terence J. Baker and Micael Ross
- 82. Irish Input-Output Income Multipliers 1964 and 1968

J. R. Copeland and E. W. Henry

- 83. A Study of the Structure and Determinants of the Behavioural Component of Social Attitudes in Ireland E. E. Davis
- 84. Economic Aspects of Local Authority Expenditure and Finance
- J. R. Copeland and Brendan M. Walsh 85, Population Growth and other Statistics of Middle-sized Irish Towns
- D. Curtin, R. C. Geary, T. A. Grimes and B. Menton 86. The Income Sensitivity of the Personal Income Tax Base in Ireland, 1947-1972
- Brendan K. Dowling 87. Traditional Families? From Culturally Prescribed to Negotiated Roles in Farm Families Damian F. Hannan and Louise A. Katsiaouni
- An Irish Personality Differential: A Technique for Measuring Affective and Cognitive Dimensions of Attitudes Toward Persons – E. E. Davis and Mary O'Neill
- 89. Redundancy and Re-Employment in Ireland
- Brendan J. Whelan and Brendan M. Walsh 90. A National Model of Fuel Allocation – A Prototype E. W. Henry and S. Scott
- 91. A Linear Programming Model for Irish Agriculture
- Robert O'Connor, Miceal Ross and Michael Behan 92. Irish Educational Expenditures—Past Present and Future A. Dale Tussing

Printed by the Brunswick Press Limited, Sandymount, Dublin 4.